PAIN MANAGEMENT IN HAIL REGION HOSPITALS IN SAUDI ARABIA

By

Hamdan Albaqawi

DipNsg, Bcs, RN, MSN

School of Health Sciences College of science, Engineering and Health RMIT University May 2014

Submitted in Total Fulfilment for the Degree of Doctor of Philosophy (PhD)

Abstract

Background and aims: Pain is a human experience that affects the overall quality of life, and it is known to be the most common reason for people seeking healthcare. Nurses play a crucial role in assessing and managing patients' pain. Effective pain management requires precise knowledge, attitudes and competent assessment skills. Knowledge deficits and inappropriate attitudes are major contributing factors to the under-treatment of pain. Pain management is a multifaceted problem that may become even more complicated in situations where there are cultural variances and differences between patients and nurses. This study aimed to determine nurses' knowledge and attitudes regarding pain management, and to identify possible barriers to achieving optimal pain management in Hail region hospitals in Saudi Arabia.

Sample and methods: This is an explorative, descriptive, mixed-methods (quantitative and qualitative) study conducted in the Hail region hospitals to identify nurses' knowledge and attitudes towards pain management. The first phase involved administering a questionnaire to a sample of 303 nurses to explore their knowledge and attitudes regarding pain management. The questionnaire used in the study is the 'Knowledge and Attitudes Survey Regarding Pain' (KASRP) tool. The second phase involves semi-structured interviews with 28 nurses who had previously completed phase one to further explore their perceived facilitators and barriers to proper pain management. The interviews elicit information on how cultural differences among Saudi national and expatriate nurses might affect the assessment and interpretation of patients' pain, and how it will affect the delivery of effective pain management, as well as identifying the barriers to achieving optimal pain management in Hail region hospitals. Data

ii

are analysed using descriptive statistics, measures of variability and inferential statistics, and thematic analysis for qualitative data.

Results: In this study, the average correct response rate to the KASRP questionnaire was 41.75 %, with rates ranging from 5–87 %. The findings indicate inadequate knowledge regarding pain, pain assessment, pain management and pain medications. When analysed using thematic analysis, the qualitative data highlight some barriers that affect nurses' ability to provide effective pain management; these include language, workload, inadequate numbers of staff, lack of education and cultural orientation courses, and religious and cultural factors.

Recommendations: Nurses require a greater knowledge base regarding pharmacological and non-pharmacological interventions in pain management, as well as training to acquire culture competence to care for patients with different cultural backgrounds. This study recommends increased education regarding pain management, as well as an annual assessment of skills for all clinical nurses. Strategies to recruit and retain experienced staff should be implemented.

Conclusion: This study provides a unique insight into pain management practice by assessing overall knowledge scores and exploring the barriers and facilitating factors that exist within a multinational workforce of nurses working in five major Saudi health services. As such, this study is able to make a claim for new and unique knowledge that is relevant to nursing practices. The results of this study indicate problems in nurses' knowledge and negative attitudes regarding pain and pain management. A lack of knowledge regarding pharmacological and non-pharmacological interventions is evident, and educational and training programs should be implemented to correct these deficiencies.

iii

Keywords: nurses' knowledge, nurses' attitudes, pain assessment, pain management, culture,

Saudi Arabia.

Declaration

I certify that:

- a) except where due acknowledgement has been made, the work is that of the author alone;
- b) the work has not been submitted previously, in whole or in part, to qualify for any other academic award;
- c) the content of the thesis is the result of work which has been carried out since the official commencement date of the approved research program;
- d) any edited work, paid or unpaid, carried out by a third party is acknowledged; and,
- e) ethics procedures and guidelines have been followed.

Hamdan Albaqawi

Acknowledgements

Completing my PhD degree is probably the most challenging activity of the first 31 years of my life. However, it is a challenge that I had to undertake, and I consider this thesis the beginning of my journey.

A thesis takes a great deal of planning and research, followed by the subsequent writing and presentation. Although this long-drawn-out and arduous task is described as my own, there are many people who assisted me in completing it. I would never have been able to finish it without their guidance, support and encouragement. Foremost, I would like to express my sincere gratitude to my supervisors, Associate Professor PhilMaude and Associate Professor Lina Shawahn-Akl, for their continuous support, patience, motivation, enthusiasm and immense knowledge. My research would not have been possible without their guidance and support.

In addition to my advisors, I would like to thank my parents, brothers and sisters for their love and support throughout my life. I offer my heartfelt thanks to my beloved wife and my children, Mohammad and Wesam, for their support, patience and encouragement during my studies and stay in Australia. I appreciate your sacrifices in the interest of my success.

To all of my friends, who provided great companionship to me during my study, thank you for your understanding and encouragement in many moments of crisis. Your friendship made my life easier and a wonderful experience.

I would also like to thank all of the staff at the University of Hail for their cooperation and support, and for the scholarship that enabled me to successfully complete my PhD.

vi

My special thanks are extended to the nursing staff of the hospitals in the Hail region of Saudi Arabia, who contributed by responding to the questionnaire survey and sharing their experiences. Without your cooperation, this research would not have been possible.

Dedication

This thesis is dedicated to my father, who was my first teacher in this world and who taught me letters and numbers when I was very young. It is also dedicated to my mother, who taught me patience and tolerance. I also dedicate this work to my brothers and sisters for their constant encouragement and support. I dedicate this achievement to my beloved wife and to my children for their love, patience and unlimited support during this journey. I would not have been able to accomplish this without you all.

Contents

Abstract	ii
Declaration	v
Acknowledgements	vi
Dedication	viii
Contents	iv
	IX
List of Appendices	xii
List of Figures	xiii
List of Tables	xiv
Chapter 1: Introduction and Background	1
1 1 Introduction	1
1 2 Statement of the Problem	3
1.3 Definition of Pain	5
1.4 Context	5
1.5 Health System in Saudi Arabia	
1.6 Nursing in Saudi Arabia	10
1.7 Rationale of the Study	12
1.8 Purpose of the Study	13
1.9 Research Questions	13
1.10 Organisation of the Thesis	14
1.10 Organisation of the Thesis	1 7
Chapter 2: Literature Review	15
2.1 Introduction	15
2.2 Part 1: Contemporary Literature on the Human Experience of Pain	15
2.2.1 Types of Pain	17
2.2.1.1 Acute Pain	18
2.2.1.2 Chronic Pain.	19
2.2.1.3 Nociceptive Pain.	20
2.2.1.4 Psychogenic Pain	21
2.2.2 Theories of Pain	21
2.2.2.1 Specificity Theory.	21
2.2.2.2 Pattern Theory.	22
2.2.2.3 Gate Control Theory of Pain.	23
2.2.3 Prevalence of Pain	24
2.2.4 Assessment of Pain	27
2.2.4.1 Assessment Tools.	29
2.2.5 Management of Pain	32
2.2.5.1 Under-treatment of Pain	34
2.2.5.2 Consequences of Under-treatment of Pain	35
2.2.6 Nurses' Roles in Pain Assessment and Management	37
2.2.7 Nurses' Misconceptions and Negative Attitudes Regarding Pain Assessment	
and Management	38
2.2.8 Barriers in Management of pain	39
2.2.8.1 Nurse-related Barriers.	40
2.2.8.2 Patient-related Barriers.	41
2.2.8.3 Healthcare system-related Barriers.	43

2.3 Part 2: Critical Review of Nurses' Knowledge and Attitudes Regarding Pain:	
Search Strategy	
2.3.1 Nurses' Knowledge and Attitudes Regarding Pain	
2.4 Part 3: Culture and Pain	
2.4.1 Definitions of Culture and Cultural Competence	
2.4.2. Conceptual Framework of Leninger's Cultural Care Diversity and Universa	lity 57
Theory (CCDU)	
2.5 Chapter Summary	
Chapter 3: Research Methodology	63
3.1 Introduction	
3.2 Mixed Methods Design	
3.2.1 Philosophical Foundation for Mixed-methodsResearch	
3.3 Study Design	67
3.3.1 Sequential Explanatory Mixed-methodsDesign	67
3.4 Study Setting	
3.5 Sampling	
3.5.1 Sample in Phase One (Quantitative)	
3.6 Research Instruments	
3.6.1 Demographic Data	
3.6.2 Knowledge and Attitudes Survey Regarding Pain (KASRP)	
3.6.3 Tests of Validity and Reliability	
3.7 Procedure of Survey Distribution	
3.8 Phase One Data Analysis (Quantitative)	73
3.8.1 PostHoc Multiple Comparison Tests	75
3.8.2 Homogeneity of Variance	
3.8.3 Statistical Significance	
3.8.4 Normality	
3.8.5 Clinical Significance	
3.9 Phase Two (Qualitative)	
3.9.1 Interviews	
3.9.2 Transcribing Qualitative Data	
3.9.3 Phase Two Data Analysis (Qualitative)	
3.10 Ethical Issues (Confidentiality, Anonymity and Protection of Human Rights)	
3.11 Chapter Summary	
Chapter 4: Quantitative Results	
4.1 Introduction	
4.2 Demographics of Participants	
4.3 Nurses' Knowledge and Attitudes Regarding Pain Management in Hail Region	
Hospitals	
4.4 Effect of Demographic and Cultural Factors on the Delivery of Effective Pain	
Management	
4.4.1 H0 ₁ : Effect of Gender	
4.4.2 H ₀ 2. Effect of Age	94
4.4.3 H ₀ 3. Effect of Nationality	96
4.4.4 H ₀ 4. Effect of Academic Qualification	
$4.4.5 H_0 5$. Effect of Religion	101
$4.4.6 H_0 6$. Effect of Experience	103
$4.4.7 H_0 7$. Effect of Position	106
4.4.8 H_0 8. Effect of the Department of Placement	108
$4.4.9 H_09$. Effect of Hospitals	111
4.4.10 H ₀ 10. Effect of Participation in Pain Courses	113

Chapter 5: Qualitative Data Analysis	117
5.1 Introduction	117
5.2 Overview of Participants	118
5.3 Findings	119
5.3.1 Perceived Knowledge and Familiarity with Pain Management and Medications.	120
5.3.1.1 Perceptions and Satisfaction with Level of Knowledge of Pain	
Management	121
5.3.1.2 Familiarity with, and Knowledge of Pain Medications Used	126
5.3.1.3 Interest in Learning about Pain Management.	127
5.3.2 Current Pain Management Processes and Practices	128
5.3.2.1 Alternative Forms of Management Used (Non-pharmaceutical).	129
5.3.2.2 Level of Satisfaction with Current Pain Management Practices and	
Processes	132
5.3.2.3 Nursing Responsibility in Pain Management.	134
5.3.3 Nursing Pain Assessment	136
5.3.3.1 Nurses' Pain Assessment in Terms of the Tools Used	136
5.3.3.2 Training for Pain Assessment	138
5.3.3.3 Perceived Efficacy of Pain Assessment Tools.	139
5.3.4 Barriers and Limitations to Optimal Pain Management	141
5.4.4.1 Language and Communication Barriers.	141
5.4.4.2 Workload Due to Nursing Staff Shortages.	144
5.4.4.3 Cultural, Religious and Other Factors	146
5.3.5 Factors to Improve Pain Management	151
5.4 Conclusion	154
Chapter 6: Discussion Chapter	158
6.1 Introduction	158
6.2 Discussion	158
6.2.1 Nurses' Knowledge and Attitudes Regarding Pain Management	159
6.2.1.1 Nurses' Pain Assessments	160
6.2.1.2 Nurses' Attitudes	161
6.2.1.3 Nurses' Knowledge of Pharmacology	163
6.2.2 Barriers to Achieving Optimal Pain Management as Perceived by Nurses	172
6.2.3 Demographic and Cultural Determinants Affecting the Delivery of Effective	
Pain Management	176
6.2.4 Pain Management, Culture and Care	179
6.3 Conclusion	182
Chapter 7: Conclusion	185
7.1 Introduction	185
7.2 Reflection on the Study and its Design	185
7.3 Strengths of the Study	187
7.4 Limitations of the Study	189
7.5 Recommendations Arising from the Study	190
7.5.1 Recommendations for practice.	190
7.5.2 Recommendations for training and education.	192
7.5.3 Recommendations for policy.	193
7.5.4 Recommendations for future research.	193
7.6 Study Conclusion	195
References	198

List of Appendices

Appendix A: Studies that Utilised the KASRP Tool	. 233
Appendix B: Demographic Data	. 235
Appendix C: Consent Form for Participants	. 236
Appendix D: Plain Language Statement	. 237
Appendix E: Letter to Participate in Interview	. 239
Appendix F: Knowledge and Attitudes Survey Regarding Pain	. 240
Appendix G: Ethics Approval from RMIT University	. 250
Appendix H: Permission from Hospitals	. 251
Appendix I: Questions for Semi-structured Interview	. 252
Appendix J: Conference Abstract	. 254

List of Figures

Figure 2.1: Sunrise Model	59
Figure 4.1: Frequency distribution of correct answers (per cent) by n = 303 participants	88
Figure 4.2: Frequency distributions of correct answers (%) by gender	93
Figure 4.3: Frequency distributions of correct answers (%) by age	95
Figure 4.4: Frequency distributions of correct answers (%) by nationality	97
Figure 4.5: Frequency distributions of correct answers (%) by academic qualification	100
Figure 4.6: Frequency distributions of correct answers (%) by religion	102
Figure 4.7: Frequency distributions of correct answers (%) by experience	104
Figure 4.8: Frequency distributions of correct answers (%) by position	107
Figure 4.9: Frequency distributions of correct answers (%) by departments	109
Figure 4.10: Frequency distributions of correct answers (%) by hospitals	112
Figure 4.11: Frequency distributions of correct answers (%) by courses	114
Figure 4.12: Comparison of effect sizes by demographic characteristics	116

List of Tables

Table 2.1: Search Terms and Process	45
Table 3.1: Reliability Testing of the KASRP	72
Table 4.1: Demographic Profile of Nurses	86
Table 4.2: Distributions of Scores for the Knowledge and Attitudes Questionnaire	
Regarding Pain	89
Table 4.3: Ranking of NKASRP Answers from Most Correct to Least Correct	90
Table 4.4: Frequency Distribution of Participants by Gender	92
Table 4.5: Descriptive Statistics for Correct Answers (%) by Gender	93
Table 4.6: Levene's Test for Equality of Variances	93
Table 4.7: ANOVA Test for Between-Subjects Effects	94
Table 4.8: Frequency Distribution of Participants by Age	94
Table 4.9: Descriptive Statistics for Correct Answers (%) by Age	95
Table 4.10: Levene's Test for Equality of Variances	95
Table 4.11: ANOVA Test for Between-Subjects Effects	95
Table 4.12: Scheffé PostHoc Multiple Comparison Test	96
Table 4.13: Frequency Distribution of Participants by Nationality	97
Table 4.14: Descriptive Statistics for Correct Answers (%) by Nationality	97
Table 4.15: Levene's Test for Equality of Variances	98
Table 4.16: ANOVA Test for Between-Subjects Effects	98
Table 4.17: Scheffé PostHoc Multiple Comparison Test	98
Table 4.18: Frequency Distribution of Participants by Academic Qualification	99
Table 4.19: Descriptive Statistics for Correct Answers (%) by Academic Qualification	100
Table 4.20: Levene's Test for Equality of Variances	100
Table 4.21: ANOVA Tests for Between-Subjects Effects	101
Table 4.22: Frequency Distribution of Participants by Religion	101
Table 4.23: Descriptive Statistics for Correct Answers (%) by Religion	102
Table 4.24: Levene's Test for Equality of Variances	102
Table 4.25: ANOVA Test for Between-Subjects Effects	103
Table 4.26:Scheffé PostHoc Multiple Comparison Test	103
Table 4.27: Frequency Distribution of Experience	104
Table 4.28: Descriptive Statistics for Correct Answers (%) by Experience	105
Table 4.29: Levene's Test for Equality of Variances	105

Table 4.30: ANOVA Test for Between-Subjects Effects	
Table 4.31: Scheffé PostHoc Multiple Comparison Test	
Table 4.32: Frequency Distribution of Participants by Position	
Table 4.33: Descriptive Statistics for Correct Answers (%) by Position	
Table 4.34: Levene's Test for Equality of Variances	
Table 4.35: ANOVA Test for Between-Subjects Effects	
Table 4.36: Frequency Distribution of Participants by Departments	
Table 4.37: Descriptive Statistics for Correct Answers (%) by Departments	
Table 4.38: Levene's Test for Equality of Variances	110
Table 4.39: ANOVA Test for Between-Subjects Effects	110
Table 4.40: Scheffé PostHoc Multiple Comparison Test	111
Table 4.41: Frequency Distribution of Participants by Hospitals	111
Table 4.42: Descriptive Statistics for Correct Answers (%) by Hospitals	
Table 4.43: Levene's Test for Equality of Variances	
Table 4.44: ANOVA Test for Between-Subjects Effects	
Table 4.45: Scheffé PostHoc Multiple Comparison Test	
Table 4.46: Frequency Distribution of Participants by Courses	114
Table 4.47: Descriptive Statistics for % Correct Answers by Courses	115
Table 4.48: Levene's Test for Equality of Variances	
Table 4.49: ANOVA Test for Between-Subjects Effects	
Table 5.1: Demographic Characteristics	
Table 5.2: Summary of Themes	119

Chapter 1: Introduction and Background

1.1 Introduction

It can safely be said that no one can escape pain. Everyone will experience pain at some point in their lives; sometimes it is transient and sometimes longstanding (chronic pain). Pain is the main reason for people seeking healthcare (Lewis, Heitkemper & Dirksen, 2004; Polomano et al., 2008) either in a medical clinic or in the hospital setting (McLean et al., 2004). Despite the development of new techniques and new guidelines for adequate pain management, many patients continue to suffer from pain (Schechter, Berde & Yaster, 2003; Sloman et al., 2005; Sloman et al., 2006; Pasero & McCaffery, 2007; Horgas & Yoon, 2008; Layzell, 2008). Pain has often been poorly assessed and inadequately managed, and the under-treatment of pain has been reported for many decades as a major and persistent clinical problem (Brown, Bowman & Eason, 1999; Fosnocht, Swanson & Barton, 2005; McCaffery & Ferrell, 1997; McCaffery & Pasero, 1999; Schafheutle, Cantrill & Noyce, 2001;Duignan &Dunn, 2009).

A Swedish study was conducted to identify the prevalence and diagnostic pattern of pain over a period of one year at the primary care level. The study found that 25 % of patients that visit general practitioners do so because of a variety of pain conditions (Mantyselka et al., 2001; Hasselstrom et al., 2002). In Australia, it is estimated that one in five people (about 3.2 million Australians), including children and adolescents, will suffer chronic pain in their lifetime (Walsh et al., 2008). In Saudi Arabia, the situation is not clear because there is a lack of published studies identifying the prevalence of pain in healthcare settings and the level of nurses' knowledge regarding pain. Nonetheless, one study by Kaki, Daghistani and Msabeh (2009) to assess nurses' knowledge of acute pain management in a tertiary hospital in Jeddah, has drawn attention to the deficits in many aspects of pain such as physical dependence,

tolerance, addiction, self-report and pharmacological knowledge. Pain assessment is considered by the Joint Commission as the fifth vital sign and it is an essential human indicator (Lorenz et al., 2009). Pain assessment and management is the most fundamental aspect of a nurse's responsibility when attending to a patient complaining of pain (Innis, Bikaunieks, Petryshen, Zellermeyer & Ciccarelli, 2004; Rejeh, Ahmadi, Mohammadi, Kazemnejad & Anoosheh, 2009). However, the task can be highly influenced by nurses' knowledge, perceptions and attitudes regarding pain. Therefore, this study proposes to explore nurses' knowledge and attitudes regarding pain management practices in Hail region hospitals in Saudi Arabia, and to identify possible barriers to effective pain management for service providers' and patients' levels.

During their nursing careers, all nurses will manage patients who suffer from pain (Matthews & Malcolm, 2007). Therefore, nurses should be well equipped with knowledge on how to assess pain, implement and evaluate interventions. Moreover, nurses are required to participate in decision-making processes regarding pain treatment (Katsma & Souza, 2000). Jones et al. (2004) identified that nurses often have knowledge deficits and incorrect beliefs about pain assessment and management. These misconceptions and deficits can lead to inappropriate, incorrect and inadequate pain management practices (McCaffery & Ferrell, 1996; McCaffery & Pasero, 1999; Molony, Kobayashi, Holleran & Mezey, 2005; Twycross, 2002). A lack of knowledge about pain and pain treatment, as well as misplaced beliefs about addiction to pain medications, are considered significant barriers to effective pain management among nursing providers'. This thesis will make an original and important contribution to nursing knowledge, as it explores knowledge and attitudes concerning pain management practices in an Islamic country.

1.2 Statement of the Problem

The primary responsibility of healthcare professionals, including nurses, is to relieve pain and suffering. It is a moral and ethical responsibility and a fundamental human right for patients (Cousins, Brennan & Carr, 2004). Although pain can, in most instances, be effectively treated and relieved, the under-treatment of pain remains a significant clinical problem, and it continues to be an area of concern among health professionals, patients and healthcare organisations. Even in hospital settings, where pain should be treated effectively, research shows that pain is managed inadequately and that a large number of patients suffer from unrelieved pain (Huang et al., 2001; Dolin, Cashman& Bland, 2002). Elliot et al. (1999) conducted a study to find out the prevalence of chronic pain in the Grampian region of the UK with a random sample of (N = 5036) patients aged 25 and over. The study surveyed 29 general practices using a postal self-completion questionnaire. A total of 3,605 questionnaires were completed and returned by the participants. The results showed that half of the sample suffered from chronic pain. Furthermore, one half of the pain sufferers were categorised as having severe pain, and the authors concluded that the pain of those who continued to suffer had been inadequately managed. Unrelieved pain can negatively affect patients' sleep functions, moods and relationships with family, and it can have a significant economic effect on patients and their families (Schopflocher et al., 2010). Absenteeism from work among adults as a result of pain-related causes can have a significant negative effect on productivity and consequently on a country's economy (Phillips & Schopflocher, 2008).

Numerous studies indicate that nurses are not managing pain properly due to deficits in their knowledge and beliefs (Molony et al., 2005; Twycross, 2002). These studies have identified notable knowledge deficits and incorrect beliefs among nurses about pain assessment and its treatment. Furthermore, Twycross (2004) considered insufficient nursing educational

preparation during their studies at college, as a significant obstacle to effective pain management. Regarding the use of pain medication, a number of studies have found that one reason for inadequate treatment by nurses is their unfounded concerns about the possibility of addiction, in addition to the underestimation of patients' pain (McCaffery & Ferrell, 1996; Schafheutle, Cantrill & Noyce, 2001). Many studies have revealed that some nurses are reluctant to administer opioids due to a general 'opiophobia', compounding their negative attitudes towards pain (Drayer, Henderson& Reidenberg, 1999; Edwards et al., 2001; McCaffery & Ferrell, 1996; Furrow, 2001; Yates et al., 1998). Other studies have found that some nurses aim only to reduce pain rather than totally relieve it (Edwards et al., 2001; Twycross, 2002). Despite numerous studies identifying knowledge deficits in general pain management, the problem of patients suffering from unnecessary pain continues (Schechter, Berde& Yaster, 2003; Sloman et al., 2005; Sloman et al., 2006; Pasero & McCaffery, 2007).

The consequences of pain mismanagement result in both human suffering and increase in economic costs, to the extent that in 1999, the Joint Comission has established pain assessment as the fifth vital sign (Brennan, Carr & Cousins, 2007; Innis et al., 2004; Maclaren & Cohen, 2005). Pain is the third most costly health problem in Australia, costing the Australian economy over \$34.3 billion per annum or \$10,847 per person affected (Cousins, Bridenbaugh, Carr& Horlocker, 2009). In the United States (US), chronic pain affects more than one-third of the population, with an estimated annual cost of US\$100 billion (National Institutes of Health, 1998). In Canada, the economic effect of pain is over \$6 billion per year and around \$37 billion in productivity costs resulting from job losses and sick days (Phillips & Schopflocher, 2008). Adequate pain assessment and management has been found to reduce medication costs, improve patient outcomes and satisfaction with care, and shorten hospital stays (Innis et al., 2004; Polomano et al., 2008).

1.3 Definition of Pain

In 1968, McCaffery defined pain as 'whatever the experiencing person says it is, existing whenever she/he says it does' (McCaffery & Pasero, 1999, p. 17). This definition emphasises that patients' self-report is the most reliable indicator of pain, and only patients have the authority on their pain. Moreover, it emphasises that pain is a subjective experience. In 1979, the International Association for the Study of Pain (IASP) proposed the most broadly used definition of pain as 'an unpleasant sensory and emotional experience associated with actual or potential tissue damage, or described in terms of such damage' (Tracey & Mantyh, 2007, p. 377). This definition not only focuses on the pathophysiological origin of pain, but it also highlights the psychological aspects of the experience of pain. While this is a useful definition, it does not highlight the definition of chronic pain, which McCaffery and Beebe (1989) defined as:

Pain that has lasted 6 months or longer, is ongoing on a daily basis, is due to non-threatening causes, has not responded to currently available treatment methods, and may continue for the remainder of the patient's life (Dunajcik, 1999, p. 471).
Pain management is defined as 'the process of providing health care that alleviates or reduces pain' (Janmohamed & College, 2009, p. 13).

1.4 Context

As the current study relate to a particular region (Hail region) of the Kingdom of Saudi Arabia, it is prudent to provide a brief description of the country and its healthcare service. Saudi Arabia is a developing country located in the Middle East, and it occupies four-fifths of the land area of the Arabian Peninsula. The Kingdom's population is estimated to be around 27 million, 20 % of which are non-Saudi-born citizens. Ninety per cent of Saudi citizens are

Arabs in ethnicity, and all are Muslims (Central Intelligence Agency, 2013). The annual population growth rate of the country is 3.19 %, with a fertility rate of 2.93 % (Ministry of Health (MOH), 2011).

The total area of Saudi Arabia is 2,240,000 square kilometres (Al-Shahri, 2002). The capital city of Saudi Arabia is Riyadh, and the country is divided into 13 provinces. Saudi Arabia was established in 1932 under the leadership of King Abdul Aziz ibn Saud (Central Intelligence Agency, 2013). In 1934, exploration for oil was conducted throughout the Kingdom, and since the discovery of vast reserves of oil, Saudi Arabia has become one of the world's most prosperous oil-based economies. Saudi Arabia has become the largest proven oil reserve in the world, containing around 20 % of the world's proven oil reserves (Central Intelligence Agency, 2013). Exploration for oil and oil revenues has changed the country from one of the poorest countries to one of the highest in per capita income.

Saudi Arabia is an Islamic country and is considered the birthplace of Islam. The system of government is a monarchy, with the constitution guided by strict instructions of Islamic law (Vidyasagar & Rea, 2004). Islamic law forms the basis of the country's constitution and civil law, and it guides Saudis' daily lifestyles, including morals, dress, eating habits and business dealings. Makkah is an important and sacred city for Muslims, and it has been the centre of attraction for millions of pilgrims from all over the world for nearly 1,400 years. In Islam, there are two major denominations: Sunni and Shia. In Saudi Arabia, the majority of Saudis are Sunni and 10 % are Shia (Majidyar, 2013). It is also relevant to refer to certain cultural aspects that are unique to Saudi Arabia—particularly those that are gender-based—as they affect all spheres of life. Women are not allowed to drive cars; they have to rely on a male member of the family for transportation needs. Men and women do not mix in public, and facilities such as health services, banks and restaurants do provide separate sections for male

and female customers. Defrin, Eli & Dorit (2011) surveyed men and women (N 548) from Jewish, Muslim-Arab and Christian-Arab backgrounds concerning pain experience and willingness to report pain. They found that the men from the three ethno groups to be more stoic than women. Muslims believe that pain is the will of God and a trial of the person's faith and, therefore, ability to endure pain will be aptly remunerated (Mills, 2004; Lovering, 2006).

The Ministry of Health (MOH) was established in 1954 to provide free healthcare services to all Saudi citizens (Al-Osimy, 1994). The aim of the MOH is to provide a range of health services, such as preventive, curative, educational and rehabilitative, to the entire population through a network of hospitals and primary healthcare centres that are distributed throughout the country.

1.5 Health System in Saudi Arabia

Saudi Arabia is divided into 18 health service regions, and each region is directed by a general director who is attached to the MOH (Aboul-enein, 2002). The responsibility of the MOH is to supervise the regions' public and private healthcare sectors. Patient visits to MOH hospitals and health centres account for 50 % of the total visits, while those to other governmental sectors such as the military hospitals represent 17.6 % and the private sector represents 32.4 % (MOH, 2011). There are 420 MOH hospitals with a total of 58,696 beds (MOH, 2011). In most hospitals, the language spoken by the staff is English; however, a large percentage of Saudi patients and families only speak Arabic.

Saudi Arabia is rapidly developing and is witnessing significant improvements in many sectors, including the health sector. The Saudi Arabian government aims to provide all Saudi citizens with free, high-standard healthcare services (Alharthi et al., 1999). During the past

few decades, the nation has experienced significant growth and improvement in healthcare provisions, both in quantity and quality (Gallagher, 2002). Mitchell (2009) noted that Saudi Arabia had continued to invest and expand services in healthcare infrastructure and human resources in an effort to address the shortage of nurses. However, despite these developments in health services, there are still delays and deficits in health professional resource development. The healthcare system of Saudi Arabia has two branches. The first branch comprises the primary healthcare centres and clinics, which provide preventive, curative, prenatal, emergency and basic services. The second branch is represented by the hospitals and specialised treatment facilities located in urban areas. The budget of the MOH is 6.9 % of the total annual government budget (MOH, 2011). At present, the Saudi MOH is the main government source and financing body of healthcare assistance in Saudi Arabia. Each institution associated with the Saudi MOH (except for referral hospitals, teaching hospitals and the Red Crescent Society) delivers services to a particular population, which includes employees and their dependents. In addition, the majority of these agencies provide health services to all residents in times of calamities and emergencies (Mufti, 2000).

The progress in the provision of high standard healthcare services, coupled with components such as an enhanced and more open educational system, greater utilisation of health services, and an augmented social environment, have all played a substantial role in the significant enhancement of health indicators of the population (Al-Homayan, Shamsudin, Subramaniam & Islam, 2013). However, it should be noted that although there is a growth in the volume of health service providers, there is still a lack of collaboration among various institutions, leading to the failure in the optimal use of resources and unnecessary duplication of efforts (Alhusaini, 2006).

To be effective and provide the community with updated, cost-effective, affordable and wellarranged intensive healthcare services, a state ruling incorporating the health strategy was implemented in 2002, by the Council of Health Services in coordination with other government authorities, the MOH and representatives from private health organisations (MOH, 2008). The aims of the health facilities were centred on the eight components of the Primary Health Care (PHC) strategy: providing awareness to the community regarding existing health concerns and various ways of preventing and managing them; providing a sufficient quantity of safe water and essential sanitation; promoting food supply and adequate nutrition; providing complete maternal and child health management; making children's immunisation programs available for infectious diseases; preventing and controlling local endemic diseases; providing adequate management of typical diseases and ailments; and providing a sufficient supply of vital medications (Al Mazrou & Salem, 2004).

Despite all of the advancements and successes associated with the implementation of better healthcare provisions, budgetary control remains a major concern for the Saudi MOH. Given that the total expenses for local healthcare provisions are sourced from the government and that health services are free to the public, there is significant pressure on the government's budget, especially in view of the rapid population growth, high costs of modern technology and the public's increasing awareness of health-related matters (Walston, Al-Harbi & Al-Omar, 2008). In response to the community's increasing demand for healthcare, and to guarantee the quality of services delivered, the government established the Council for Cooperative Health Insurance in 1999 (Walston, Al-Harbi & Al-Omar, 2008). The main functions of the council include the introduction, regulation and supervision of a health insurance plan for the Saudi healthcare market.

1.6 Nursing in Saudi Arabia

Nurses are the primary caregivers for patients and are considered the main professional branch of frontline staff in the healthcare system. Nurses play an essential role in delivering safe and effective care. However, an ageing population worldwide, along with the resulting demand on healthcare and the strain on the capacity of nurse-training institutions, has resulted in most countries having difficulties in recruiting and retaining qualified nursing staff (Cohen & Van Nostrand, 1995). In Saudi Arabia, the population is growing rapidly, and the need for better-prepared nurses and high-quality health services is necessary to meet the health needs of the country. A primary concern for the effective transformation of the Saudi healthcare structure is the need to create and implement practical strategies to retain and attract more Saudi nationals to medical and healthcare professions—particularly nursing. Several initiatives have been implemented by the local government to educate and train Saudi nationals for health-related jobs. Since 1958, several medical, nursing and healthcare institutions have been established around the country to meet this target (Aldossary, While & Barriball, 2008). Mebrouk (2008) suggests that Saudi Arabia's dependents on immigrant nurses is due to the following factors: the poor public image of the nursing profession and rapid population growth.

Over the past few years, the nursing shortage has become a worldwide dilemma, and Saudi Arabia is no exception. However, the situation in Saudi Arabia is unique. Despite nursing education being available in Saudi Arabia since 1961, the number of Saudi nurses is only slowly increasing (Al-Ahmadi, 2009). There are many reasons for this, including low salaries compared to workloads, shift schedules and social perceptions of the nursing profession (Al-Ahmadi, 2002). To fill this gap, Saudi Arabia is still extensively dependent on expatriate nurses who are recruited from countries such as India, the Philippines, North America, the United Kingdom, Australia, South Africa, Malaysia and other Middle Eastern countries to provide care for the Saudi Arabian population (Miller-Rosser, Chapman & Francis, 2006; Tumulty, 2001). Large numbers of expatriates come to work in Saudi Arabia; they are attracted by the low cost of living, tax-free salaries, annual leave of up to 54 days, yearly service awards, provision of free food and furnished accommodation, an annual round trip ticket to the home country from point of hire, and free medical coverage. However, the principal disadvantage is that they come from traditions and cultures that differ from those of the Saudi nationals for whom they are expected to provide care. Migrant nurses often arrive with only a partial awareness of the culture, traditions and religion of Saudi nationals. Aldossary, While and Barriball (2008) outlined this issue and highlighted the challenges faced by migrant nurses in attending to Saudi patients.

The government introduced the concept of 'Saudization', which means finding local workers to perform specific jobs that only immigrants have been willing to handle—particularly at the prevailing wage rates. Saudization has been implemented since the year 2000, and has been applied-to the nursing profession as well with the view to replace the largely expatriate nurses with those of Saudi origin (Tumulty, 2001). Healthcare is one of the largest sectors in Saudi Arabia that engaged in the process of Saudization to increase the number of Saudi workers by targeting nurses for recruitment and training (Aboul-enein, 2002). In this regard, the budgetary allocation for training and scholarships has been augmented, and several employees within the MOH have been offered an option to continue their studies abroad (Tumulty, 2001). This strategic plan aims to enhance the skills of current employees, increase the quality of healthcare and lower the turnover rate among healthcare professionals. The King Abdullah international scholarship program, which was founded by the Ministry of Higher Education, seeks to encourage Saudis to enter medical professions such as medicine, nursing, pharmacy and other health majors (Alamri et al., 2006).

In 1996, Saudi nurses comprised only 9 % of the total nurse workforce (Jackson & Gary, 1991). By 1999, it had increased slightly to 17 % (Marrone, 1999). In 2004, the fraction increased dramatically to 35 %, with further progress annually leading to 51.9% of the nursing work force in 2011 being Saudi nurses. (MOH, 2011). The total number of nurses in Saudi Arabia is now 55,429.

The Nursing Board was established in 2002 as a professional regulatory board under the supervision of the Saudi Council of Health Specialties. The aims of the Nursing Board are to: define the nursing profession and its members; determine the scope of practice; develop educational and ethical competency standards and practice them; and establish accountability systems and credentialing processes (Abu-Zinadah, 2006).

1.7 Rationale of the Study

Although there is a large volume of literature on nurses' knowledge and attitudes regarding pain management (Gunningberg & Idvall, 2007; Matthews & Malcolm, 2007; Vallerand et al., 2007; Young et al., 2006), these studies have generally been conducted in Western countries. Only a few studies have investigated this subject in the context of the Middle East region, where Saudi Arabia is located (Yava et al., 2013; Rahimi-Madizeh, Tavakol & Dennick, 2010; Yildirim, Cicek & Uyar, 2008). The present study is significant because of the scarcity of data pertaining to this region in general, and to Hail region hospitals in particular. Therefore, this study's findings will contribute new information concerning nurses' knowledge and attitudes towards pain management in an Islamic society. As the study is conducted in Hail region hospitals and hence involves nurses from different cultures, it will provide data on how different demographics affect the delivery of effective pain management.

The findings will complement the existing non-Saudi data on the subject. This project identifies additional barriers to effective pain management, and accordingly proposes further areas of research and recommendations for changes to policies, practices, and education and training for nurses.

1.8 Purpose of the Study

The purpose of this explorative, descriptive study is to determine nurses' knowledge and attitudes towards pain management and to identify possible barriers to achieving optimal pain management in Hail region hospitals in Saudi Arabia. A central concept for examination relates to how nurses from different cultures affect the delivery of effective pain management within a largely Muslim population.

1.9 Research Questions

The research questions are:

- What knowledge and attitudes do nurses hold regarding pain management in Hail region hospitals?
- What are the barriers to achieving optimal pain management as perceived by nurses working in Hail region hospitals?
- What demographic and cultural factors affect the delivery of effective pain management?

1.10 Organisation of the Thesis

This thesis is structured in six chapters. Chapter 1 is the introduction, which comprises the background and definition of pain, a statement of the problem, an overview of Saudi Arabia and its health system, the justification and purpose of the study, and the research questions. Chapter 2 contains a review of the available literature related to the study and the theoretical framework. Chapter 3 sets out the methodology, setting, sample and methods of data collection and analysis. Chapters 4 and 5 outline the quantitative and qualitative results. Chapter 6 discusses the findings of the study. Chapter 7 presents the conclusion, recommendations for future work and limitations of the study.

Chapter 2: Literature Review

2.1 Introduction

This chapter presents a review of the literature related to the subject of pain and is presented in three parts. The first part provides a contextual background for this study, including the prevalence of pain, types of pain and theories on the physiology of pain, pain assessment and management, nurses' roles in pain management, under-treatment and barriers to effective pain management. The second part reviews the existing literature on nurses' knowledge and attitudes regarding pain. The third part considers culture and pain, and it introduces Leininger's theoretical model, which is reputed to be useful when considering pain in a cultural context. This literature will be further considered in the discussion chapter.

2.2 Part 1: Contemporary Literature on the Human Experience of Pain

Pain is a common human experience, and it is the most notorious symptom presented by patients seeking medical assistance (Rosdahl & Kowalski, 2008). Therefore, it is considered a major health burden (Argoff &Fine, 2010). The phenomenon of pain serves a physiological function and warns humans about the actual condition of their bodies (Chaudhari &Feaver, 2011). Pain is identified by the American National Chronic Pain Outreach Association to be a reactive indicator of an injury (Foster et al., 2008). Thus, pain serves as a clinically vital indicator not only for diagnosing and assessing a disease, but also as a warning to prevent possible damage. Moreover, it indicates the limits that should not be exceeded (Brennan et al., 2007).

As it is a personal experience, there is no definite objective means for the quantification of pain. Any discussion of pain must consider all of its different aspects, such as physical, emotional and mental (Shannon, 2011). Moreover, the quality of pain assessment differs depending on various factors, including the aim of the assessment, the environment, patients' health status and healthcare professionals. Therefore, no single strategy can be considered suitable for all patients in all circumstances.

Although pain is a helpful indicator of possible disease, injury or danger, it interferes with the internal balance of a person and consequently affects the quality of a person's life (Vitor et al., 2008). When pain is persistent in nature, it may result in transmission to the nervous system, leading to intense phases that can prolong and aggravate pain (Foster et al., 2008). For instance, when surgical pain is not managed effectively, the outcome might be a longer stay in hospital, a longer cycle of recovery and higher medical expenses (Wells, Pasero & McCaffery, 2008;Wu, Naqibuddin & Rowlingson, 2003; Young et al., 2008). Persistent pain can lead to more critical physical predicaments, such as the direct or indirect suppression of an individual's immune system, which protects the body against harmful infections, or even tumours (Foster et al., 2008).

The occurrence of pain entails two elements: 1) sensory-selective, which provides knowledge on the position, type and concentration of stimuli; and 2) emotional-motivating, which represents behavioural reactions to pain. The sensory-selective element is characterised by a withdrawal impulse or fight-escaping response (Almeida, Roizenblatt &Tufik, 2004).

Pain is a common phenomenon of human existence and can be experienced by people of all ages, cultures and social status (Gregory &Haigh, 2008). It can significantly affect not only the sufferer's life, but also those of his or her family and friends. Providing relief from pain is

considered one of the most basic responsibilities of medical personnel, and it is a fundamental human right (Brennan et al., 2007). However, there is still minimal current information on the worldwide prevalence and outcomes of acute pain. This is partly due to the perception that pain is a symptom of a probable disease or underlying injury, and also due to the nature of pain itself, which is generally not documented in national statistics (Helms, Quan, Herfindal & Gourley, 2006).

2.2.1 Types of Pain

As previously mentioned, pain is the body's natural alarm system to injury or malfunction. This alarm alerts the individual to injury and makes him or her stop a harmful behaviour or seek medical attention if needed (American Pain Foundation, 2009). Pain assessment was considered extremely important by the Joint Commission for Accredited Hospitals (JCAHO) that it has established standards for its assessment calling it the fifth vital sign. There are different experiences and intensities of pain. Pain can be categorised based on its duration such as acute or chronic. Pain may occur for less than three months and thus be classified as 'acute', whereas pain that lasts longer than three months is classified as 'chronic'. In addition, the cause of pain can also be the basis for its classification. For example, pain caused by damage to tissue is classified as 'nociceptive', while pain caused by nerve damage or an ailment is called 'neuropathic pain' (Macintyre &Schug, 2007).

A proper understanding of the different types of pain, their causes and mechanisms can lead to the selection of adequate pain treatment strategies. Effective pain management can therefore be beneficial in reducing the effects of pain. This also entails the effective management of acute pain in its initial phase in order to prevent it from turning into chronic pain.

2.2.1.1 Acute Pain.

Pain is the defensive reaction of the body, warning the system of an imminent or actual tissue injury. In this case, it stimulates synchronised reflexes and behavioural reactions that are intended to control the tissue damage within its boundaries (Woolf, 2004). This type of pain is also referred to as 'acute pain' (Janssen, 2002), and it is the most frequent type of pain experienced by people throughout the world. It is an important component and indicator of injury, medical operations, childbirth and acute illness. In fact, acute pain is the leading cause of more than two-thirds of patients' visits to hospitals (Cordell et al., 2002). Acute pain can also result from surgery: Sommer et al. (2008) assessed 1,490 surgical patients in the Netherlands and found that 41 % suffered from moderate to severe pain on the first day of the surgery. On the fourth day after surgery, 15 % of patients were still suffering from moderate to severe pain.

Generally, acute pain is mostly nociceptive, but it can also be neuropathic. The most frequent causes of acute pain include shock, childbirth, surgery, postoperative and acute disease conditions. Acute pain usually accompanies serious events, such as injury, along with other involuntary reactions, which may include muscle contractions (Jänig, 2012). Nonetheless, the stress hormone reaction induced by acute damage can also result in negative physiological and emotional outcomes (Solowiej, Mason & Upton, 2009). Moreover, the sensation caused by a heart attack or the pain experienced after an operation may result in a succession of involuntary reactions that hinder the functioning of a patient's essential body systems— particularly the heart and lungs (Symes, 2011). The activity of the sympathetic nervous system (SNS) produces symptoms such as increased pulse rate and arterial blood pressure (Vallerand, Reily-Doucet, Hasenau&Templin, 2004). With the initiation of medication and treatment, acute pain usually diminishes and the restorative process begins (Strassels, 2008).

2.2.1.2 Chronic Pain.

When the feeling of pain continues longer than the acceptable period needed for the treatment of an ailment, or when it persists beyond the normal period of a critical ailment, the condition is regarded as 'chronic pain'. In a survey conducted on patients who suffered from postoperative pain, a high-level incidence of chronic postoperative pain was ascertained in subjects whose acute postoperative pains were ineffectively managed (Goldstein et al., 2004). This highlights the importance of proper management of this type of acute pain to minimise the occurrence of complications, including the progression to chronic pain (Hampton, 2005). However, despite significant advances in the research and management of pain, large numbers of people continue to suffer due to ineffective pain treatment. The suffering worsens when pain is experienced by people who face catastrophes, accidents or violence. Unmanaged acute pain not only leads to more distress and suffering, but it can also result in other unwanted consequences, including delayed recovery, increased chance of morbidity, extended hospitalisation and the hazard of developing chronic pain (Macrae, 2008). Even in developed countries, inappropriate assessment and ineffective treatment of pain still occurs (Benhamou et al., 2008). Researchers have indicated that an enhanced awareness of the nature of acute pain can significantly improve its clinical management and could assist in determining strategies to directly improve the understanding and management of the physiopathology of particular pain conditions (Macintyre et al., 2010).

Breivik et al. (2006) conducted a telephone survey in 15 European countries and Israel to explore the prevalence, severity, treatment and effect of chronic pain on patients. Interviews with 4,839 participants showed that 66 % had moderate pain, while the rest (34 %) had severe pain. In 46 % of patients, the pain was persistent, while the rest (54 %) had discontinuous

pain. Those who had suffered for 2–15 years comprised 59 % of the participants. The other devastating effects resulting from chronic pain included depression (21 %), inability to work outside the home (61 %), loss of employment (19 %) and having to change jobs (13 %). Moreover, at the time of the survey, one-third of participants were not being treated and were still suffering from chronic pain (Breivik et al., 2006). In addition, surveys of diverse households showed that over one-third of people suffered from chronic pain, with a 36 % occurrence rate in Europe and 43 % in the US (International Association for the Study of Pain & European Federation of IASP Chapters, 2005). Further, pain prevalence increases among people with advancing age, especially in women engaged in physically demanding work.

Leaving a disease untreated can result in chronic pain, and this type of pain may persist even after recovery. Chronic pain associated with a disease condition may subside when the primary disease is cured (McLean, Clauw, Abelson &Liberzon, 2005). Chronic pain may have a debilitating effect on patients, significantly influencing their capacity to carry out dayto-day activities. Untreated chronic pain can not only result in continued, unnecessary suffering for the patient, but it can also lead to excessive medical expenses (Disorbio, Bruns & Barolat, 2006).

2.2.1.3 Nociceptive Pain.

Pain can be classified according to its cause. The outcome of continued chemical and involuntary impulses of nociceptors is classified as 'nociceptive pain'. When activated, this type of mechanism can send out warning indicators to the brain, thus causing the pain—for example, as experienced by individuals suffering from cancer (Millan, 1999; Fein, 2012). Nociceptive pain is divided into visceral and somatic pain. Visceral pain is a common pain that affects internal organs such as ureters (ureteral colic), urinary bladder (bladder

distention), appendix (appendicitis), kidney (renal colic), prostrate gland (prostatitis), heart (myocardial infarction) etc. Somatic pain is the most common pain that affect patients with cancer or bone metastases (Gerwin, 2002).

2.2.1.4 Psychogenic Pain.

'Psychogenic pain' also called 'psychalgia' is caused by psychological factors and is commonly experienced by individuals suffering from depression or anxiety (Ganzberg, 2010). However, real psychogenic pain is very rare, but often physicians do not carry out an accurate assessment of the pain leading to misdiagnosis of it as of psychogenic origin (Winterowd, Beck & Gruener, 2003).

2.2.2 Theories of Pain

For a long time, two generally accepted theories—namely, the 'specificity theory' and the 'pattern theory'—were used to explain the phenomenon of pain. More recent research has provided insights into the nature and causes of pain. As a result, the 'Gate Control Theory of Pain' was developed.

2.2.2.1 Specificity Theory.

René Descartes, a philosopher and mathematician, put forward the specificity theory in the seventeenth century. He described pain as the movement of a particular group of perimetric nerve tissues through the spinal column up to the central area of pain, or the pineal gland in the forebrain (Wozniak, 1992). This theory posits that the pain and touch indicators on the skin are connected to the focal point of pain in the brain. Receptors transmit the feeling of
pain straight to the brain, and any resulting sensation is understood as a mere response to the initial impulse of pain. The specificity theory was broadly accepted for several years; however, the theory is biologically based and does not acknowledge any psychological elements in the occurrence of pain. This theory does not consider the phenomenon of pain when the organic basis for pain is absent. For instance, consider an athlete who continues to play, unaware of an acquired injury, until the end of the game. In this situation, although the organic basis of pain is present, the sensation is not experienced until the individual focuses attention on the painful part of the body.

2.2.2.2 Pattern Theory.

The pattern theory of pain was developed in 1894 by German neurologist Alfred Goldschneider. He argued that there was no single structure for recognising pain, and that pain receptors are distributed with those for other sensations, such as touch, throughout the body (Melzack & Wall, 2008). According to the pattern theory, individuals experience pain when particular patterns of neural movement occur—for instance, when certain forms of activity reach extremely high levels in the brain. This theory suggests that these patterns emerge only with strong stimulation, as different intensities of stimulus give rise to varied patterns of sensory activity. This can be seen in the different sensations experienced when one is hit hard (which is painful), as opposed to being merely touched (which is not painful).

2.2.2.3 Gate Control Theory of Pain.

A more recent development is the Gate Control Theory of pain proposed by Melzack and Wall (1965). This posits the perception of an entryway that exists in the main nervous system. This in turn allows or disallows the transmission of pain signals all the way to the brain. A cluster of nerve cells, identified as the 'substantia gelatinosa', exists within the spinal cord. Similar to a gate system, the substantia gelatinosa functions as a 'gate keeper'; it decides when a sensory message is allowed or not allowed to make contact with the brain (Gilman &Newman, 1992). The theory suggests that the gate serves a significant function in the pain management of the main nervous system. The pain messages that cross the gate arouse the carrying cells of the dorsal horn located in the spinal cord, which in turn aid in conveying the pain messages to the brain (Buxton, 1999).

The Gate Control Theory asserts that pain messages are transmitted to the brain through stimuli that are received based on their phase and intensity. This also seems to have a bearing on the location where they enter. The Gate Control Theory is also based on the notion that both physical and psychological elements control the manner in which the brain translates pain and the ensuing reaction. Most individuals who experience pain observe that it worsens if they are distressed and lessens when they shift their focus to something that requires concentration or that is pleasurable (McCaffrey, Frock &Garguilo, 2003).

The Gate Control Theory of pain claims that painless impulses can hinder painful impulses (Buxton, 1999; Coon & Mitterer, 2008). The perception of pain relies on whether the dominant message ascends to transmit the message of pain or descends to restrain the painful impulse (Watt-Watson, 1999). The implications of the Gate Control Theory explain the efficacy of pain relievers and other counter-irritants in modifying the sensation of pain. The

Gate Control Theory appears to be valid. Its development has resulted in considerable progress worldwide in relation to pain research (Porth, 2004).

2.2.3 Prevalence of Pain

Despite the significant advances in the understanding of pain achieved in the past 20 years, it appears that the provision of treatment is still inadequate. The considerable number of pain sufferers at any given time, as well as its effects, characterise pain as a common health concern (Perry, Nicholas &Middleton, 2010). However, it appears that pain and its consequences do not receive due attention. Among the likely reasons for this state of affairs, the most critical is the confusion associated with its occurrence. Although it is a challenge to quantify the epidemiology of pain worldwide due to the ambiguous nature of the phenomenon, there is a degree of doubt regarding whether its occurrence is too high. The US National Centre for Health Statistics (2006) estimated the prevalence of chronic pain to be approximately 20–25 % of all individuals worldwide. In fact, the World Health Organization has estimated that one in 10 adults is affected by chronic pain symptoms annually (NCHS, 2006). Moreover, apart from the pervasiveness and occurrence of chronic pain, the seriousness of pain and the extent of its associated disabilities are regarded as major elements in the assessment of the physical problem.

Basic healthcare settings in the US, Europe, Asia and Africa have reported on the pervasiveness of persistent pain, estimating it to be as high as 10–25 %. The incidence of pain in the US is estimated to be 12–25 %, while a pain frequency of 20 % has been recorded in Europe (Breivik, Collett, Ventafridda, Cohen &Gallacher, 2006). Even in affluent environments, moderate to severe pain is prevalent at a rate of 10–25 % (Breivik et al., 2009). Meanwhile, as assessed through the impairment of daily functions, the combination of

persistent pain and simultaneous psychological ailments lead to critical disadvantages (Foley, 2011).

A systematic review conducted in the US reported a higher prevalence of pain among cancer patients, ranging from 14–100 %. Approximately 70 % of patients undergoing active therapy were ascertained to have severe pain (Christo & Mazloomdoost, 2008). These statistics confirm the severity of the phenomenon of pain for millions of individuals worldwide. They also confirm that pain is an inevitable actuality of life (Taylor, 2007).

For the above reasons, pain has become a subject of intense global research. A study conducted by Tsang et al. (2008) aimed to ascertain the frequency of typical chronic pain cases of headaches, arthritis and back pains by age and gender, and their relation to both depression and anxiety symptoms in 10 developed and seven developing countries. The study drew on data from 18 general adult population surveys using a common survey questionnaire (N = 42,249). The results indicated that the frequency of chronic pain cases for the preceding year (2007) was 37.3 % in developed countries and 41.1 % in developing countries. Among participants aged 65 and above, females showed a noticeably greater susceptibility to chronic pain. The study determined a positive correlation between chronic pain and depressive and anxiety symptoms in both developing and developed countries (Tsang et al., 2008).

Based on earlier surveys in Western and Central Europe, prevalence rates showed an increasing annual trend of 17–29 %. These statistics appear to suggest that the incidence of chronic pain may be rising globally. If the 17 countries (Colombia, Mexico, United States, Belgium, France, Germany, Italy, Netherlands, Spain, Ukraine, Israel, Lebanon, Nigeria, South Africa, Japan, People's Republic of China: Beijing, Shanghai and New Zealand) observed are regarded as representative of the global population, it would be logical to

conclude that chronic pain is prevalent internationally (Tsang et al., 2008). Other studies (e.g. Cicero et al., 2009) investigated the relationship between gender, age and the occurrence of both acute and chronic pain, as well as usage of healthcare provisions, health conditions and expertise of the health institution. The Cicero's study determined the effect of chronic pain on a group, as well as the effect on those who were chronic opioid users within that group.

A study conducted in The United Kingdom by Wand et al. (2004) to assess the consequences of pain, ailments, moods and health conditions and found that the timing of the intervention influenced the type of psychosocial outcome that was achieved. Meanwhile, a randomised controlled trial with 67 subjects who had venous leg ulcers and were treated in a public nursing institution confirmed that an all-around care strategy involving nutrition, social involvement and mental reactant, including treatment of the ailment and awareness of the treatment management, delivers more effective results compared to nursing home treatments (Edwards et al., 2009).

A recent study on the prevalence of pain among 1,134 adults in the United Arab Emirates, Bahrain, Oman and Kuwait who were suffering from chronic back pain for more than three months found that neuropathic pain was present in more than half (55.4 %) of the study's participants (El Sissi et al., 2010). Additional outcomes of the study indicated that only 11.4 % of the patients who experienced pain actually received effective treatment, which led the authors to advocate for appropriate treatment for pain.

2.2.4 Assessment of Pain

The objective of pain assessment is to ascertain the trigger points of pain, the effect on the quality of life, the most appropriate therapies and the efficacy of existing treatments. Many researchers have noted that the under-treatment of pain in US hospitals is mostly associated with the failure of healthcare professionals in assessing pain and providing pain relief (Breivik et al., 2008). Current initiatives to promote awareness have increased. Moreover, it has also been recognised that the assessment of pain should be consistently practised by healthcare professionals as a fifth vital sign and an essential human indicator (Lorenz et al., 2009) to ensure that there is increased responsibility for pain assessment and to promote awareness of pain as a significant health concern (Igumbor, Puoane, Gansky &Plesh, 2011).

An appropriate assessment of pain should include information on the patient's age, history of previous medical and surgical conditions, medications used, and physical and cognitive status. The pain assessment usually commences with enquiries regarding the patient's pain history, including a description of the pain, its initial occurrence, duration and the regularity of the sensation. Other important details that should be obtained are the factors that may worsen or lessen the effects of pain on the individual's daily life, and the usage of all prescribed medication and over-the-counter medicines and supplements. The patient's records should also include information on the possible social, ethnic and spiritual factors contributing to the pain. Ultimately, a precise physical assessment should be performed to pinpoint the probable causes of pain (Breivik et al., 2008). Another important factor of pain management that is often neglected is the need for persistent reassessment (Hader &Guy, 2004), which is crucial because it determines whether the pain management plan is effective or whether adjustments need to be made. After every intervention, a reassessment should be conducted and its results recorded. Healthcare professionals need to be aware that patients sometimes do not ask for

analgesia unless they are experiencing severe pain. In most cases, this makes the process of pain management difficult. Thus, asking patients if they are experiencing pain is a valuable strategy for starting pain evaluation and for informing the patient regarding the pain control process (Hader &Guy, 2004).

The assessment of pain is a crucial tool in providing an effective pain treatment. However, defects in pain assessment have been found in several settings (Brawley, Smith &Kirch, 2009). Several proposals and guidelines have been offered to explain the components of effective pain management. However, many guidelines are not applicable in the actual management of acute pain. For instance, nurses who attend to patients who are experiencing acute pain need to choose the appropriate elements of assessment for the existing clinical situation and to be aware of the patient's beliefs, level of knowledge, attitudes and previous experiences with pain. There should be a reassessment of pain after every intervention to analyse the results and verify whether the intervention was effective or not and whether further changes in the treatment are necessary.

The Joint Commission's core principles (Joint Commission Resources, 2010) state that patients are entitled to receive a proper assessment and management of their pain. However, pain is a sensation experienced by the patient; hence, it is challenging for the healthcare provider to assess and manage it. Therefore, it is crucial for all nurses to have the most comprehensive knowledge required to provide pain relief to patients.

An extensive pain assessment strategy outline is provided by the guidelines and serves as a useful guide during surgical operations. Part of the pain evaluation involves obtaining the patient's history of pain, where the nurse determines the patient's behaviour, including his or her cultural background, intellectual capacity and preceding encounters with pain (Powell,

Downing, Ddungu & Mwangi-Powell, 2009). The patient's anticipation, as well as the expectations of family members regarding the management of postoperative pain, could result in impractical assumptions that can be dealt with before surgery. The complete account of the occurrence of pain serves as the basis for planning appropriate pain management subsequent to surgery, which is executed together with the patient, the family and the clinicians.

2.2.4.1 Assessment Tools.

The occurrence of pain is a complicated phenomenon. It is influenced by a person's earlier experiences of pain, as well as psychosocial factors and the cognitive interpretation of the pain messages received by the brain (Newton-John, 2005). To aid in pain analysis and ascertain the efficacy of any treatment, a pain assessment tool should be utilised. In this regard, the assessment tool should be uncomplicated in order to provide a good understanding to both the healthcare provider and the patient, and it should be a justifiable and dependable gauge of pain.

The Joint Commission on Accreditation of Healthcare Organizations (JCAHO) established standards for pain evaluation and management in accordance with Acute Pain Clinical Practice Guideline (Gordon et al., 2008). Hospitals are required to choose and utilise consistent pain evaluation tools in all departments. The uniform usage of the pain evaluation tools has been set as the standard for verifying the pain experienced by patients, and for recording and measuring pain evaluations periodically. Four widely used tools for evaluating acute and chronic pain are: (1) the numerical rating scale, (2) the Wong–Baker Faces Scale, (3) the Verbal Graphic Rating Scale, and (4) the Pain Aid Scale (Daniels &Nicoll, 2011). The measurements are straightforward, and instructions for using them are given in various languages. According to the guidelines, the patient must be informed regarding the usage of

the evaluation tools, and the same tool should be consistently used every time. This will ensure the precise evaluation and recording of pain. Maio et al. (2002) established the necessity for the consistent usage of pain evaluation tools to accurately evaluate pain.

An exploratory study was conducted by Bouvette, Fothergill-Bourbonnais and Perreault (2002) to determine the feasibility of implementing the 'Pain and Symptom Assessment Record' (PSAR) in a variety of settings. PSAR is a different tool that is used to assess patients' pain based on symptoms only. The sample of this study comprised 180 nurses at 12 sites. By evaluating specific groups and utilising charts, the recorded information indicated that pain evaluation occurred 93 % of the time. With enhanced accuracy in the records of pain and indicator management, a high level of patient satisfaction occurred with the treatment of pain. The researchers concluded that patient awareness was a vital component in verifying whether pain treatment providers could carry out an efficient evaluation. After proper information was provided to the patient, the pain was gauged and managed better (Bouvette et al., 2002). After surgical operations, it is recommended that pain evaluation be conducted in a concise and straightforward manner (Abdalrahim, 2009). The treatment criteria, including the type and dosage of medication, should consider the severity of pain. Every evaluation of pain needs to include these types of measurements. In this case, various pain-complexity evaluation methods have been established and utilised. For instance, a mere one-dimensional tool with a 0–10 numerical rating scale (where zero signifies 'no pain' and 10 'worst pain') only measures the severity of pain that is appropriate for an emergency room or a recovery room. Healthcare professionals in other circumstances should be persuaded to gauge and document the trend and type of pain in conjunction with pain intensity. A nurse who evaluates the efficacy of provided pain management may need a multidimensional tool that investigates the effects and emotional consequences of the pain being experienced by the patient. In this

circumstance, the severity of pain and the McGill Pain Questionnaire may be utilised (Melzack, 1975).

In 1992, Ambuel et al. established the earliest Comfort Behaviour Scale to be utilised in evaluating the pain experiences of paediatric patients in the intensive care unit (ICU). Subsequently, it has been customised for different settings and uses. The Comfort Behaviour Scale is particularly used to evaluate the attitudes of patients who do not have sufficient capacity to react to pain evaluation scales. The components of the Comfort Behaviour Scale are used to assess a patient's attentiveness, composure, respiratory function, complaint levels, physical activities, muscular movement, facial reactions and contrast of current heart rate and arterial blood pressure with the baseline gauges (van Dijk, Peters, van Deventer & Tibboel, 2005). A range of 1–5 is assigned for each component. A visual analogue scale is utilised to demonstrate the level of a patient's pain. In utilising the Comfort Behaviour Scale, a form is used to record the details of the treatment and the condition of the patient. This enables both the nurse and the family to identify suitable treatment options and the response required to the total evaluation ratings on the scale. In a standard evaluation, pre- and post-medications are recorded on the form as well. A standard evaluation can be performed every two hours after surgery for the initial 24 hours. Further evaluations and records of the patient's responses to analgesic medication can be conducted before or after pain management procedures (van Dijk, Peters, van Deventer & Tibboel, 2005).

The pain assessment tool that is chosen should be used to check the pain and the response to the interventions. It is important to take into account a patient's mental, physical and emotional health, as well as his or her cognitive condition, before a nurse selects an assessment tool (Jamison, Serraillier &Michna, 2011). The patient's preference must also be considered, and it should be a joint decision between the care provider and the patient. It is

advisable that this decision be taken in the pre-operative phase so that the patient becomes familiar with the instrument's scale. Patients who are awake and alert but unable to respond verbally may point to a picture of a face or a number to indicate their distress. However, the caretaker should not solely rely on the tool's measurement of pain. Instead, they must consider the patient's vital signs and combine the findings with their clinical judgment (Gordon et al., 2005).

2.2.5 Management of Pain

The objectives of successful pain management are to provide pain relief, restore functionality and improve quality of life for the patient. The aim is to intervene with the minimum use of medication. The first step in pain management is the identification of the type of pain. This will help in the diagnosis and in developing an appropriate pain management plan to suit the patient's needs (Hader &Guy, 2004). The present focus is on the application of a suitable schedule for the administration of analgesic medications, providing physical comfort evaluations and applying appropriate pain management to aid in relieving the contributing factors to pain, such as anxiety, fear, vulnerability, depression and insufficient coping strategies. Acute pain requires interventions to eliminate the discomfort by removing its cause, treating it with analgesics and other physical or behavioural medical approaches (Wells et al., 2008). In most instances, treatment with analgesic drugs is the initial approach in the treatment of acute pain.

According to Wells et al.,(2008), patients and their families can experience physical, mental and emotional suffering when the patient's pain is not managed adequately. In addition, persistent unmanaged pain stimulates the pituitary-adrenal alignment, which can weaken the body's immune system, leading to postoperative infections and compromising the woundhealing process. Supportive stimulation can have adverse effects on the gastrointestinal, cardiovascular and renal systems, making patients susceptible to conditions such as intestinal obstruction and ischemic heart disease (Wells et al., 2008). Additionally, of particular significance to nursing care, untreated pain may reduce the patient's ability to move, leading to more clinical impediments such as sudden blockage in a lung artery, deep-vein thrombosis and pneumonia. Postoperative complications associated with insufficient pain management adversely affect the patient's well-being and the hospital's operations due to prolonged hospital stays and unnecessary readmissions, which lead to increased medical expenses (Wells et al., 2008).

Persistent, unmanaged pain also affects the mental and emotional states of patients and their families. Typical psychological reactions to pain include anxiety and dejection. The incapacity to relieve pain may create a sense of vulnerability and even depression, which may lead to a more chronic depressive state for the patient (Klauenberg et al., 2008). As a result, patients who have not been given sufficient pain treatment may be hesitant to ask for medical assistance for their other health concerns. In addition, there can be legal consequences for healthcare providers as a result of the inability to manage a patient's pain. The present benchmark for the treatment of pain-particularly the criterion established by the JCAHOrequires that pain should be treated without delay (JCAHO, 2001). The establishment of professional standards for effective pain treatment decreases the chance of legal action associated with shortcomings in pain management (Wells et al., 2008), thus ensuring that clinicians and institutions act responsibly. As members of healthcare teams accountable for treating pain, nurses may also be liable for legal action. Patients' satisfaction is highly associated with their experiences of pain management. Greater pain levels are connected with unsatisfactory experiences of treatment in ambulatory circumstances (Bair et al., 2007). Evaluation reports on medical institutions are becoming more widespread, and it seems that

performance that is relevant to the treatment of pain is one of the important considerations in such reports.

Actions to manage pain should consider aetiology, physiopathology and repercussions; they should aim to eliminate the causative factors and treat the patient with analgesic and antiinflammatory medications. Indeed, most analgesics also have anti-inflammatory properties (Mendell & Sahenk, 2003). Such treatment should be complemented with other approaches, such as physiotherapy, psychotherapy and rehabilitation (Catalano & Hardin, 2004). In certain instances, it may be necessary to resort to anaesthetic or neurosurgical procedures when the pain is resistant to other forms of treatment.

2.2.5.1 Under-treatment of Pain

The problem of under-treatment of pain was first recognised in the remarkable research conducted in 1973 by Marks and Sachar. They interviewed 37 medical in-patients who had been treated with narcotic analgesics for pain and found that 73 % of surgical or medical patients suffered from moderate to severe pain. The American Geriatrics Society (AGS) indicated that approximately 45–80 % of elderly patients in hospitals suffer from under-treatment of pain (AGS, 2002). In 2003, 30 years after the study by Marks and Sachar, Apfelbaum et al. conducted a study in the USA that involved 250 adults who had undergone surgical procedures. The results showed that 80 % of hospitalised patients suffered acute, postoperative pain, and that 86 % of these patients experienced mild to severe pain (Wells et al., 2008). These and other studies suggest that patients suffering from mild to acute pain have only a 50 % chance of receiving sufficient relief.

The availability of Acute Pain Services (APS) with the services of advanced practice nurses specialising in pain management are limited to a few major teaching hospitals, and patients often have to endure long waiting periods for admission to one of these units (Jovey, 2008). There are greater limitations in resources in medical units because there is an assumption that patients in medical units have their pain better managed compared to those in surgical units (Sawyer et al., 2010). There is a tendency to give post-surgical pain and trauma priority, while medical patients are given less consideration.

2.2.5.2 Consequences of Under-treatment of Pain

Under-treatment of pain has many physiological and psychological consequences. According to Wu, Naqibuddin and Rowlingson (2003), under-treatment of pain increases postoperative morbidity and delays recovery. Inadequately treating pain for a prolonged time—particularly acute pain—could result in chronic pain (Wells et al., 2008). According to Joshi and Ogunnaike (2005), unrelieved pain also has many psychological consequences, such as anxiety, depression, fear, anger and reduced patient satisfaction. Effective pain management is critical in preventing negative outcomes for patients, both during and after treatment. Further, it helps alleviate the degree of suffering that a patient may be experiencing at various stages of medical intervention (Vadivelu, Mitra & Narayan, 2010). Pain control strategies must be effectively administered for effective healing. For example, Valivelu et al. (2010) stated that poor pain management can result in other problems, such as poor wound healing and insomnia.

In addition to the physical, neurosensory dimension of pain, where an individual becomes physically incapacitated due to its effects, pain has an emotional dimension. It is a complex experience that affects the mind (i.e. the thought processes of an individual) and behaviour of an individual. Socially, pain can be a legitimising factor for isolating an individual where others view him or her as abnormal, giving society a rationale for ascribing the sick role to the individual (Arnstein & Marie, 2010).

In a state of social isolation, psychologically, the patient may experience feelings of rejection coupled with the experience of pain. If the health system is not congruent with its cultural background, then depression may appear and make the situation even worse. People with chronic pain have a very high tendency to develop psychiatric problems, which is normally in the form of mood or anxiety disorders (McWilliams, Cox & Enns, 2003). Depression is a catalyst for the development of further pain (Klauenberg et al., 2008). Excruciating pain in an individual and a lack of enough social support worsens the social situation of both healthcare givers and patients. While people in some cultures may visibly express their pain, others will withdraw inwardly and stoically accept pain, which will only worsen their emotional state as the pain becomes unbearable. This may lead to poor relationships with others, depending on how much help and attention they require. Thus, they become emotional and psychological burdens to family members and society, who may be physically drained when they are not competent to handle patients' pain (Poole, White, Blake, Murphy & Bramwell, 2009).

Financially, poor pain management may lead to the inappropriate use of resources. For example, the fact that pain has been treated does not necessarily mean that its root cause has been eliminated.

Gordon et al. (2002) found that patients' satisfaction with the care they received from healthcare givers was at a higher level when their pain was well managed. It therefore follows that poor pain management leads to a feeling of dissatisfaction regarding services and employees, and patients feel that they have not received value for their money. Patient

satisfaction becomes an integral factor in evaluating the pain management strategies that nurses and doctors use to control and manage pain in patients (Hanna, González-Fernández, Barrett, Williams & Pronovost, 2012). In this study, they conducted a survey of 4,349 adult patients who were admitted to any surgical unit over an 18-month period.

2.2.6 Nurses' Roles in Pain Assessment and Management

With regard to pain management, the involvement of nurses is highly significant, as they are responsible for pain-relieving interventions, including administering medication, evaluating procedures and implementing any required changes (Twycross, 2002). Nurses should have knowledge about the assessment and management of pain because they play an important role in clinical settings. Nurses can provide appropriate pain medication consistently and without bias towards or against any particular pain condition. Establishing accountability for pain management and creating regular functional comfort goals will decrease the interference of personal attitudes of healthcare providers regarding pain (Pasero &McCaffrey, 2004).

Patients expect nurses to be pivotal in caring for their pain. They place trust in nurses as the primary carers for pain, and patients should co-operate with nurses and provide the necessary information for an adequate assessment. Partnerships between nurses and patients can improve pain relief outcomes. Therefore, by enhancing nurses' knowledge of the methods for assessing pain and how to obtain valuable information from patients, pain management can be improved and can assist in determining various choices of care (D'Arcy, 2007). However, knowledge regarding the use of pharmacological interventions is not enough, thus highlighting the importance of existing research on pain management (Textor &Porock, 2006).

2.2.7 Nurses' Misconceptions and Negative Attitudes Regarding Pain Assessment and Management

Nurses need to be fully focused and approach the task of pain assessment and management with the appropriate attitude, regardless of the patient's age or other considerations (Lui, So &Fong, 2008). Nurses' excessive reliance on their own subjective judgement has been shown to be the greatest limitation to effective pain management (Schafheutle, Cantrill & Noyce, 2001). A nurse should trust a patient's self-report, even if it seems to be incompatible with the point of view of the nurse or the patient's own nonverbal behaviour at first. This would help in assessing and managing the pain. Nurses' knowledge of pain assessment and interventions is an essential component in promoting positive patient outcomes. It is a healthy practice to assess and manage the pain for the benefit of both nurses and patients.

Liu, So and Fong (2008) examined the knowledge and attitudes of 370 ICU nurses from 16 hospitals in Hong Kong. The results showed lower-than-expected knowledge and attitudes, with an average Knowledge and Attitudes Survey Regarding Pain (KASRP) score of 47.7 %. There were some discrepancies in the answers to specific questions. For example, although 71 % of nurses thought that patients' self-reports were the most accurate indications of pain, only 1.4 % believed that patients never overestimated the degree of pain. In this case, nurses believed that patients who grimaced were suffering more. Moreover, 71 % of participants believed that patients should tolerate the least possible amount of pain; however, 64 % would advise patients to use non-pharmacological means alone instead of using them concurrently with pain medications (Liu et al., 2008). In this study, the nurses were shown to be deficient in knowledge of pain and to have misconceptions regarding pain management. For a long time, the general view was that nurses lacked adequate knowledge in pharmacology (Sawyer et al., 2010). For instance, despite the fact that morphine is a commonly used medication,

many nurses did not know of the duration, peak effect, ceiling effect or safe amounts to be administered. A recent descriptive and cross-sectional study conducted in Turkey by Yava et al. (2013) utilised the KASRP tool to explore nurses' knowledge and attitudes towards pain. The study sample comprised 246 nurses from different departments. The results showed that the overall mean score for participants was 39.65 %. The results also indicated that nurses' knowledge was inadequate and that they had negative attitudes towards pain management.

Generally, nurses seem to avoid administering opioids to older patients because they are uncomfortable with it (Gregory &Haigh, 2008) due to fears of overdosing and addiction (McCaffrey, Ferrell & Pasero, 2000). In addition to knowledge deficiencies, nurses' attitudes seem to stand in the way of carrying out proper pharmacological interventions, thereby leading to inconsistent and unreliable pain management. Further, there is a greater tendency for licensed practical nurses not to believe patients' pain reports and to under-document pain, and they are more hesitant to administer opioids than registered nurses.

It is equally important for nurses to ask questions and address patients' concerns (Registered Nurses Association of Ontario (RNAO), 2002). Further, some nurses and physicians may have doubts about a patient's description based on the latter's subjective assessment, particularly in situations where the patient belongs to a marginalised or under-privileged segment of society (Peter & Watt-Watson, 2002).

2.2.8 Barriers in Management of pain

Pain management is a complicated process, and it includes many variables that play an important role in the process. Pain can be inadequately treated because of a combination of cultural, societal, educational, political and religious constraints (Zuccaro et al., 2012).

Barriers that interfere with adequate pain management have been classified as problems related to nurses, patients and the healthcare system, as outlined below.

2.2.8.1 Nurse-related Barriers.

Poor assessment of pain and inadequate knowledge regarding pain have been identified as major barriers to adequate pain management (Bruera et al., 2005). Some studies suggest that nurses are not properly assessing patients. In this regard, research has shown a variety of contributing factors, such as nurses' disbelief of what patients say about their pain (Clarke & Iphofen, 2008). Nurses do not always ask patients about their pain and are not always able to assess it adequately (Watt-Watson, Stevens, Garfinkle, Steiner & Gallop, 2001). In one study, nurses said that their actions depended on how patients described their pain; however, many of them actually rely on how patients look rather than listening to them (Chang, Kim, Sjöström &Schwartz-Barcott, 2001).

In some instances, the barrier to effective nursing in pain management can result from a lack of cooperation from physicians. Van Niekerk and Martin (2003) utilised a survey to examine the barriers to providing optimal pain management. The subjects were a sample of 1,015 nurses in hospitals in Tasmania, Australia. The results showed that more than one-third of the participating nurses indicated a lack of cooperation from physicians as a barrier. In the nurses' opinion, physicians were not allowing adequate analgesic medication. In such situations, the patients' pain was inadequately treated (McCaffery, 2002). For instance, a study conducted by Sawyer et al. (2010) indicated that 50 % of medical patients were prescribed PRN, but only 14 % of these orders were administered to patients.

Several studies indicated a high nurse–patient ratio as an additional contributing factor, particularly because it affects patient education, which is an essential part of pain management (Van Niekerk &Martin, 2003; Elcigil et al., 2011). Education should extend beyond the patient and include the medical and nursing staff in training on standardised clinical pain guidelines, which will have a positive influence on pain management practices (Musclow, Sawhney &Watt-Watson, 2002). It is essential to adopt evidence-based approaches, standardised orders and protocols to ensure consistent standards of care in order to uphold the criteria of professional practice and improve patient care outcomes (Canadian Nurses Association, 2011).

Nurses provide pain education to patients during their stay in hospital. This helps patients to prevent and recognise the side effects of analgesics and enforce good pain management techniques. A lack of support from nurses' institutions and colleagues in developing knowledge, making decisions and effecting change can result in feelings of tension, learned helplessness and low self-efficacy (Wilson, 2007).

2.2.8.2 Patient-related Barriers.

Due to factors such as language, beliefs and cultural influences, as well as certain misconceptions regarding pain, not all patients are able to report their pain experience adequately (Huffman &Kunik, 2000). Further, many patients are reluctant to complain of pain unless specifically asked, or they may avoid reporting pain because of a fear of medication addiction (Erdek & Pronovost, 2004). Patients may not complain of pain because they want to be 'good' patients. Culture also affects patients' attitudes and motivation to cooperate. Some worry about being thought of as a nuisance, so they will not interrupt busy nurses, while others are afraid of not being taken seriously by staff or think they will not be believed.

Certain patient-related problems, such as economic, social and cultural barriers that prevent them from seeking health treatment, together with personal and societal beliefs, discourage patients from accessing pain management. Brennan et al. (2007) described the extent to which such defeatist views of pain are part of an individual's state and pervade belief systems. These perceptions about pain may result in patients struggling to attain credibility regarding their pain (Monsivais, 2011).

Culture can impact on the experience of pain at several levels. In some cultures, even when people are in pain, they would like to project themselves as 'good' patients, and they attempt to face pain stoically (Lasch et al., 2000). Such patients may under-report their pain to nurses for the fear of being judged as weak. Further, some people are reluctant to take opioid pain medications because of cultural taboos or fears about their use (Lovering, 2006). These patients would be more comfortable with familiar, culture-based remedies such as medicinal herbs or energy therapies (Cherniack et al., 2008). Such practices may not meet the approval of Western medical practitioners, thereby creating a conflict between the patient and the clinician, and a barrier to the latter's ability to help the patient.

A study conducted by Weech-Maldonado, Elliot, Schiller, Allyson and Hays (2012) used the Hospital Consumer Assessment of Healthcare Providers and Systems (HCAHPS) score system to examine the relationship between the cultural competency of a hospital and inpatients' expectations of healthcare. It was a comprehensive study involving 19,583 HCAHPS respondents from 66 hospitals throughout the US. The variables being studied included caregiver–patient communication, pain control, general hospital rating and willingness by patients to recommend the hospital. The findings led to the conclusion that those hospitals with greater cultural competence provided better experiences and outcomes for patients in their interactions with nurses and other hospital staff. Not surprisingly, the survey also

revealed that minorities, such as Hispanics, scored higher in the level of satisfaction with the hospital stay and care they received at hospitals with better cultural competence (Weech-Maldonado et al., 2012). These findings lend credit to the importance of nurses having cultural competence in their approach to pain management and control in patients.

2.2.8.3 Healthcare system-related Barriers.

Organisational factors can be a hindrance to pain management. For example, people who are kept in institutions such as jails, rehabilitation centres or-care facilities for lengthy periods have a higher probability of developing comorbid conditions connected with pain, which are likely to remain undiagnosed, and consequently left untreated (Baidawi et al., 2011; Baldridge & Andrasik, 2010). Additionally,, badly designed policies, deficiencies in healthcare delivery and poor accessibility to care and pain management facilities in distant locations are all obstacles in this context. Chen, Gelgor and Bajorek (2004) drew attention to distance, the shortage of professional personnel and rural cultural factors as the main reasons that prevent a rural population from receiving appropriate care services. The nurses ranked inadequate staffing—particularly at times of overcrowding with acutely ill patients—as the number one barrier to pain intervention. Other listed barriers include methods of health service provision and service funding. Concerns have often been expressed about the financial arrangements for funding pain care centres and how such funds are managed (Gatchel &Okifuji, 2006).

A study was conducted in Iran by Rejeh, Ahmadi, Mohammadi, Anoosheh and Kazemnejad (2008) to clarify Iranian nurses' perceptions of the barriers and facilitators influencing their management of postoperative pain. This qualitative study was based on interviews with 26 nurses. The barriers to effective postsurgical pain treatment as perceived by the nurses included their powerlessness in making decisions, the policies of the organisation (Coker et

al., 2010), physicians that lead the practice, time constraints (as a result of staff shortage), limited communication and interruption of pain management activities as a result of work overload (Rejeh et al., 2008).

A recent study conducted by Elcigil et al. (2011) utilising a self-report questionnaire to explore the barriers to pain management. The sample of the study included 114 nurses working in medical, oncology and surgery clinics. The findings indicated that the most commonly perceived barriers to pain management were system-related, such as the lack of psychosocial support services, patient-to-nurse ratio and inadequate time for nurses to engage in health education with patients (65 %). The most common barriers related to physicians were inadequate assessment and management of pain by doctors (63 %) and physicians' indifference (47 %) (Elcigil, Maltepe, Esrefgil & Mutafoglu, 2011). A lack of support from nurses' institutions and colleagues in developing knowledge, involvement in decision-making and effecting change can result in feelings of tension, helplessness and low self-efficacy (Wilson, 2007).

Advanced Practice Services (APS) has introduced important changes, including creating special categories of nurses such as clinical nurse specialists (CNSs) and nurse practitioners (NPs) who are authorised to prescribe.

2.3 Part 2: Critical Review of Nurses' Knowledge and Attitudes Regarding Pain: Search Strategy

This part explains the search strategy and the process of searching for related articles in databases and other sources. A review of the literature was conducted using online databases such as ProQuest, PubMed, CINAHL, EBSCO and Google Scholar. A keyword search of

ProQuest using 'nurses knowledge and attitudes regarding pain' returned 10,712 articles. The researcher limited the results by using keywords such as 'assessment', 'management' and 'culture', which reduced the number of results to 4,588. These were limited again to related subjects, articles in English, full-text articles and articles published within the last 11 years (2002–2013), which resulted in 25 articles. The aforementioned process was applied to the other databases, and the results from all databases were combined (see Table 2.1). Duplicate and unrelated articles were excluded. A total of 39 subject-related studies were selected and reviewed. Twenty of these articles used the KASRP tool (Appendix A). In addition, theses and books were used to obtain additional data and information for the literature review.

Database	Search Terms and Process				
	Nurses' knowledge and	AND	AND	AND	Relevant to Study
	attitudes regarding pain	Assessment	Management	Culture	(2002–2013)
PROQUEST	10,712	8,804	7,696	4,588	48
PUBMED	671	311	260	16	11
SCOPUS	57	27	26	2	2
CINAHL	46	20	20	2	2
Total					63
Included in Review					39
After Removal of Duplicates					24

Table 2.1: Search Terms and Process

2.3.1 Nurses' Knowledge and Attitudes Regarding Pain

It is important to note that having knowledge about pain and knowing how to deal with it are two different issues. Ferrell and McCaffrey (2008) warned that nurses lacked knowledge of pharmacology, as revealed by their answers to the pharmacology questions on the KASRP. The KASRP tool has been validated; therefore, the results of the above studies are given significant credence. Together, they point to a serious problem of knowledge inadequacy among nursing staff, which is prevalent around the world, regardless of a country's economic status. Lewthwaite et al. (2011) conducted a study of large, urban tertiary hospitals in Canada to explore registered nurses' knowledge and attitudes using the KASRP tool, including their knowledge of medicine. The sample included 761 nurses from a variety of clinical units. The study found that 49 % of participants scored 80 % or more on the KASRP, with a mean score of 79 %. The nurses had Bachelor degrees, although they were young and had limited work experience. In contrast, in Hong Kong, Liu et al. (2008) failed to find a significant correlation between educational level and pain management knowledge or attitudes. However, nurses with more professional clinical experience were able to apply their knowledge to daily practice and scored higher on the KASRP than those with less clinical experience (Liu et al., 2008). These authors have argued that a lack of formal pain education during nursing training was responsible for the nurses' low scores on the KASRP. Brunier et al. (1995) found a KASRP mean passing score of 41 %, while Lewthwaite et al. (2011) found a mean passing score of 79 %.

Since the introduction of the KASRP tool, it has been used extensively to explore the knowledge and attitudes of healthcare staff regarding pain management. Thus, it was an appropriate instrument to use in this current study of knowledge and attitudes towards pain management in Saudi Arabia. Lai et al. (2003) examined the knowledge of 1,797 nurses randomly drawn from nine hospitals throughout Taiwan. An overall correct response to the questions was found to be 50.5 %, indicating a substandard level of knowledge. Those who scored above average had a Bachelor of Science or higher degrees, had received pain education, had more work experience and had always worked in cancer wards. In a similar study by Tsai et al. (2007), nurse subjects (N = 249) were recruited from nine hospitals chosen by stratified sampling throughout Taiwan. Data were collected using the Nurses' Knowledge and Attitudes Survey—Taiwanese version, a scale to assess perceived barriers to pain management and a background information form. The overall average correct response rate for the knowledge scale was 49.2 %, with a range of 4.8–89.2 % for each survey question (Tsai et al., 2007). A major barrier to managing pain was identified by nurse participants as

the responsibility of caring for other acutely ill patients in addition to patients with pain. Knowledge of pain management had a significant negative relationship with perceived barriers to pain management and a significant positive relationship with the extent of clinical care experience and the total hours of prior pain management education (Tsai et al., 2007).

In another Taiwanese survey, Wang and Tsai (2010) examined the knowledge of 370 ICU nurses using the Nurses' Knowledge and Attitudes Survey—Taiwanese version. An overall correct response score of 53.4 % demonstrated poor knowledge of pain management, which was in accordance with the findings of the previous reports, indicating that the situation had not changed over time. Another study conducted in the Australian state of Tasmania by Van-Niekerk and Martin (2001) explored the knowledge of pain management practice among the state's nursing population. Based on the responses to a 29-item survey by 1,015 nurses, the authors concluded that the knowledge level was poor. Tse and Chan (2004) determined the knowledge level and attitudes regarding pain of 678 registered nurses working in three hospitals in Hong Kong. The KASRP questionnaire was used after being translated into Chinese. To ensure the contextual relevancy and consistency of the questionnaire, content validity and test–retest reliability tests were performed. The overall correct response averaged 44 %.

In a recent study conducted in Jordan, Al-Khawaldeh, Al-Hussaini and Darawad (2013) explored the knowledge and attitudes regarding pain management among baccalaureate nursing students. A sample of 240 students from three different nursing schools participated in the survey. The average correct response score of only 34.1 % (SD=9.9) provides further support for the widespread phenomenon of inadequate knowledge and attitudes among nurses in relation to pain management. Another recent study was conducted in Lebanon by Abdul Rahman, Abu-Saad and Noureddine (2013) to assess nurses' knowledge and attitudes

regarding pain. The total sample was 88 nurses working in a tertiary medical centre. Using the KASRP tool, the results showed that the mean score was 56.15 %, and only 3.4 % of the nurses obtained a passing score of 80 %. The results indicated poor knowledge regarding pain assessment and management.

A survey by Coulling (2005) included 82 nurses and doctors and found that a lack of adequate knowledge was the principal barrier to the management of postsurgical pain in patients. Results from other studies (Rieman & Gordon, 2007; Plaisance & Logan, 2006) reinforced the observation that the main constraint to pain relief treatment is healthcare professionals' lack of appropriate knowledge.

A study in the US examined both the knowledge of nurses in pain management and barriers to optimal pain management (Tapp & Kropp, 2005). Two studies explored the effects of a pain education program in altering the knowledge level and attitudes of nurses (Huth & Gregg, 2010; Howell, Butler, Vincent, Watt-Watson & Stearns, 2000). These studies were conducted in Mexico and Canada respectively, and they found that a pain intervention nurse education program significantly improved participants' skills in effective pain management. These results indirectly confirm the findings regarding nurses' lack of knowledge in pain management. It was based on responses to a survey by 23 registered nurses at an urban teaching hospital in Ohio, US. The average performance was below what it should be, with an overall score of 69.4 %. Although the nurses showed better knowledge regarding general pain assessment and management, their scores were severely reduced by their weak knowledge in the areas of analgesics and their usage.

However, more encouraging scores have emerged from a few recent studies conducted in hospitals in the US. In a survey by Moceri and Drevdahl (2012), who used the KASRP tool

on 91 nurses in the emergency departments of five hospitals in north-western America, the knowledge level (average correct score = 76 %) among nurses was comparatively higher than those of other similar studies. They found no correlation of the test score with factors such as age, education level, years of general nursing experience or duration of emergency department experience. Similarly, Al-Shaer, Hill and Anderson (2011) surveyed 129 registered nurses from 10 separate nursing units in a mid-western US metropolitan hospital. Data were collected using the KASRP. In this survey, the nurses scored relatively high (average = 81.2 %, SD = 8.1). With no significant trend in the relationship between knowledge and any of the demographics tested (age, years of experience), the reason for the superiority in performance is unclear. Perhaps it can be attributed to better preparation at the primary nurse training level. Another US study that examined the same parameters in relation to children's pain was reported by Vincent and Denyes (2004). Their survey of 67 nurses yielded an average correct response of 77 %. In relation to attitude, 55 % of nurses believed that 20 % of children generally over-reported their pain. Regarding their ability to detect pain and its severity, 82 % of nurses did well when the child's report was coupled with behavioural manifestations. However, in the absence of behavioural clues, only 49 % detected pain accurately.

Interestingly, a survey of the university curricula in medicine, nursing, dentistry, pharmacy and veterinary medicine found that two-thirds do not include specific teaching on pain in the undergraduate curriculum, with veterinary students being exposed to five times more instructions on pain management topics than medical or nursing students (Watt-Watson et al., 2009). In contrast, according to Brown et al. (1999), KASRP scores based on clinical specialty or practice setting did not correlate significantly with nurses working in various units in a hospital in the US. However, this was based on a poor response rate, and the mean KASRP score was 65 %, limiting the validity of the data. However, the study recognised the

existence of the problem of inadequate knowledge of pain management, although most nurses considered their knowledge of pain to be good (Brown et al., 1999). Problems related to inadequate knowledge and improper attitudes regarding pain management may be why there is a relatively higher prevalence of pain in medical units (Dix, Sandhar, Murdoch & MacIntrye, 2004; Whelan, Jin & Meltzer, 2004; Gregory &Haigh, 2008; Sawyer, Haslam, Daines &Stilos, 2010). Appendix A outlines recent studies that have utilised the KASRP tool to examine nurses' knowledge and attitudes regarding pain.

Students studying for a BSc (Nursing) degree in Iran were found to have limited knowledge of pain management, as revealed by Rahimi-Madizeh, Tavakol and Dennick (2010). Based on responses to a questionnaire of 36 standard items, the overall average correct score was 36.9 % (SD=7.7). However, the authors of the study suggested a possible negative bias in the results due to some issues regarding the translation of the survey tool from English to Persian, and due to the influence of cultural factors.

In a recent study conducted in the US by Duke, Haas, Yarbrough and Northam (2013), the KASRP tool was used to identify the knowledge and attitudes of nursing students and faculty regarding pain. The study surveyed a sample of 162 junior and senior students enrolled in a baccalaureate nursing program and 16 nursing faculty. The results showed that while senior students scored only 68 %, the faculty did not perform much better, with a mean score of only 71 % (Duke et al., 2013), indicating the existence of a serious problem regarding knowledge of pain and its management. A survey involving 313 nursing students in degree programs in the state of Louisiana, US, was carried out to determine their knowledge and attitudes towards pain management (Plaisance & Logan, 2006). The overall correct response rate was 64 %, again showing an unsatisfactory level of knowledge. Their performance was particularly

found wanting on analgesic administration and duration of action, and they seemed to possess an exaggerated fear of addiction to analgesics.

Kaki (2011) assessed knowledge and attitudes regarding cancer pain management among a sample of 325 medical students at the King Abdul Aziz University Hospital in Saudi Arabia using a self-conducted questionnaire. The results revealed that, overall, students' knowledge on the subject was poor. Further, they displayed a poor understanding of cancer-related pain and negative attitudes regarding pain management. This conclusion is based on the findings that 54 % of respondents believed that less than 40 % of cancer patients really had pain. Forty per cent believed that cancer pain could not be relieved with medication, while 68 % were reluctant to use opioids for pain relief because they were afraid of addiction problems.

In many of the above mentioned studies, nurses were found to be particularly deficient in certain knowledge areas, such as pharmacology, addiction, under-estimation of pain, medication withdrawal, substance abuse and cancer-related pain. Chronic conditions are viewed by nurses as conditions with less pain compared to that of acute conditions. This assumption is detrimental to older patients with chronic conditions such as diabetes and renal diseases.

2.4 Part 3: Culture and Pain

Nowadays, the populations of the majority of countries are becoming more diverse as a result of immigration and emigration between countries. Consequently, the multiculturalism extends to the work place as well. This is particularly true for the healthcare sector due to a shortage of locally qualified healthcare staff in many countries, resulting in their dependence on international recruiting and therefore ethnic and cultural diversity in many healthcare systems, including that of Saudi Arabia. In situations of culturally diverse healthcare settings, the cultural differences, language problems and diverse expectations make the pain assessment and management processes even more complex. The experience of pain and its overt expression, as well as the means of communicating pain and pain tolerance, are influenced by peoples' cultural backgrounds. Similarly, from the perspective of professional nurses, cultural factors have a bearing on their ability to assess pain and provide meaningful, appropriate and satisfying healthcare.

2.4.1 Definitions of Culture and Cultural Competence

Culture has been defined as the beliefs, values, behaviour and material objects that define people's way of life (Standage, 2005). Culture also encapsulates how people act and behave when they experience different phenomena in their social lives (Elazia, 2012). In this context, culture defines the social conditions of pain and gives it meaning. It also has a bearing on people's experience of pain and how they will deal with the problem of pain based on their cultural experiences (Low, 1985).

Despite being a small-scale study, Lovering (2006) presented evidence on the influence of culture on the experience of pain based on the knowledge and experiences of a group of culturally diverse nurses working in a Saudi Arabian hospital. Saudi Arabian, Asian, Filipino,

Irish and South African cultures were represented. The methodology of the data collection was based on Herron's (1996) model of cooperative inquiry and comprised a series of meetings with the 10 participants. The findings showed that the cultures shared some beliefs about pain, but were mostly different in their perceptions of the causes and expressions of pain and the necessity for medical intervention.

Low (1985) suggested a three-dimensional view of pain, comprising medical, social and psychological perspectives. The social perspective is rooted in the socio-cultural background of the healthcare giver and the patient or individual suffering the pain. Callister (2003) explained further that the pain experience is complex and based on multiple factors that influence the perception of pain and behaviours within the premises of the socio-cultural context of the individual experiencing it.

Magnusson and Fennell (2011) viewed pain as a multidimensional experience and explained it in terms of sensory, emotional, motivational and social factors. Pain is a universal experience and the experience is felt differently by individuals, both within and between varied cultural groups (Davidhizar & Bartlett, 2000). The interpretation of pain is based on cultural experience and a system of meanings that act as a reference for such interpretation (Elwell, 2000). In this case, social factors determine how pain will affect the individual in line with the meaning they derive from the experience of pain. Davidhizar and Giger (2004) argued that culture is pivotal in shaping the values, beliefs, norms and practices of people as individuals in the way they respond to pain.

The 'culture of pain' refers to the way a society construes the meaning and treatment of pain, while the 'culture in pain' refers to how the perceptions of individuals and their expressions of pain are developed by their cultural orientation. Culture therefore provides the patterns of behaviour to express pain based on the significance attached to it by society. This has a bearing on the individual's perception of the experience of pain (Magnusson & Fennell, 2011). The fact that people are social in nature implies that they are:

Greatly influenced by each of the cultural groups we belong to...Each of the groups influences the way we think and act by instilling in us both general and specific expectations of how the world works and how we should interact with it (Narayan, 2010, p. 38).

Pain will therefore be perceived from a specific cultural orientation to which the individual has been socialised, leading to the suggestion by Fenwick (2006) that pain is 'culturally constructed'. From earlier observations, Fenwick (1998) indicated that Indigenous people in Australia do not attract attention to themselves when in pain; as a result, non-Indigenous nurses considered them as 'unobtrusive' when experiencing pain. Therefore, nurses and other healthcare personnel must develop a sensitivity to the different cultural perceptions of patients relating to pain (Callister, 2003; Blaxter, 2010).

Understanding the cultural orientation of an individual will help a healthcare giver design effective pain management strategies for the patient, where the patient's cultural practices can be incorporated in the pain intervention program (Narayan, 2010; Richardson, 2012). Ignoring the cultural background of the patient may result in the failure of the applied strategies, as the patient may negatively perceive the strategy and refuse it.

Pain supersedes cultural limitations and affects everyone, but how an individual responds to pain is influenced by the individual's previous encounters in his or her experience with life, age, socioeconomic status and gender, among other factors. The expression of pain invariably differs according to the cultural settings and backgrounds of people. In some cultures, people want to know and understand the origins of their own pain, and at the same time be concerned about its implications for healthcare professionals and consequently under-report (Fenwick, 2006). Thus, understanding the phenomenon of pain in an individual is important; however, the health professional must develop a cultural competence, especially when working in communities that are not of his or her cultural orientation (IASP, 2013). Cultural competence has numerous characteristics and involves knowledge and skills, as well as:

Developing an awareness of one's own existence, sensations, thoughts, and environment without letting it have an undue influence on those from other backgrounds; demonstrating knowledge and understanding of the client's culture; accepting and respecting cultural differences; adapting care to be congruent with the client's nature (Purnell, 2005, p. 8).

In this context, it is important for nurses and other healthcare givers to improve their knowledge about the variety of cultures in order to understand and interpret these cultural foundations and how they influence patients. This is relevant to understand the health-seeking behaviours of patients and exploit the motivations that led them to seek healthcare services (Jones, Brownlee & Cantor, 2002; Mazzilli &Davis, 2007). This is a way of making pain intervention strategies effective in the context of the patient's perspective and interpretation of the meaning of his or her pain in his or her milieu (Narayan, 2010).

In building culturally competent care for patients, understanding the cultural orientation of the patient in the context of his or her pain is not enough. Cultural competence enables nurses to distinguish their own cultural backgrounds from those of patients, which have influenced the cultural patterns in which patients' pain is projected (Callister, 2003). Cultural competence enables healthcare givers to satisfy the diverse needs of patients in a multicultural society. Its greatest advantage is that it enables nurses to design healthcare delivery packages that meet

people's needs (Callister, 2003; Flowers, 2004). In this way, the safety of patients is enhanced and the quality of healthcare delivery improves, while medical errors are reduced—especially those that may arise out of misdiagnosis. Generally, for many adult patients with pain, cultural competence eliminates the feeling of a paternalistic relationship with healthcare givers, as they are recruited into designing the pain intervention to which they become the objects (Habiba, 2000; Williams, Haskard & DiMatteo, 2007). Nurses therefore have an ethical duty to afford their medical clients' correct appraisal of their pain and administer suitable pain relief interventions (Fenwick, 2006; Macintyre, 2001).

Cultural competency is a response strategy to the multicultural and multilingual needs of an emerging diverse population globally, and it is becoming an important approach in handling healthcare diversities and disparities (Flowers, 2004). Cultural competency is therefore a key way of improving patient outcomes in the treatment and management of pain, regardless of any cultural differences between healthcare workers and patients(IASP, 2013).

Learning about pain begins in one's childhood. In this learning and socialisation process, an understanding develops regarding the 'normal' and 'right' ways to deal with pain, as well as 'abnormal' or 'wrong 'ways (Davitz & Davitz, 1985). During their training, nurses receive additional knowledge and skills regarding the 'right' way to care for patients in pain. These lead to a strong, albeit unconscious, sense of how competent nurses think and practice (Ludwig-Beymer, 2008). The complete and thorough assessment of pain, which is a prerequisite for its successful management, requires effective nurse–patient communication. Clinicians do not generally use an interpreter service when interviewing patients, regardless of whether there is language incompatibility. Such situations make it almost impossible for caregivers to adequately assess pain and consequently treat patients and provide information on pain management principles (Wilson-Stronks et al., 2008). This problem can be overcome

to an extent by using tools designed to assess pain in children or cognitively impaired patients, but this may also be ineffective and may result in suboptimal pain outcomes. Nurses can use many cues in addition to direct communication, such as facial expression, body posture and activity level, to assess patients' pain (McCaffery, Ferrell & Pasero, 2000). However, it is well known that nonverbal communication patterns also vary across cultures and are therefore also likely to be misinterpreted (Brinkus& Narayan, 2002).

Nowadays, in most healthcare settings, accreditation and regulatory standards require healthcare providers to use competent medical interpreters for effective communication with patients whose language differs from that of the doctor or the nurse (Wilson-Stronks et al., 2008). Such formalisation is a timely need because of existing cultural diversity among healthcare providers, and it is no longer appropriate to depend on family members or other informal interpreters that may compromise patients' ability to understand and be understood (Divi, Koss, Schmaltz & Loeb, 2007). Communication lapses will invariably result in inadequate pain assessment and management.

2.4.2. Conceptual Framework of Leninger's Cultural Care Diversity and Universality Theory (CCDU)

Culture has been explored by a number of theorists, such as Leininger, Purnell and others. Leininger's cultural care diversity and universality theory (CCDU) provides culturally congruent care to individuals, families, groups, communities and institutions. The CCDU theory has been widely used as a theoretical framework in culture-related subjects. A theoretical framework can be used to describe relations between different variables, and it can be considered the map that guides researchers through the process of their research (Sinclair, 2007). Different studies on particular subjects use different theoretical frameworks; there is
no right or wrong theoretical framework to use when examining a topic, as every topic can be examined from different perspectives. As the major component of this study is the qualitative part, which mainly focuses on barriers and cultural factors that affect pain management, Leininger's CCDU theory is useful because it describes the relations between the different variables of cultural aspects. In addition, the CCDU theory highlights the factors that influence care—especially in multicultural settings such as Saudi Arabia. Although Leininger's CCDU theory is specific for nursing care and research, it is broad enough to identify the cultural factors that influence pain. Thus, Leininger (1996) advocated that professional nurses should develop an appreciation of the cultural conceptualisation of pain in all of its different aspects, such as personal experiences, responses to it, communication about it and pain management.

Leininger's CCDU theory has its background in the concept of care as a central component of nursing (Erkes, Parker, Carr & Mayo, 2001). During the development of this theory, Leininger identified care and cultural knowledge as factors that determine nurses' understanding of the various forms of compliance, healing and wellness. This brought about Leininger's CCDU theory, which is the only theory of nursing that outlines a relationship between cultures and nursing (Al-Aameri, 2000).

Leininger later developed the Sunrise Model (1991) to serve as a cognitive map to support and guide nursing practice. The Sunrise Model (see Figure 2.1) demonstrates the interrelationships of the concepts of the CCDU theory and highlights the factors that influence care, such as religion, politics, economics, worldview, environment, cultural values, history, language and gender.



Figure 2.1: Sunrise Model

The CCDU theory assists nurses to learn about the worldview of a group or individual. A cultural group derives its cultural and social structure dimensions from its members' individual worldviews, but it is also shaped by the environment and language contexts in which it exists. There is variation among different cultural groups in relation to the manner in which each cultural and social structure dimension is lived and experienced. The CCDU lists the following seven cultural and social structure dimensions:

- technological factors
- religious and philosophical factors

- kinship and social factors
- cultural values and lifeways
- political and legal factors
- economic factors
- educational factors.

In their effort to provide culturally compatible care, nurses use a combination of aspects from generic (traditional) and professional healthcare systems. This approach ensures unique, custom-made care for each individual or group. In this regard, the following three modalities are the considerations that guide nurses' judgement, decision-making and actions:

- cultural care preservation/maintenance
- cultural care accommodation/negotiation
- cultural care re-patterning/restructuring.

Culture care preservation and maintenance implies the need to conserve existing behaviours and lifestyles that are good for health. Culture care accommodation 'refers to those assistive, supporting, facilitative, or enabling professional actions and decisions that help people of a designated culture to adapt to, or to negotiate with, others for a beneficial or satisfying health outcome with professional care providers' (Leininger, 1991, p. 48). Culture care repatterning and restructuring:

refers to those assistive, supporting, facilitative, or enabling professional actions and decisions that help a client(s) reorder, change, or greatly modify their lifeways for new, different, and beneficial healthcare patterns while respecting the client(s) cultural values and beliefs and still providing a beneficial or healthier lifeway than before the changes were co-established with the client(s) (Leininger, 1991, p. 49).

Leininger emphasises the meaning and significance of one's culture in the process of generating health and caring behaviour for an individual (Lasch, 2000). This followed the realisation that patients from different backgrounds place greater importance on the care they receive than the nurses attending to them. Thus, there was a need to establish a theoretical framework in order to discover, explain and predict aspects of care and develop the CCDU in response to the various studies on culture care (Cairns, Thompson & Wainwright, 2003).

According to Leininger (1995), the four nursing paradigms of person, environment, health and nursing are not enough. For Leininger, nursing is both a discipline and a profession; therefore, the word 'nursing' cannot explain the phenomenon of nursing. Instead, 'care' is the word that should be used to explain the practice of nursing. This is markedly different from that of other scholars and authors. In addition, Leininger finds the term 'person' too limiting and culture-bound to explain nursing, as the concept of 'person' as an individual entity does not exist in all cultures. The word 'health' also tends to mean different things to different people, depending on the discipline through which they first knew the word (Al-Aameri, 2000). As a substitute for 'environment', Leininger prefers to use the term 'environmental context', which is more encompassing of physical, ecological, socio-political and even cultural settings.

In this case, care is seen to arise from the need to solve human problems; thus, it occurs in a cultural context (Watt-Watson et al., 2001). Culture is therefore defined as a set of behaviours that set one community apart from any other that exists on earth. This definition presents care as a universal and diverse concept. Culture is composed of emic (insider) and etic (outsider) points of view. When studying the relationship between care and culture, ethno-history is another factor that influences this relationship profoundly.

2.5 Chapter Summary

Pain is a common human experience and the most prevalent symptom associated with sickness and injury. Pain essentially arises from a signal sent from the peripheral nerves to the brain. It is part of the survival mechanism in all species of animals to advise the organism that something is wrong. However, pain is not exclusively physiological; it includes spiritual, emotional and psychosocial dimensions. Pain experiences may be acute or chronic, with chronic pain deriving from a chronic, deteriorating condition, and acute pain as one of many symptoms of a patient in palliative care. Whether acute or chronic, pain can change one's life. Regardless of the intensity of the pain, there are many treatment options available due to the vast advances made in the past few decades in understanding pain and treatment methods However, a review of the scientific literature and medical reports reveals that the majority of patients do not receive adequate pain management. A variety of factors, including inaccurate information, nurses' lack of knowledge and skills, myths, rumours, fear and cultural issues contribute to inadequate pain management.

The goal of pain management is to address all aspects of pain and to provide pain relief with minimal side effects. As frontline healthcarers, nurses have a responsibility to present and implement treatment options to patients. They also have a responsibility to ensure that patients are provided with pain management education. Topics that are relevant to nurses' roles in pain management, including assessment, treatment and barriers to effective pain management, have been discussed in this chapter.

Chapter 3: Research Methodology

3.1 Introduction

A mixed-methods research design was used to determine nurses' knowledge and attitudes towards pain management and to identify barriers to achieving optimal pain management in Hail region hospitals in Saudi Arabia.

The purpose of this chapter is to present and describe the methods and procedures that were employed in the process of data collection, including the research questions, design, sample, setting, instruments, procedure, data collection, data analysis and ethical issues.

3.2 Mixed Methods Design

Creswell and Plano Clark (2007) presented a complex and inclusive definition of mixed methods research:

Mixed methods research is a research design with philosophical assumptions as well as methods of inquiry. As a methodology, it involves philosophical assumptions that guide the direction of the collection and analysis of data and the mixture of qualitative and quantitative approaches in many phases in the research process. As a method, it focuses on collecting, analysing, and mixing both quantitative and qualitative data in a single study or series of studies. Its central premise is that the use of quantitative and qualitative approaches in combination provides a better understanding of research problems than either approach alone (p. 5).

As Creswell and Plano Clark (2011) illustrated, the application of mixed methods as a research design is beneficial in promoting the validity and reliability of the outcomes being

sought, as the magnitude or effect of the outcome is usually stronger when results from different methods are brought together. At times, errors may occur in the analysis process, which may lead to faulty findings. In such a scenario, using a mixed method approach would enable the researcher to identify the existing discrepancies between the results from the two methods, and thus identify and correct any errors before presenting or publishing the information.

When dealing with complex research outcomes, Creswell (2009) asserted that the mixedmethods approach, as opposed to the single-method approach, can be highly efficient in enabling the researcher to understand the different facets of complex research outcomes. This usually promotes a better understanding of the research outcomes as opposed to when the researcher utilises a single method in the research design. The quantitative analysis method is usually efficient when dealing with pre-established research variables; as such, combining quantitative and qualitative analysis enables the researcher to capture other unanticipated information revolving around the topic of study, thereby providing the researcher with a better understanding of the research topic.

As Johnson and Onwuegbuzie (2004) suggested, the application of mixed methods in the research design is usually beneficial for the researcher to ensure that the sampling frame and sampling technique is effective in terms of being representative of the entire population. Creswell (2009) argued that when conducting a research study, no single analysis method could be more appropriate and effective than any other in terms of providing more valid and reliable information. Therefore, the use of mixed methods in research analysis provides an avenue for using both quantitative and qualitative analysis methods to promote increased validity and reliability of the research outcome.

3.2.1 Philosophical Foundation for Mixed-methods Research

As noted by Migiro and Magangi (2011), there has been continuing debate over the philosophical basis of mixed-methods research. This debate has been revolving around the question, 'Do philosophical paradigms (e.g. post-positivism and constructivism) and research methods have to fit together?' (Migiro & Magangi, 2011, p. 3758). This debate emerged in the 1960s and 1970s due to the increased popularity of qualitative research methods, as well as the identification of philosophical destinations between the traditional post-positivist and naturalistic methods.

Creswell (2003) observed that a point was reached where mixed-methods research was regarded as unachievable because the methods were not compatible. Nonetheless, Migiro and Magangi (2011) noted that some researchers refuted and countered this reasoning, arguing that different research methods and philosophical paradigms were compatible and could be combined efficiently in mixed-methods research. This perspective was supported by Greene and Caracelli (2003), who claimed that different methods could be combined and used together in a single research study as a means of capitalising on the strength of one method to reduce the weaknesses of the other.

As noted by Migiro and Magangi (2011), a significant issue in conducting mixed-methods studies lies in the question, 'What philosophical paradigm issue is the best foundation for mixed methods research?'(Migiro & Magangi, 2011, p. 3758). In this regard, Teddlie and Tashakkori (2009) argued that the issue could be addressed from several perspectives. The first perspective suggests that in mixed-methods research, the paradigms usually compete, thus giving each paradigm equal chance and merit. The competing nature of the paradigms gives mixed-methods research contradictory ideas and contentious issues that cannot be

reconciled. In this perspective, mixed-methods research is just a method that provides researchers with philosophical foundations and justifications for using this method.

According to Greene and Caracelli (2003), another perspective proposes that practicality is the best paradigm for mixed-methods research. This perspective presents for consideration the value of both objective and subjective knowledge in mixed-methods research. Migiro and Magangi (2011) stated that Rosseman and Wilson pioneered the incorporation of pragmatism into mixed-methods research. They explained the existing difference among methodological purists, pragmatists and situationalists. Purists believe that quantitative and qualitative research methods are derived from completely different research assumptions. Situationalists feel that both quantitative and qualitative methods are valuable, but the appropriateness of each depends on the prevailing circumstances. Pragmatists believe that both methods are valuable and can be used in a single research study, regardless of the prevailing circumstances.

Recently, there have been attempts to connect pragmatism and mixed-methods research. One such attempt is the work of Teddlie and Tashakkori (2009), who attempted to link pragmatism and mixed-methods research and proposed that the research question should be more important than all other research elements. These sentiments are shared by other scholars, who observe that pragmatism provides the most significant philosophical basis of mixed-methods research. In this vein, one cannot rule out the contributions of both the quantitative and qualitative methods in mixed-methods research. Nonetheless, pragmatism emerges as the best philosophical foundation that can be used in the process of justifying the use of different methods within the boundaries of a single research study.

3.3 Study Design

After reviewing the literature on mixed-methods approaches, the researcher decided that a mixed-methods approach was the most appropriate way of gaining an in-depth understanding of the current situation regarding nurses' knowledge and barriers regarding pain management. Bryman (2001) described the research design as a guide to the procedures to be used in the process of conducting a research study. In the present study, a mixed-methods research design was deemed an appropriate approach to the topic.

This explorative and descriptive mixed-methods study utilised two phases. The first phase involved the administration of a questionnaire, which aimed to determine Hail region nurses' knowledge and attitudes regarding pain. The second phase used a qualitative methodology (semi-structured interviews) to explore the barriers perceived by nurses to achieving optimal pain management. The interviews in this phase were audio-recorded. The number of participants in the first phase comprised 303 nurses working in Hail region hospitals, and the second phase included 28 participants. The inclusion criteria for the selection of participants were nurses aged 21–65 with more than six months of nursing experience in Saudi Arabia and willing to share their experience.

3.3.1 Sequential Explanatory Mixed-methods Design

This research used a sequential explanatory design. Creswell (2008) and Creswell and Clark (2007) argued that this design is one of the most popular forms of mixed-methods designs. According to Creswell (2003), it is the most straightforward approach compared to other mixed-methods designs. Creswell (2003) pointed out that the sequential explanatory mixed-methods design is a research methodology that is applied in situations where the data are collected in two different phases. In utilising this design, the researcher began by gathering the quantitative data and conducting statistical analysis to obtain the numeric information. This was followed by gathering and analysing the qualitative data to provide explanations for the numerical results from the analysis of the quantitative data gathered during the first phase. The qualitative data provided explanations for the quantitative data, by providing the connection between the two types of data. The rationale behind using the sequential explanatory mixed-methods design was that the quantitative data and analysis process provided a general understanding of the research topic. Nonetheless, the qualitative information gave a detailed explanation of the quantitative results to provide further understanding of the research topic by obtaining a deeper review of the views of the research participants.

3.4 Study Setting

This study was conducted in five hospitals in the Hail region, which is located in the northern region of Saudi Arabia. There are 11 hospitals in the region—including four central and seven peripheral hospitals—including general, maternity and psychiatric hospitals. The survey was distributed to five general hospitals—two of which are central and three are peripheral. Hospital A (King Khalid Hospital) is the largest hospital in the region, with a capacity of 210 beds. Hospital B (Hail General Hospital) is the second largest and has 200 beds. Hospital C (Bagaa General Hospital) is a peripheral hospital with 30 beds. Hospital D (Al Ghazala Hospital) is a peripheral hospital with 50 beds. Hospital E (Alshamli Hospital) is a peripheral hospital with 30 beds. Hospital by the MOH.

3.5 Sampling

As observed by Denzin and Lincoln (2011), sampling is the process through which research participants are selected in a manner that ensures that they effectively represent the population from which they are selected. This representative group of research participants is known as a research sample. Similarly, Castillo (2009) described sampling as the process of selecting a representation of the total research population. Welman, Kruger and Mitchell (2005) pointed out that some researchers are usually biased in the process of selecting a research sample and therefore collect data that do not represent the views of the entire intended research population. As illustrated by Teddlie and Tashakkori (2009), different sampling techniques can be applied in the process of selecting research samples for mixed-methods research. This explorative, descriptive mixed-methods study sampled the local and expatriate nurses who were working in Hail regional hospitals, and the procedure included two phases.

3.5.1 Sample in Phase One (Quantitative)

This phase involved the distribution of 500 KASRP surveys (NKASRP) to the total estimated population of nurses in the above five hospitals. A survey return rate of 60 % (N = 303) was achieved. In this quantitative phase, a convenience sampling method was employed to sample local and expatriate nurses working in Hail region hospitals. Convenience sampling is probably the most commonly used of all sampling techniques. As defined by Creswell (2003), convenience sampling is a form of non-probability sampling where research participants are selected based on convenience in accessibility and proximity to the researcher. As such, this method promotes the selection of the most conveniently available individuals as study participants.

3.6 Research Instruments

3.6.1 Demographic Data

The first section of the research instrument sought demographic data from the participants, including: gender, age, nationality, educational achievements, religious affiliation, years of work experience, occupation/position, area of assignment/department, hospital where the nurse is working and the number of pain management courses undertaken by the nurse (Appendix B). The demographic data were obtained from the questionnaire that was distributed along with the Consent Form (Appendix C), Plain Language Statement (Appendix D) and Letter of Invitation to the Interview (Appendix E).

3.6.2 Knowledge and Attitudes Survey Regarding Pain (KASRP)

The KASRP questionnaire was developed by Betty Ferrell and Margo McCaffery in 1987. It consists of 22 true or false items, 13 multiple choice items and two case studies with two multiple choice items each. It has been widely used and designed to measure the knowledge and attitudes of healthcare professionals regarding pain. The content of the tool was extracted from the current standards of pain management formulated by the World Health Organization, the American Pain Society and the Agency for Health Care Policy and Research. A copy of the questionnaire, along with permission to use it, was obtained from the City of Hope Pain & Palliative Care Resource Center (Appendix F).

3.6.3 Tests of Validity and Reliability

Henson (2001) described reliability as the consistency of a measurement process or a measuring instrument. A reliable instrument should assign similar scores to objects or

elements with equal values. A reliable measuring instrument is known to maximise the true score component while simultaneously reducing the error component of the outcome or the results. In research procedures, obtaining reliable data means that if similar research instruments are applied to a different research sample from the same research population, the results obtained should be in close agreement with those obtained in the earlier measure. Content validity has been previously established by a review of pain experts (Ferrell & McCaffery, 2008). Zikmund (2003) considered validity as the extent to which a measuring instrument efficiently measures the variables that it is meant to measure. In measuring validity, the focus shifts from the scores or results obtained using the measuring instrument to the conclusions or inferences that are made from these scores or results. As such, a valid instrument leads a researcher to make the most appropriate or efficient inferences. Validity in this research was promoted by using the mixed-methods approach, which ensured that the obtained results were counterchecked before making any conclusions. The KASRP construct validity was established by the contrasted-groups method of comparing the scores of nurses of varying levels of expertise, such as students, new graduates, oncology nurses, graduate students and senior pain experts. The tool was identified as discriminating between levels of expertise. Test-retest reliability was established by repeat testing in a continuing education class of staff nurses (r > 0.80). Internal consistency reliability was established (alpha r > 0.70) with items reflecting both knowledge and attitude domains (Ferrell & McCaffery, 2008).

Although the instrument had already been validated by the original author and other pain experts, the researcher also subjected the instrument to certain validity and reliability testing. The Kuder–Richardson Formula was used to measure the internal consistency coefficient and aided by other measures of central tendency and variability based on the actual scores of the nurses in the pain management questionnaire.

As shown in Table 3.1, the reliability estimate of the questionnaire was not that far from the reliability estimate conducted by the previous researcher. The computed Kuder–Richardson internal consistency coefficient of 0.79 fell within the rule-of-thumb range of reliability estimates of 0.60–0.90. This meant that the instrument was reliable enough to measure the knowledge and attitudes regarding pain management among nurses at hospitals in the Hail region. Additionally, the mean score of the responses to the questionnaire was 16.70, with a standard deviation of 3.93 and a computed standard error of mean of 1.80. The mean success rate +/- standard error of 16.7 +/- 1.8 on a 40-item questionnaire suggests that the knowledge of participating nurses regarding pain management is low.

Tab	le	3.	1:	Rel	liab	ility	Testi	ng	of	the	KA	SRP
-----	----	----	----	-----	------	-------	-------	----	----	-----	----	-----

Statistics	Values
Mean Score	16.70
Standard Deviations of Score	3.93
Standard Error of Mean	1.80
Reliability	0.79

3.7 Procedure of Survey Distribution

After ethics approval was obtained from the university (Appendix G) and permission was received from the general directors of the five hospitals (Appendix H), an invitation to participate in the study (Plain Language Statement) and the questionnaire were distributed to the staff via the Nursing Education Department. The researcher was available for contact in Saudi Arabia during the data collection phase to answer any questions from the participants concerning the study. The questionnaire was voluntarily completed by participants in their own time, and it took approximately 20 minutes to complete. An invitation letter accompanied the questionnaire to invite nurses to participate in the second phase of this study (semi-structured interviews). To make it convenient and easy for all respondents, and to increase the response rate, participants were asked to drop the questionnaire, after completion,

into a sealed box labelled 'Completed Questionnaire', that was placed in each ward. These boxes were collected after three weeks. The data collection period covered one month, from July 2011 to August 2011.

3.8 Phase One Data Analysis (Quantitative)

The quantitative data from the questionnaire were analysed using the Statistical Package for the Social Sciences (SPSS) Version 18 following the methods described by Field (2009). The dependent variable for the descriptive and statistical analyses consisted of the proportion (per cent) of correct answers.

To address the first research question, the proportion (per cent) of N = 303 participants who responded with correct answers (per cent) for each of the 40 questions was computed and ranked in order of magnitude from the highest to the lowest. The top 10 questions, which received the highest percentage of correct answers, and the bottom ten questions, which received the lowest percentage of correct answers, were identified and compared. Descriptive statistics (means and standard deviations) and inferential statistics (univariate analysis of variance (ANOVA) were used to address research question number three and determine whether the mean percentage of correct answers varied significantly with respect to the 10 independent variables. Each independent variable represented mutually exclusive groups of participants who were categorised according to their demographic/cultural characteristics—specifically, gender, age, nationality, qualifications, religion, experience, position, department, hospital and courses.

The following null hypotheses (H₀) were tested. Each null hypothesis proposed that there were no significant differences between the mean values of the dependent variable with respect to each of the independent variables:

- H₀1: The mean correct answers (per cent) did not differ significantly with respect to gender.
- H₀2: The mean correct answers (per cent) did not differ significantly with respect to age.
- H₀3: The mean correct answers (per cent) did not differ significantly with respect to nationality.
- H₀4: The mean correct answers (per cent) did not differ significantly with respect to qualifications.
- H₀5: The mean correct answers (per cent) did not differ significantly with respect to religion.
- H₀6: The mean correct answers (per cent) did not differ significantly with respect to experience.
- H₀7: The mean correct answers (per cent) did not differ significantly with respect to position.
- H₀8: The mean correct answers (per cent) did not differ significantly with respect to department.
- H₀9: The mean correct answers (per cent) did not differ significantly with respect to hospitals.
- H₀10: The mean correct answers (per cent) did not differ significantly with respect to pain courses.

3.8.1 PostHoc Multiple Comparison Tests

ANOVA was used to test H0—that there are no overall significant differences between two or more mean values across specified groups of participants. The alternative hypothesis (HA) is that at least one of the mean values is different to the others. SPSS computes the F statistic, which is the ratio of the variance for the between-subjects effects to that of the unexplained effects (error variance); however, the F statistic does not test the significance of difference between a given pair of means. SPSS carries out an array of 18 multiple comparison tests after ANOVA has been performed to compare specific pairs of mean values and to determine which mean values are significantly different from each other (Field, 2009). The Scheffé test, which is based on homogeneous subsets, was chosen for the purpose of this study because: (a) it does not, like other multiple comparison tests (e.g. Tukey's HSD), assume that the sample size is the same in each group of participants; and (b) Sheffé's test permits *posthoc* comparisons between all pairs of mean values, whereas other *posthoc* tests permit only a limited number of *posthoc* comparisons (Toothaker, 1991).

3.8.2 Homogeneity of Variance

ANOVA and the Scheffé test assume homogeneity of variance, meaning that the variance (the square of the standard deviation) of the dependent variable must be equal across each level of the independent variables (Rutherford, 2001). If the variances of the correct answers (per cent) were not equal across each group of participants, then the statistical inferences of ANOVA would be compromised and it would be difficult to make definitive conclusions. Therefore, homogeneity of variance was checked using Levene's F statistic, which is supported by SPSS as part of the ANOVA (General Linear Model) procedure (Field, 2009).

The null hypothesis was that the variances were equal. Rejection of H0 provided evidence to conclude that the variances were not homogeneous.

3.8.3 Statistical Significance

Statistical significance was evaluated by comparing the p-values of the F statistics and the Scheffé test statistics against a prescribed significance level ($\alpha = .05$). If the p-value of the F statistic was $\leq .05$, then the H0 was rejected (assumed to be false). Rejection of the null hypothesis at $\alpha = .05$ provided evidence to accept the HA and conclude that the mean value of the dependent variable differed significantly with respect to the independent variables. If the p-value of the F statistic was > .05, then the null hypothesis was not rejected (assumed not to be false) and it was concluded that the mean scores did not differ significantly across the groups of participants.

The value of α reflected the extent to which the inferential test may produce a Type I error (i.e. the false rejection of the null hypothesis by random chance, when in fact it should not be rejected). This limit is conventionally set to a small value—typically $\alpha = .05$ —so that the probability of a Type I error is reduced. The prescription of $\alpha = .05$ implied a one-in-20 chance of making a Type I error, which is conventionally agreed to be an acceptable level; however, $\alpha = .05$ is not a gold standard. It is only one of many rules of thumb used in practice to interpret inferential test statistics (Baguley, 2004). One of the limitations of using $\alpha = .05$ as the level of significance is that when multiple null hypotheses are tested consecutively, the chance of making Type I errors is elevated. The probability of making a Type I error when testing 10 consecutive null hypotheses, as in this study, is $1 - (1-.05) \ 10 = .40$ (Hair et al., 2010). This implies that four out of the 10 null hypotheses could be rejected by random

chance and not because there were any important or meaningful differences between the mean values.

The p-values for ANOVA statistics vary with respect to the sample size. When the sample size increases, the p-value decreases. When the sample size is small, the p-value is high, and there may be insufficient power to reject the H0 at $\alpha = .05$. When the sample size is too small, a Type II error could occur (i.e. the null hypothesis is falsely not rejected because the p-value is < .05 when in fact it should be rejected) (Zodpey, 2004). For this reason, there is little point in conducting ANOVA if the sample size is too small (Rutherford, 2001).

Attempts were made to ensure that the sample size in each group of participants in this study was above the minimum necessary to avoid Type II errors. Ideally, the group sizes should be equal to provide the highest power for ANOVA, but this was not possible due to a variety of reasons. The minimum sample size in each group recommended by Cohen (1992, p. 158, Table 2) for conducting ANOVA was applied as a guideline, assuming $\alpha = .05$, a power of .8 (i.e. 20 % chance of making a Type II error) and a medium effect size. Cohen recommended a minimum of n = 32 participants in each group if g = 2 groups, n = 17 if g = 3, n = 11 if g = 4and n = 8 if g = 5. To test for the effects of age, nationality, qualification, experience, position and department, two or more similar categories had to be collapsed (i.e. combined) to ensure that there were enough participants in each group. Gender (male and female) and religion (Muslim, Christian or Hindu) could not be logically collapsed; therefore, the number of males (n = 21) and the number of Hindus (n = 11) were less than the minimum required to conduct ANOVA with two or three groups respectively, potentially limiting the validity of the statistical inferences. The results were also influenced by missing values. The sample sizes in all of the groups were reduced because some participants did not report their demographic/cultural characteristics.

3.8.4 Normality

ANOVA is a parametric inferential test, meaning that the dependent variable should theoretically be normally distributed (i.e. its frequency distribution should approximate a symmetrical bell-shaped curve). However, ANOVA is robust, meaning that years of practical use have revealed that Fstatistics are not necessarily biased by deviations of the data from normality. As long as the frequency distribution is approximately mound-shaped, with the mode (highest frequency) close to the centre, the distribution is not heavily skewed (with the mode at the extreme left- or right-hand side) and there are few outliers (i.e. extremely large or small values that are not contiguous with the frequency distribution), then the statistical inferences obtained using ANOVA are not severely compromised (Rutherford, 2001). Consequently, the shapes of the frequency distributions for the correct answers (per cent) of each participant were visually checked in this study using histograms.

3.8.5 Clinical Significance

The *p*-values computed using ANOVA only reflected statistical significance (i.e. whether or not the differences between the mean values were real and not caused by random chance), but they did not imply that the results were important or had any meaningful implications in reality. Statistical significance is not equivalent to clinical significance, referring to the magnitudes of the differences between mean values and the scientific importance of the results, including their meaningful implications, with respect to the context of the research. For this reason, many researchers favour the interpretation of effect sizes instead of *p*-values (Brown, 2008; Ferguson, 2009; Kotrlik & Williams, 2003; Hill & Thompson, 2004). The reporting of effect sizes in addition to *p*-values was recommended by the APA task force on statistical inference (Wilkinson, 1999). Effect sizes were therefore computed and interpreted

in this study because they reflected the clinical significance of the results, providing more useful information about the effects of demographic/cultural characteristics on the levels of knowledge of the participants than could be inferred from statistical significance alone. Another advantage of effect sizes is that, unlike *p*-values, they are not a function of the sample size and are not influenced by Type I or Type II errors. The effect size computed using the ANOVA (General Linear Model) procedure in SPSS was eta squared (η^2), representing the proportion of the variance in the dependent variable explained by the independent variables (Brown, 2008). Applying Ferguson's (2009) subjective criteria for the guidance of clinicians and researchers, $\eta^2 \leq .04$ indicated a minimal or negligible effect; $\eta^2 =$.05 to .25 indicated a relatively small effect; $\eta^2 = .25$ to .64 indicated a relatively moderate effect and $\eta^2 \geq .65$ indicated a relatively strong effect.

3.9 Phase Two (Qualitative)

3.9.1 Interviews

In the second phase of this research, the researcher utilised a purposive sampling approach to interview participants who had participated in the first phase. Twenty-eight nurses who had participated in Phase I have signed the consent form and contacted the researcher to be interviewed. They were interviewed individually in a meeting room in the hospital during their break and asked open-ended questions (Appendix I) regarding their experience of pain and pain management. The interview questions were in English and developed by the researcher to explore the barriers that face nurses during pain assessment and management, and to answer research question number two. The duration of each interview was about 30 minutes and no participants had any difficulties answering the questions. Interviews were audio-recorded in order to be transcribed for further analysis. The researcher did the transcription and each participant was allocated an alphabetical letter (A, B, C etc.) for

identification. As noted by Cohen, Manion and Morrison (2013), interviews are a way of collecting research data by talking and listening to the research participants as they air their experiences on the research topic. The interviewees usually provide the primary data on the subject being investigated. In this research, the use of interviews was relevant because it was a means of collecting data and gaining knowledge on the research topic from the participants. Kvale (1996) stated that interviews are 'an interchange of views between two or more people on a topic of mutual interest, sees the centrality of human interaction for knowledge production, and emphasizes the social situatedness of research data' (p. 14).

Equally, the use of interviews promotes the collection of research information from participants within their local settings and ensures the accuracy of the data collected. While conducting interviews, the researcher should always ask questions in a way that elicits valid responses from the interviewees. In this vein, Hoyle, Harris and Judd (2002) pointed out that the researcher has the 'dual goals of motivating the respondent to give full and precise replies while avoiding biases stemming from social desirability, conformity, or other constructs of disinterest' (p. 144).

By using interviews in this research, highly personalised data were gained from the perspectives of the interviewees. This gave the interviewer the opportunity to probe for further views from the interviewees and enabled the interviewees to further explain themselves. David and Sutton (2004) observed that semi-structured interviews are usually non-standardised and commonly used in qualitative analysis.

In semi-structured interviews, Gray (2004) affirmed that note-taking is essential. Moreover, this method of data collection gives the researcher the chance to probe for further views of the interviewee if he or she feels that the outcome is not satisfactory. David and Sutton (2004)

suggested that having 'key themes and sub-questions prepared in advance helps in giving the researcher a sense of order from which to draw questions from unplanned encounters' (p. 87). Thus, planning is essential before conducting semi-structured interviews. Nonetheless, probing enables a researcher to explore new research dimensions that had not been taken into consideration while planning the research study.

3.9.2 Transcribing Qualitative Data

Malterud (2001) commented that the increased use of qualitative research calls for efficient data collection techniques and documentation procedures. Similarly, the increased use of computer applications in qualitative data analysis requires more efficiency—especially in the process of transcribing the research data. Although computerised qualitative data analysis promotes efficiency in the management and processing of research data, researchers have continued to play a crucial role in the transcription, processing and management of qualitative research data. Qualitative data should be transcribed word by word because such transcription promotes the reflection of the respondent's knowledge in the transcribed data.

Long and Johnson (2000) observed that transcription guidelines are meant to guide researchers in the process of organising and analysing the data despite the analytical tools and techniques applied. Moreover, the guidelines ensure that the data transcripts are made efficiently, systematically and consistently. In the present research, the data were transcribed word by word to ensure that the respondents' views and meanings were not altered in the transcription process. Further, a categorisation scheme was used and guided in the process of organising, classifying and indexing the research data.

3.9.3 Phase Two Data Analysis (Qualitative)

Thematic analysis was used to analyse the qualitative data. The qualitative data were analysed by the researcher and checked by the supervisors. Tobin and Begley (2004) asserted that thematic analysis involves seeking important themes in the process of describing or explaining the research phenomenon. The important themes are identified by reading through the data carefully and then re-reading the data to confirm any omitted themes. These important themes usually form the basis for research analysis. In this research, thematic analysis was used to support the quantitative research results. The important themes were identified in each transcript. Qualitative data were assigned numerical codes to promote the classification of research themes based on the frequency of theme identification by the respondents in each research question. While conducting a thematic analysis of open-ended research questions, the themes in each question were identified and counted.

3.10 Ethical Issues (Confidentiality, Anonymity and Protection of Human Rights)

In this study, several methods were employed to ensure the protection of the participants' rights, confidentiality, anonymity, privacy and protection from discomfort. Participants were fully informed about the nature of the study before obtaining their written consent and involving them in the research procedures. Additionally, the aims of the research were explained to them to ensure that they understood what was required of them. Their consent was sought in addition to being assured that any information they provided would be used confidentially for research purposes only, without being divulged to any third parties. The respondents were also informed that their participation in the research process was voluntary and that they had the right to withdraw their participation if they felt that their confidentiality/anonymity or basic rights were being violated at any point.

The survey explored nurses' knowledge and attitudes regarding pain management. It was anticipated that the survey would not judge, offend or affect the cultural beliefs of the participants; rather, it sought to clarify these cultural beliefs in relation to pain management. Both the transcribed data and demographic questionnaires were coded, and the participants' names were removed. The Ethics Committee at the Royal Melbourne Institute of Technology (RMIT) approved the research proposal, and prior approval to perform the data collection was obtained from the MOH in Saudi Arabia.

As pointed out by Haverkamp (2005), ethics must also consider the effects of the research on the participants. Extreme precaution is required, particularly when the research involves direct human contact. On this note, the human rights of the participants were protected. Equally, their personal details, such as names and other information, were only used for research purposes and were highly safeguarded to ensure that their details would not end up in the wrong/unintended hands. In this study, data were stored on a computer, which was password-protected. Hard copies of the data were stored in a locked filing cabinet. The researchers are the only people who have access to the data. Signed consent forms and questionnaires were obtained in hard copy and sent back to Australia by registered post. The interviews were scanned and backed up on the external mini hard drive. This hard drive was transported as cabin baggage on return to Australia. This process enabled the researcher to transfer the data back to Australia while making it difficult for anyone else to access or use the information. Regarding the storage of data, the researcher will keep the data in secure storage for five yeas.

3.11 Chapter Summary

This chapter described the methodology used to conduct the present research. A mixedmethods approach was used to obtain answers to the research questions. In the process, both qualitative and quantitative analysis methods proved to be highly efficient in analysing the data to provide the required information. Equally, the application of mixed methods as a research design was appropriate for the validity and reliability of the research findings. Utilising a sequential explanatory mixed-methods design, the study began by gathering quantitative data and analysing them to provide numeric information. This was followed by gathering and analysing qualitative data to provide explanations for the numerical results that were obtained during the first phase of analysing the quantitative data. Both the quantitative and qualitative results were reviewed, and the university statistician guided the statistical analysis.

Confidentiality, anonymity and protection of human rights were considered during the research process, as recommended in the human research ethics guidelines. Convenient sampling formed the basis for the quantitative research samples, while purposive sampling proved efficient in selecting the qualitative research samples. Data collection was through semi-structured interviews, which gave the interviewer the chance to probe and explore the participants' stories. Thematic analysis generated the research themes that were central to describing the research phenomenon. The following chapter presents the results of the statistical analysis.

Chapter 4: Quantitative Results

4.1 Introduction

This chapter presents a descriptive analysis of the correct responses obtained for each question by all participants, followed by the results of the tests on the 10 null hypotheses concerning mutually exclusive groups of participants, using ANOVA. These 10 null hypotheses were developed from the aims of the research, which are to investigate the demographic and cultural factors that affect the delivery of effective pain management.

The KASRP items can be used to assess nurses' knowledge and attitudes regarding pain management. Therefore, the authors of the KASRP tool suggested that it is better to analyse the data as complete scores and also analyse each item individually rather than distinguishing between items that measure knowledge or attitudes. The data were analysed and evaluated as overall percentage scores obtained, as recommended by the authors of the tool (Ferrell & McCaffery, 2008).

4.2 Demographics of Participants

Demographic data collected in this study consisted of gender, age, nationality, education level, religion, work experience, position, department, hospital and whether participants had followed any pain management courses. As shown in Table 4.1, the majority of participants were female (n=281, 93 %). More than half of the participants (n=177, 59.8 %) were aged 21–30. Participants were from various countries, with those originating from India (n = 124, 41.2 %), the Philippines (n = 112, 37.2 %) and Saudi Arabia (n = 57, 18.9 %) constituting the majority. Over half of the participants held Bachelor's degrees (n = 159, 52.6 %), nearly half had Diploma-level qualifications (n = 138, 45.7 %) and only five participants had a Master's degree. On the basis of religion, the majority of participants were Christian (n = 193, 63.9 %), followed by Muslim (n = 97, 32.1 %). Regarding the years of experience, the majority of participants had worked for 1–5 years (n = 117, 38.6 %). Almost 90 % of participants were staff nurses (n = 268), and the rest were head nurses, nursing managers and clinical instructors. Twenty per cent of participants were working in the emergency department, and 20 % were working in medical departments. Those attached to hospital A comprised the largest group (n = 116, 38.3 %) followed by hospital B (n = 66, 21.8 %). Three-quarters of participants indicated that they had never attended any formal pain management courses (n = 228, 75 %), and 19 % had enrolled in a pain course (n = 57).

Table 4.1: Demograp	ohic Profile of Nurses
---------------------	------------------------

Socio-demographic Profile	Frequency	%	
Gender			
Male	21	6.93	
Female	281	92.74	
Missing	1	0.33	
Total	303	100.0	
Age Group			
21–30	177	58.42	
31–40	87	28.71	
41–50	19	6.27	
51-60	13	4.4	
Missing	7	2.31	
Total	303	100.0	
Nationality			
Saudi	57	18.81	
Filipino	112	36.96	
Indian	124	40.93	
Chinese	2	0.66	
Indonesian	6	1.98	
Missing	2	0.66	
Total	303	100.0	
Educational Attainment			
Diploma	138	45.54	
Bachelor	159	52.48	
Master	5	1.65	
Missing	1	0.33	
Total	303	100.0	
Religious Affiliation			

Muslim	97	32.01
Christian	193	63.70
Hindu	11	3.63
None (Atheist)	1	0.33
Missing	1	0.33
Total	303	100.0
Work Experience		
6–12 months	7	2.31
1–5 years	117	38.61
5-10 years	96	31.68
10–15 years	41	13.53
15-20 years	14	4 62
20+ years	21	6.93
Missing	7	2.31
Total	303	100.0
Position	200	100.0
Staff Nurse	268	88 45
Head Nurse	23	7 59
Nursing Manager	9	2 97
Clinical Instructor	2	0.66
Missing	1	0.33
Total	303	100.0
Denartment/Ward Assigned	505	100.0
Surgical Department	23	7 59
Medical Department	59	19.47
Emergency Department	62	20.24
	16	5 28
Burn Unit	7	2 31
Pedia Department	18	5.94
NICU (Neonatal Intensive Care Unit)	6	1.08
CCU (Coronary Care Unit)	8	2.64
PICU (Pediatric Intensive Care Unit)	8 22	2.04
AKU (Artificial Kidney Unit)	30	9.90
Nurgery	8	9.90 2.64
Education Department	3	2.04
Operation Department	2	0.00
OB/GVN Department	3	2.05
Nursing Office	5	0.00
	12	3.96
Delivery Room	10	3.30
Missing	5	1.76
Total	303	1.70
Hospital	505	100.0
	116	28.28
Hospital P	66	30.20 21.79
Hospital C	25	21./8 11.55
	33 24	11.33
Hospital D	50 50	11.88
Total	30 202	10.50
	505	100.0
None	229	
none	228	13.23

Once	57	18.81
Iwice	13	4.29
Three Times	2	0.66
Four Times +	2	0.66
Missing	1	0.33
Total	303	100.0

4.3 Nurses' Knowledge and Attitudes Regarding Pain Management in Hail Region Hospitals

The KASRP tool covers many aspects of pain management, and it is widely used to assess nurses' knowledge and attitudes in relation to pain assessment, patient variables, addiction, knowledge in pharmacology and interventions to manage pain.

The frequency distribution of the correct answers (per cent) obtained by n = 303 participants was normal, indicated by the symmetrical bell-shaped histogram (Figure 4.1).



Figure 4.1: Frequency distribution of correct answers (per cent) by n = 303 participants

The scores obtained by each participant ranged from a minimum of 5.0 % to a maximum of 87.5 %, with a mean score of 41.76 % (SD = 9.83). One hundred and fifty-one nurses scored

above the mean and 152 below the mean, as would be expected from a normally distributed variable. The majority (n = 227, 75.0 %) scored between 35 % and 47.5 %. There were a few outliers, reflected by n = 7 (2.3 %) of the participants who scored ≤ 20 % and n = 11 (3.6 %) with scores ≥ 60 % and only two participants obtained a passing score of ≥ 80 %. Table 4.2 shows the distributions of scores for the Knowledge and Attitudes Questionnaire Regarding Pain.

 Table 4.2: Distributions of Scores for the Knowledge and Attitudes Questionnaire Regarding

 Pain

% Correct	N (%)
70 +	3 (1%)
60–69	8 (2%)
50–59	52 (17%)
40–49	136 (44%)
30–39	84 (27%)
Below 30	20 (6%)

As shown in Table 4.2, which presents the distribution of correct answers, the majority (44 %) of participants scored in the range of 40–49 % correct answers, with only 1% receiving scores above 70 and 6 % below 30. It is also evident that only 67 % of participants were able to score 40 % or above, indicating that one in three nurses lacked adequate knowledge on the material included in the questionnaire.

To understand the specific areas of strengths and weaknesses in knowledge, the NKASRP questions were ranked based on the percentage of correct answers obtained, and the results are presented in Table 4.3.

Table 4.3: Ranking of NKASRP Answers from Most Correct to Least Correct

Question	Correct answers <i>n</i> (%)
16. After an initial dose of opioid analgesic is given, subsequent doses should be adjusted in accordance with the individual patient's response: (T).	255 (84.2%)
22. Narcotic/opioid addiction is defined as a chronic neurobiological disease, characterised by behaviours that include one or more of the following: impaired control over drug use, compulsive use, continued use despite harm and craving: (T).	250 (82.5%)
34. The time to peak effect for morphine given intravenouslyis: (15 min.).	233 (76.9%)
15. Patients' spiritual beliefs may lead them to think pain and suffering are necessary: (T).	209 (69.0%)
24. The recommended route administration of opioid analgesics for patients with brief, severe pain of sudden onset such as trauma or postoperative pain is: (intravenous).	209 (69.0%)
25. Which of the following analgesic medications is considered the drug of choice for the treatment of prolonged moderate to severe pain for cancer patients: (morphine).	207 (68.3%)
7. Combining analgesics that work viadifferent mechanisms (e.g. combining an opioid with an NSAID) may result in better pain control with fewer side effects than using a single analgesic agent: (T).	196 (64.7%)
21. Benzodiazepines are not effective pain relievers unless the pain is due to muscle spasm: (T).	189 (62.4%)
27. Analgesics for postoperative pain should initially be given: (around the clock on a fixed schedule).	186 (61.4%)
12. Elderly patients cannot tolerate opioids for pain relief: (F).	181 (59.7%)
31. The most accurate judge of the intensity of the patient's pain is: (the patient).	181 (59.7%)
32. Which of the following describes the best approach for cultural considerations in caring for patients in pain: (Patients should be individually assessed to determine cultural influences).	169 (55.8%)
6. Respiratory depression rarely occurs in patients who have been receiving stable doses of opioids over a period of months: (T).	175 (57.8%)
14. Children less than 11 years old cannot reliably report pain, so nurses should rely solely on the parent's assessment of the child's pain intensity: (F).	156 (51.5%)
29. The most likely reason a patient with pain would request increased doses of pain medication is: (The patient is experiencing increased pain).	155 (51.2%)
2. Astheir nervous system is underdeveloped, children under two years of age have decreased pain sensitivity and limited memory of painful experiences: (F).	151 (49.8%)
20. Anticonvulsant drugs such as gabapentin (Neurontin) produce optimal pain relief after a single dose: (F).	150 (49.5%)
8. The usual duration of analgesia of 1–2 mg morphine IV is 4–5 hours: (F).	131 (43.2%)
35. The time to peak effect for morphine given orally is: (1–2 hours).	124 (40.9%)
18. Vicodin (hydrocodone 5 mg + acetaminophen 500 mg) PO is approximately equal to 5–10 mg of morphine PO: (T).	122 (40.3%)
9. Research shows that promethazine (Phenergan) and hydroxyzine (Vistaril) are reliable potentiators of opioid analgesics: (F).	113 (37.3%)
30. Which of the following is useful for treatment of cancer pain: (All of the above).	109 (36.0%)
3. Patients who can be distracted from pain usually do not have severe pain: (F).	108 (35.6%)
5. Aspirin and other non-steroidal anti-inflammatory agents are NOT effective analgesics for painful bone metastases:(F).	102 (33.7%)
1. Vital signs are always reliable indicators of the intensity of a patient's pain: (F).	99 (32.7%)
38 A. On the patient's record, you must mark his pain on the scale below. Circle the number that represents your assessment of Robert's pain: (8).	97 (32.0%)
13. Patients should be encouraged to endure as much pain as possible before using an opioid: (F).	94 (31.0%)
26. Which of the following IV doses of morphine administered over a four-hour period would be equivalent to 30 mg of oral morphine given q 4 hours: (Morphine 10 mg IV).	91 (30.0%)
33. How likely is it that patients who develop pain already have an alcohol and/or drug abuse problem: (5–15 per cent).	86 (28.4%)

11. Morphine has a dose ceiling (i.e. a dose above which no greater pain relief can be obtained): (F).	85 (28.1%)
17. Giving patients sterile water by injection (placebo) is a useful test to determine if the pain is real: (F).	71 (23.4%)
10. Opioids should not be used in patients with a history of substance abuse:(F).	66 (21.8%)
4. Patients may sleep despite severe pain: (T).	59 (19.5%)
28. A patient with persistent cancer pain has been receiving daily opioid analgesics for twomonths. Yesterday the patient was receiving morphine 200 mg/hour intravenously. Today he has been receiving 250 mg/hour intravenously. The likelihood of the patient developing clinically significant respiratory depression in the absence of new comorbidity is: (less than 1 per cent).	56 (18.5%)
36. Following abrupt discontinuation of an opioid, physical dependence is manifested by the following: (sweating, yawning, diarrhoea and agitation with patients when the opioid is abruptly discontinued).	55 (18.2%)
23. The recommended administration of opioid analgesics for patients with persistent cancer- related pain is: (oral).	44 (14.5%)
38 B. Check the action you will take at this time: (Administer morphine 3 mg IV now).	38 (12.5%)
37 A. On the patient's record, you must mark his pain on the scale below. Circle the number that represents your assessment of Andrew's pain: (8).	28 (9.2%)
19. If the source of the patient's pain is unknown, opioids should not be used during the pain evaluation period, as this could mask the ability to correctly diagnose the cause of pain: (F).	22 (7.3%)
37 B. Check the action you will take at this time: (Administer morphine 3 mg IV now).	9 (3.0%)

T= True, F= False

Ten questions at the top of Table 4.3 were correctly answered by the majority of nurses (n = 180, 59.7 % to n = 255, 84.2 %). Nine of these 10 questions required factual knowledge about the use of analgesics (e.g. opioids, morphine, NSAID, Benzodiazepines), suggesting that nurses were quite knowledgeable in this area. Ten questions at the bottom of Table 4.3 were correctly answered by less than one-quarter of nurses (n = 9, 3.0 % to n = 23.4 %). These questions did not require as much factual knowledge about the use of analgesics as the top 10. This showed that the questions that the nurses found most difficult to answer were those that required them to: make decisions or personal value judgements, assess the value of a particular treatment, determine what action should be taken in a given situation, assess the outcome of a particular treatment, and evaluate the severity and source of pain. In general, 25 questions were answered incorrectly by more than 50 % of participants.

4.4 Effect of Demographic and Cultural Factors on the Delivery of Effective Pain Management

The next 10 sections sequentially address each of the null hypotheses (H_01 to H_010).

4.4.1 H0₁: Effect of Gender

As an overwhelming fraction of the total number of participants (N = 303) comprised females (n = 281, 92.7 %) (see Table 4.4), the results of this study may not necessarily be representative of the views with respect to male nurses. One participant did not disclose his or her gender (missing value). The percentage of correct scores stratified by gender were approximately normally distributed (Figure 4.2).

Table 4.4. Frequency Distribution of FatterpartGendern%Male216.9Female28192.7Missing1.3

Total

303

Table 4.4: Frequency Distribution of Participants by Gender

100.0



Figure 4.2: Frequency distributions of correct answers (%) by gender

The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.5–4.7. The mean score for n = 21 male participants (M = 38.57, SD = 6.30) was slightly lower than that of n = 281 females (M = 42.00, SD = 10.03). The variances were homogenous, indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should not be rejected (F(1,300) = 2.380, p = .124). The effect size ($\eta^2 = .008$) was negligible. Consequently, it is inferred that gender had no statistically or clinically significant effect on nurses' levels of knowledge regarding pain management. Clinical significance does not depend on numbers as statistically significant rather than it is a matter of judgment.

Table 4.5: Descriptive Statistics for Correct Answers (%) by Gender

Gender	М	SD	п	
Male	38.57	6.30	21	
Female	42.00	10.03	281	
Total	41.76	9.85	302	

Table 4.6: Levene's Test for Equality of Variances

F	df1	df2	р			
3.063	1	300	.081			
Source	Type III Sum of Squares	df	Mean Square	F	р	Effect Size η^2
-----------------	-------------------------	-----	-------------	-------	------	----------------------
Gender	229.93	1	229.93	2.380	.124	.008
Error	28987.39	300	96.62			
Total	555956.25	302				
Corrected Total	29217.32	301				

Table 4.7: ANOVA Test for Between-Subjects Effects

4.4.2 H₀2. Effect of Age

The reported ages of participants (Table 4.8) ranged from 21 to 60 years. Over half (n = 177, 58.4 %) were in the youngest age group (21–30 years). Seven participants did not disclose their ages (missing values). The percentage of correct scores stratified by age were approximately normally distributed (Figure 4.3). The curve showed a slight rightward shift in the 41–60-year age group. The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.9–4.12. The mean score (M = 47.73, SD = 11.82) for n = 32 older participants (aged 41–60) was higher than that for younger participants (M = 40.25, SD = 8.36 for n = 177 at age 21–30 to M = 42.78, SD = 10.97 for n = 87 at age 31–40).

Age	n	%
21-30	177	58.4
31–40	87	28.7
41–50	19	6.3
51-60	13	4.3
Missing	7	2.3
Total	303	100.0

Table 4.8: Frequency Distribution of Participants by Age



Figure 4.3: Frequency distributions of correct answers (%) by age

Age (Collapsed)	М	SD	n	
21-30	40.25	8.36	177	1
31-40	42.78	10.97	87	
41-60	47.73	11.82	32	
Total	41.81	9.85	296	1

Table 4.9: Descriptive Statistics for Correct Answers (%) by Age

Table 4.10: Levene's Test for Equality of Variances

F	df1	df2	р
1.547	2	293	.215

Table 4.11: ANOVA Test for Between-Subjects Effects

Source	Type III Sum of Squares	df	Mean Square	F	p	Effect Size η^2
Age	1634.65	2	817.32	8.874	<.001*	.057
Error	26985.86	293	92.10			
Total	545987.50	296				
Corrected Total	28620.52	295				

Note: * Significant at $\alpha = .05$

Table 4.12: Scheffé PostHoc Multiple Comparison Test

Age	п	Homogeneous Subsets		
		1	2	
21-30	177	40.25	•	
31-40	87	42.78		
41–60	32		47.73	

The variances were homogenous, indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F (2,293 = 8.874, p < .001). However, the effect size ($\eta^2 = .057$) was relatively small. The Scheffé test divided the mean scores into two homogeneous subsets. The mean score for the 41–60 age group was significantly higher than the mean scores for both the 31–40 and 21–30 age groups.

These results provide evidence that the age of participants had a statistically and clinically significant effect on the scores. Thus, the superiority in knowledge regarding pain management was directly correlated with the age of the nurse.

4.4.3 H₀3. Effect of Nationality

The self-reported nationalities of the participants (Table 4.13) included Saudi, Filipino, Indian, Chinese and Indonesian. Those of Indian (n = 124, 40.9 %) and Filipino (n = 112, 37.0 %) origin, taken together, represented more than three-quarters of participants. Two participants did not disclose their nationalities (missing values). The percentage of correct scores for each nationality were approximately normally distributed (Figure 4.4). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.14– 4.17.

Nationality	n	%
Saudi	57	18.8
Filipino	112	37.0
Indian	124	40.9
Chinese	2	0.7
Indonesian	6	2.0
Missing	2	0.7
Total	303	100.0

 Table 4.13: Frequency Distribution of Participants by Nationality

The number of Chinese and Indonesian participants was too small to include them as separate groups for the purpose of conducting ANOVA. Consequently, they were pooled with the Indian nurses in a category called 'Other Asian'.



Figure 4.4: Frequency distributions of correct answers (%) by nationality

Nationality	М	SD	п	
Saudi	38.77	10.19	57	
Filipino	44.44	11.13	112	
Other Asian	40.95	7.71	132	
Total	41.84	9.79	301	

Table 4.15: Levene's Test for Equality of Variances

F	df1	df2	р
2.748	2	298	.066

Table 4.16: ANOVA Test for Between-Subjects Effects

Source	Type III Sum of Squares	df	Mean Square	F	p	Effect Size η^2
Nationality	1400.07	2	700.04	7.630	.001*	.049
Error	27342.03	298	91.75			
Total	555556.25	301				
Corrected Total	28742.11	300				
N. + G: : : C			•	•		

Note: * Significant at $\alpha = .05$

Table 4.17: Scheffé PostHoc Multiple Comparison Test

Nationality	Ν	Subset	
		1	2
Saudi	57	38.77	
Other Asian	132	40.95	
Filipino	112		44.44

The mean score (M = 44.44, SD = 11.13) for n = 112 Filipino participants was higher than that for n = 57 Saudi participants (M = 38.77, SD = 10.19) and n = 132 other Asian participants (M = 40.95, SD = 7.71). The variances were homogenous, indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F (2, 298 = 7.630, p < .001). However, the effect size ($\eta^2 = .049$) was relatively small. The Scheffé test divided the mean scores into two homogeneous subsets. The mean score for the Filipino participants was significantly higher than the mean scores for the other two ethnic groups. Consequently, evidence shows that ethnicity had a statistically and clinically significant effect on the scores. Filipino nurses exhibited the highest levels of knowledge regarding pain management.

4.4.4 H₀4. Effect of Academic Qualification

The self-reported qualifications of the participants (Table 4.18) included Diplomas,

Bachelor's and Master's degrees. The predominant group, representing 98 % of participants, had Bachelor's degrees (n = 159, 52.5 %), followed by Diploma holders (n = 138, 45.5 %). One participant did not disclose his or her educational qualification (missing value). The percentage of correct scores stratified by level of education were approximately normally distributed (Figure 4.5). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.19–4.21.

Table 4.18: Frequency Distribution of Participants by Academic Qualification

Qualification	n	%
Diploma	138	45.5
Bachelor's Degree	159	52.5
Master's Degree	5	1.7
Missing	1	0.3
Total	303	100.0

The mean score (M = 42.42, SD = 9.72) for n = 164 participants with Bachelor's/Master's degrees was slightly higher than that for n = 138 participants with Diplomas (M = 40.98, SD = 9.99). The variances were homogenous, indicated by p > .05 for Levene's test.



Figure 4.5: Frequency distributions of correct answers (%) by academic qualification

Table 4.19: Descriptive Statistics for Correct Answers (%) by Academic Qualification

Degree	М	SD	п
Diploma	40.98	9.99	138
Bachelor/Master	42.42	9.72	164
Total	41.76	9.85	302

Table 4.20: Levene's Test for Equality of Variances

F	df1	df2	р
0.329	1	300	0.567

ANOVA indicated that the null hypothesis should not be rejected (F(1, 300 = 1.617, p = .205)). The effect size ($\eta^2 = .005$) was negligible. Consequently, the results indicate that academic qualifications had no statistically or clinically significant effects on nurses' levels of knowledge regarding pain management.

Source	Type III Sum of Squares	df	Mean Square	F	р	Effect Size η^2
Qualification	156.590	1	156.590	1.617	.205	.005
Error	29060.732	300	96.869			
Total	555956.250	302				
Corrected Total	29217.322	301				

Table 4.21: ANOVA Tests for Between-Subjects Effects

4.4.5 H₀5. Effect of Religion

According to the self-reported religions of the participants (Table 4.22), they belonged to the Muslim (n = 97, 32.0 %), Christian (n = 193, 63.7 %) and Hindu (n = 11, 3.6 %) faiths. Two participants did not disclose their religion (missing values). The percentage of correct scores stratified by religion were approximately normally distributed (Figure 4.6). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.23–4.26.

Religion	n	%	-
Christian	193	63.7	
Muslim	97	32.0	
Hindu	11	3.6	
Atheist	1	0.3	
Missing	1	0.3	
Total	303	100.0	_

Table 4.22: Frequency Distribution of Participants by Religion



Figure 4.6: Frequency distributions of correct answers (%) by religion

1 able 4.23:	Descriptive	Statistics 1	for Cor	rect Ansv	vers (%)	by	Religion

Religion	Μ	SD	п
Christian	42.80	9.59	193
Muslim	39.69	10.29	97
Hindu	41.82	8.81	11
Total	41.76	9.87	301

Table 4.24: Levene's Test for Equality of Variances

F	df1	df2	р
0.211	2	298	0.810

The mean score (M = 42.80, SD = 9.59) for n = 193 Christian participants was higher than that for n = 97 Muslim participants (M = 39.69, SD = 10.29) and n = 11 Hindu participants (M = 41.82, SD = 8.81). The variances were homogenous, indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F(2, 298 = 3.240, p = .040)). However, the effect size ($\eta^2 = .021$) was negligible. The Scheffé test divided the mean scores into two homogeneous subsets. Consequently, it was concluded that religion had a statistically significant but not a clinically significant effect on the scores. Christian nurses exhibited the highest levels of knowledge regarding pain management; however, the mean score difference (less than 2 % between Christians, Muslims and Hindus) had limited importance in the context of this research. Further, the marginally higher score of the Christian group may reflect the country-of-origin effect, as the majority of Christians are also Filipino.

Table 4.25: ANOVA Test for Between-Subjects Effects

Type III Sum of Squares	df	Mean Square	F	р	Effect Size η^2
623.30	2	311.65	3.248	.040*	.021
28593.48	298	95.95			
554150.00	301				
29216.78	300				
	Type III Sum of Squares 623.30 28593.48 554150.00 29216.78	Type III Sum of Squares df 623.30 2 28593.48 298 554150.00 301 29216.78 300	Type III Sum of SquaresMean Square623.302311.6528593.4829895.95554150.0030129216.78	Type III Sum of SquaresMean SquareF623.302311.653.24828593.4829895.95554150.00554150.0030129216.78300	Type III Sum of SquaresMean SquareFp623.302311.653.248.040*28593.4829895.95.040*554150.00301.040*.040*29216.78300.040*.040*

* Note: Significant at $\alpha = .05$

Table 4.26: Scheffé PostHoc Multiple Comparison Test

Religion	n	Homoger	Homogeneous Subsets		
		1	2		
Muslim	97	39.69			
Hindu	11	41.82			
Christian	193		42.80		

4.4.6 H₀6. Effect of Experience

The self-reported experience (Table 4.27) ranged from 6–12 months to over 20 years. The majority, representing 70.3 % of participants, belonged to those with 1–5 years of experience (n = 117, 38.6 %) and 5–10 years experience (n = 96, 31.7 %). Seven participants did not disclose their work experience duration (missing values). The percentage of correct scores stratified by experience were approximately normally distributed (Figure 4.7). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.28–4.31. The mean scores for nurses with the longest experience (M = 45.24, SD = 7.45 for n = 21 with 20+

years experience and M = 46.78, SD = 10.98 for n = 14 with 15–20 years experience) were the highest. Participants with the least experience had the lowest mean score (M = 31.43, SD = 12.40 for n = 7 with 6–12 months experience).

Experience	n	%
6–12 months	7	2.3
1-5 years	117	38.6
5-10 years	96	31.7
10-15 years	41	13.5
15-20 years	14	4.6
20+ years	21	6.9
Missing	7	2.3
Total	303	100.0

Table 4.27: Frequency Distribution of Experience



Figure 4.7: Frequency distributions of correct answers (%) by experience

The variances were homogenous, as indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F (5, 290 = 3.286, p = .007). However, the effect size ($\eta^2 = .054$) was relatively small. The Scheffé test divided the mean scores into two

homogeneous subsets. The two groups of nurses with the longest experience (15–20 years and 20+ years) collectively achieved higher mean scores than the other four groups classified by experience. Therefore, it is concluded that the number of years of work experience had a statistically and clinically significant effect, with nurses who had the longest experience performing the best in the knowledge test regarding pain management.

Experience	М	SD	n
6–12 months	31.43	12.40	7
1-5 years	40.89	8.87	117
5-10 years	42.58	9.84	96
10-15 years	40.92	11.35	41
15-20 years	46.78	10.98	14
20+ years	45.24	7.45	21
Total	41.81	9.85	296

Table 4.28: Descriptive Statistics for Correct Answers (%) by Experience

Table 4.29: Levene's Test for Equality of Variances

F	df1	df2	р
.739	5	290	.602

Table 4.30: ANOVA Test for Between-Subjects Effects

	Type III Sum					Effect Size
Source	of Squares	df	Mean Square	F	р	η^2
Experience	1534.76	5	306.95	3.286	.007*	.054
Error	27085.76	290	93.39			
Total	545987.50	296				
Corrected Total	28620.52	295				

Note: * Significant at $\alpha = .05$

Table 4.31: Scheffé PostHoc Multiple Comparison Test

		Homogeneous Subsets		
Experience	n	1	2	
6-12 months	7	31.429	· · · · · ·	
1-5 years	117	40.897		
10-15 years	41	40.915		
5-10 years	96	42.578		
20+ years	21		45.238	
15-20 years	14		46.786	

4.4.7 H₀7. Effect of Position

According to the self-reported positions (Table 4.32), participants belonged to one of the following professional positions: staff nurse, head nurse, nursing manager or clinical instructor. The most common group consisted of staff nurses (n = 268, 88.4 %). One participant did not disclose his or her position (missing value). The percentage of correct scores stratified by experience were approximately normally distributed (Figure 4.8). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.33–4.35. ANOVA indicated that the null hypothesis should not be rejected (F (1,300 = 1.929, p = .166). The effect size (η^2 = .006) was also negligible. Consequently, it is concluded that the position in the professional hierarchy of nurses had no statistically or clinically significant effect on their knowledge of pain management.

Position	n	%
Staff Nurse	268	88.4
Head Nurse	23	7.6
Nursing Manager	9	3.0
Clinical Instructor	2	0.7
Missing	1	0.3
Total	303	100.0

Table 4.32: Frequency Distribution of Participants by Position



Figure 4.8: Frequency distributions of correct answers (%) by position

Table 4.33: Descriptive Statistics for Correct Answers (%) by Position

Position	М	SD	n
Staff Nurse	41.48	9.91	268
Head Nurse/Manager/Instructor	43.97	9.21	34
Total	41.76	9.85	302

Table 4.34: Levene's Test for Equality of Variances

F	df1	df2	р
.038	1	300	.845

Table 4.35: ANOVA Test for Between-Subjects Effects

Effect	Type III Sum o Squares	of df	Mean Square	F	р	Effect Size η^2
Position	186.68	1	186.677	1.929	.166	.006
Error	29030.65	300	96.769			
Total	555956.25	302				
Corrected Total	29217.32	301				

4.4.8 H₀8. Effect of the Department of Placement

On the basis of a one's principal workplace, participants reported that they worked in one of the 17 different hospital departments (Table 4.36). The most frequent groups (n = 174, 57.4 %) were in the emergency (n = 63, 20.5 %), medical (n = 59, 19.5 %), artificial kidney unit (n = 30, 9.9 %) and surgical (n = 23, 7.6 %) departments. Five participants did not disclose their departments (missing values). The percentage of correct scores stratified by department were approximately normally distributed (Figure 4.9). The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.37–4.40.

Department	п	%
Surgical Department	23	7.6
Medical Department	59	19.5
Emergency Department	62	20.5
ICU (Intensive Care Unit)	16	5.3
Burn Unit	7	2.3
Pediatric Department	18	5.9
NICU (Neonatal Intensive Care Unit)	6	2.0
CCU (Coronary Care Unit)	8	2.6
PICU (Pediatric Intensive Care Unit)	22	7.3
AKU (Artificial Kidney Unit)	30	9.9
Nursery	8	2.6
Education Department	2	0.7
Operation Room	8	2.6
OB/GYN Department	2	0.7
Nursing Office	5	1.7
OPD (Outpatients Department)	12	4.1
Delivery Room	10	3.3
Missing	5	1.7
Total	303	100.0

Table 4.36: Frequency Distribution of Participants by Departments



Figure 4.9: Frequency distributions of correct answers (%) by departments

Department	M	SD	n
Surgical Department	39.13	10.38	23
Medical Department	41.44	10.25	59
Emergency Department	41.05	8.95	62
ICU (Intensive Care Unit)	44.53	5.93	16
Burn Unit	38.21	5.54	7
Pediatric Department	37.50	13.06	18
NICU/OB/GYN/Nursery/Delivery Room	47.98	11.38	26
CCU (Coronary Care Unit)	47.81	14.04	8
PICU (Pediatric Intensive Care Unit)	40.57	7.94	22
AKU (Artificial Kidney Unit)	39.17	7.97	30
Education Department/Nursing Office	50.00	12.16	7
Operation Room	40.31	6.61	8
OPD (Outpatients Department)	41.88	3.86	12
Total	41.67	9.86	298

Table 4.37: Descriptive Statistics for Correct Answers (%) by Departments

Table 4.38: Levene's Test for Equality of Variances

F	df1	df2	p
1.650	12	285	.078

Table 4.39: ANOVA Test for Between-Subjects Effects

Type III Sum						Effect Size
Source	of Squares	df	Mean Square	F	р	η^2
Department	2755.968	12	229.664	2.508	.004*	.096
Error	26094.724	285	91.560			
Total	546281.250	298				
Corrected Total	28850.692	297				
M + 40						-

Note: * Significant at $\alpha = .05$

The variances were homogenous, indicated by p > .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F (12, 285 = 2.508, p = .004). However, the effect size (η^2 = .096) was relatively small. The Scheffé test divided the mean scores into two homogeneous subsets. The nurses in the ICU (n = 16, M = 44.53, SD = 5.93), CCU (n = 8, M= 47.81, SD = 14.04), NICU/OB/GYN/nursery/delivery room (n = 26, M = 47.98, SD =11.38) and education department/nursing office (n = 7, M = 50.00, SD = 12.16) achieved higher mean scores than any other departments. Evidence indicates that statistically and clinically significant differences existed between participants placed in different departments in the hospital system. Nurses associated with intensive care, coronary care, neonatal care, obstetrics/gynaecology, nursery care, education and the nursing office collectively exhibited significantly higher levels of knowledge regarding pain management compared with those associated with other departments.

Department	n	Homogenous S	Subsets
		1	2
Pediatric Department	18	37.50	
Burn Unit	7	38.21	
Surgical Department	23	39.13	
AKU (Artificial Kidney Unit)	30	39.17	
Operation Room	8	40.31	
PICU (Pediatric Intensive Care Unit)	22	40.57	
Emergency Department	62	41.05	
Medical Department	59	41.44	
OPD (Outpatients Department)	12	41.88	
ICU (Intensive Care Unit)	16		44.53
CCU (Coronary Care Unit)	8		47.81
NICU/OB/GYN/Nursery/Delivery Room	26		47.98
Education Department/Nursing Office	7		50.00

Table 4.40: Scheffé PostHoc Multiple Comparison Test

4.4.9 H₀9. Effect of Hospitals

Participants reported that they worked in one of five different hospitals (Table 4.41). The most frequent groups (n = 116, 38.3 %) were Hospital A with n = 66, 21.8 %, in Hospital B. One of the other three hospitals was the workplace of 11.6–16.5 % of participating nurses. The percentage of correct scores stratified by hospital were approximately normally distributed (Figure 4.10).

Hospitals	п	%
Hospital A	116	38.3
Hospital B	66	21.8
Hospital C	35	11.6
Hospital D	36	11.9
Hospital E	50	16.5
Total	303	100.0

Table 4.41: Frequency Distribution of Participants by Hospitals



Figure 4.10: Frequency distributions of correct answers (%) by hospitals

The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.42–4.45. The variances were homogenous, indicated by p> .05 for Levene's test. ANOVA indicated that the null hypothesis should be rejected (F (4, 298 = 5.213, p <. 001). However, the effect size (η^2 = .065) was relatively small. The Scheffé test divided the mean scores into two homogeneous subsets. The n = 50 nurses at Hospital E achieved higher mean scores (M = 46.85, SD = 5.93) than nurses in any other hospitals (from M = 38.86, SD = 8.32 for n = 66 at Hospital B to M = 42.36, SD = 7.63 for n = 36 at Hospital D).

These results conclude that there were statistically and clinically significant differences between nurses' knowledge depending on the hospital they worked at. Participants at Hospital E exhibited significantly higher levels of knowledge regarding pain management compared to those at Hospitals A, B, C and D. However, this result must be considered in the context of any potential confounding effects of other variables—for example, there being a relatively larger number of Filipino nurses (who scored highly) working at hospital E.

Hospitals	M	SD	п	
Hospital A	41.19	10.15	116	
Hospital B	38.86	8.32	66	
Hospital C	41.21	7.08	35	
Hospital D	42.36	7.63	36	
Hospital E	46.85	12.14	50	
Total	41.76	9.84	303	

Table 4.42: Descriptive Statistics for Correct Answers (%) by Hospitals

Table 4.43: Levene's Test for Equality of Variances

F	df1	df2	р	
1.517	4	298	.197	

Table 4.44: ANOVA Test for Between-Subjects Effects

Source	Type III Sum of Squares	df	Mean Square	F	р	Effect Size η^2
Hospitals	1910.810	4	477.702	5.213	<.001*	.065
Error	27309.611	298	91.643			
Total	557556.250	303				
Corrected Total	29220.421	302				

Note: * Significant at $\alpha = .005$

Table 4.45: Scheffé PostHoc Multiple Comparison Test

Nationality	п	Homogenous Subsets		
		1	2	
Hospital B	66	38.86		
Hospital A	116	41.18		
Hospital C	35	41.21		
Hospital D	36	42.36		
Hospital E	50		46.85	

4.4.10 H₀10. Effect of Participation in Pain Courses

Participants reported that they attended at least one of up to 10 different formal instruction courses concerned with pain management (Table 4.46), with one missing value. Most (n = 228, 75.2 %) had not attended a course, while n = 57, 18.8 % had attended one course and n = 228, 75.2 %)

5, 1.6 % had attended more than two. The percentage of correct scores stratified by hospital were approximately normally distributed (Figure 4.11).

Courses	Frequency	%
None	228	75.2
Once	57	18.8
Twice	13	4.3
Three Times or more	4	1.3
Missing	1	0.3
Total	303	100.0

 Table 4.46: Frequency Distribution of Participants by Courses



Figure 4.11: Frequency distributions of correct answers (%) by courses

The descriptive statistics and results of ANOVA and associated tests are presented in Tables 4.47–4.49. The mean score was lowest (M = 41.36, SD = 9.80) for n = 228 nurses who had not attended a course. The mean score was highest (M = 43.29, SD = 9.31) for n = 57 nurses who had attended one course. The variances were homogenous, indicated by p> .05 for Levene's test. ANOVA indicated that the null hypothesis should not be rejected (F (2, 299 =

.882, p = .415). The effect size ($\eta^2 = .006$) was negligible. These results show that attendance at courses had no statistically or clinically significant effects on nurses' levels of knowledge regarding pain management.

Courses	М	SD	Ν	
None	41.36	9.80	228	
One	43.29	9.31	57	
Two or More	42.06	12.22	17	
Total	41.76	9.85	302	

Table 4.47: Descriptive Statistics for % Correct Answers by Courses

Table 4.48: Levene's Test for Equality of Variances

F	df1	df2	р
.119	2	299	.887

Table 4.49: ANOVA Test for Between-Subjects Effects

Source	Type III Sum of Squares	df	Mean Square	F	р	Effect Size η^2
Courses	171.398	2	85.699	.882	.415	.006
Error	29045.924	299	97.144			
Total	555956.250	302				
Corrected Total	29217.322	301				



Figure 4.12: Comparison of effect sizes by demographic characteristics

Statistical inferences were obtained using multiple ANOVA tests; as a result, this could potentially be compromised by Type I and/or Type II errors. Clinical significance, rather than statistical significance at $\alpha = .05$, is used to address this question. The effects of the 10 demographic characteristics on nurses' levels of knowledge regarding pain management are compared visually using a barchart (Figure 4.12). A vertical line is drawn at $\eta^2 = .04$, below which the effects of qualification, position, courses, gender and religion are assumed to be minimal or negligible (Ferguson, 2009). The values of η^2 between .05 and .1 for the other five demographic characteristics reflected relatively small effect sizes; nevertheless, they were clinically significant results with practical implications for pain management.

Chapter 5: Qualitative Data Analysis

5.1 Introduction

Analysis was conducted on the qualitative data obtained from the responses of 28 participants to the open-ended, semi-structured interview questions. Statements made by the participants that were relevant to the research questions were considered reflective of their experiences and perceptions (Moustakas, 1994). A qualitative process of coding and thematising (Creswell, 2009; Merriam, 2009) was used to analyse the statements, from which several themes were identified with regard to knowledge, attitudes and factors affecting nurses' provision of pain management to patients.

The coding process was completed through the systematic identification and categorisation of participants' responses to the open-ended interview questions, and the codes were grouped according to content using a combination of inductive and deductive reasoning, allowing for the identification of similarities between responses (Merriam, 2009). The thematic categories (and subcategories) were then further reviewed and compared, yielding the overall themes that were representative of the different perceived elements central to the phenomenon for the group of participants.

In addition to the presentation of the thematic categories with response frequencies, verbatim textual responses of participants are included in the findings to highlight and clarify the key themes or concepts revealed, thereby providing an in-depth understanding of the themes and allowing for a more vivid portrayal of participants' experiences and perceptions (Creswell, 2009). NVivo 9® qualitative analysis software was used to assist in the coding and development of themes and patterns from the data by assisting in the classification, sorting

117

and arranging of information across the data. The concluding themes that were revealed represent the perceptions of the group as a whole and are presented according to the relevant associated research question. The qualitative analysis was conducted by the researcher and then checked by the supervisors.

5.2 Overview of Participants

The interview sample consisted of 28 nurses from various departments within the facility. They represented different age groups, nationalities, religious beliefs, years of experience and career levels. The variety of demographic characteristics is presented in Table 5.1.

Participant	Gender	Age	Nationality	Religion	Years Worked
A	F	27	Expatriate	Christian	2
В	F	29	Expatriate	Christian	7
С	F	38	Expatriate	Christian	17
D	F	25	Expatriate	Christian	4
Е	F	28	Expatriate	Christian	7
F	F	26	Expatriate	Christian	7
G	F	39	Expatriate	Christian	8
Н	F	58	Expatriate	Christian	10
Ι	F	40	Expatriate	Christian	15
J	F	39	Expatriate	Christian	19
Κ	F	27	Expatriate	Hindu	6
L	F	26	Expatriate	Christian	4
М	М	28	Expatriate	Muslim	7
Ν	F	60	Expatriate	Christian	30
0	F	55	Expatriate	Christian	30
Р	F	25	Expatriate	Christian	3
Q	М	42	Expatriate	Muslim	14
R	М	40	Expatriate	Christian	19
S	М	34	Expatriate	Christian	10
Т	F	33	Expatriate	Christian	9
V	F	46	Expatriate	Christian	22
W	F	33	Expatriate	Christian	8
Х	F	50	Expatriate	Christian	25
Y	F	55	Expatriate	Christian	30
Z	М	32	Saudi	Muslim	10
A2	М	29	Saudi	Muslim	7
B2	М	29	Saudi	Muslim	7
C2	М	31	Saudi	Muslim	10

Table 5.1: Demographic Characteristics

5.3 Findings

Through the process of data analysis, common relevant occurrences (responses, statements, or expressed perceptions or thoughts) of the interview participants were coded and documented. These occurrences were then categorised into five related thematic categories, which are presented individually:

- 1. Perceived knowledge level and familiarity with pain management and medications
- 2. Current pain management process and practice
- 3. Nursing pain assessment
- 4. Barriers and limitations to optimal pain management
- 5. Factors to improve pain management.

Table 5.2 summarises the themes, subthemes and meaning units arising during the interview.

Thematic categories	Subthemes	Meaning Unit
1. Perceived knowledge level and familiarity with pain management and medications	Perceptions and satisfaction with level of knowledge of pain management	-Satisfied - Knowledge of newcomers not good - Nurses are not knowledgeable or well trained
	Familiarity with, and knowledge of, pain medications used	 Brand name differs but the generic name is the same Lack of using patient-controlled analgesia(PCAs)
	Interest in learning about pain management	-Interested to learn more - Kept up with learning in India because had monthly tests
2. Current pain management processes and practices	Alternative forms of management used (non- pharmaceutical)	 Some nurses are using non- pharmaceutical interventions Some nurses do not effectively assess pain or try alternative pain management
	Level of satisfaction with current pain management practices and processes	 Satisfied Not satisfied because nurses are not using assessment tools and alternative pain management methods

Table 5.2:	Summary	of Themes
------------	---------	-----------

	Nursing responsibility in	-Nurses have significant
	pain management	responsibility because they are with
		patients most of the time
3. Nursing pain	Nurses' pain assessment in	-Wong Baker and numerical scale for
assessment	terms of the tools used	assessment
		-Do not use tools for assessment
		-No form for pain assessment
	Training for pain	- No training had been given
	assessment	- Self-learning about the use of tools
	Perceived efficacy of the	- Effective
	assessment process	
4. Barriers and limitations	Language and	-Language; communication barriers
to optimal pain	communication barriers	-Different accents
management	Workload due to nurse staff	-Shortage of nurses and/or workload
	shortages	-
	Cultural, religious and	-Cultural factors (shame on men to
	other factors	complain of pain)
		-Religious factors (rewards from
		god)
	Lack of education and	-No time to update knowledge as a
	nurses' updated knowledge	result of high workload
	regarding pain management	
5. Factors to improve	Perceived education and	-Cultural orientation courses
pain management	development needs	- Pain management courses
		-Self-learning
		-Uniform protocol or policy
		-Improved assessment guidelines
		-Assessment form should be
		implemented to ensure
		documentation
		-Less workload; more nursing staff

5.3.1 Perceived Knowledge and Familiarity with Pain Management and

Medications

The first thematic category revealed in the analysis was the perceived level of knowledge of nurses with regard to pain management and medications used. The following three subthemes were included under this category: perceptions and satisfaction with level of knowledge of pain management; familiarity with, and knowledge of, pain medications used; and interest in learning about pain management.

A majority of the nurses that were interviewed perceived that both their own and other nurses' knowledge levels with regards to pain management is satisfactory, and that they are familiar

with the medications used at this facility, as most are the same as those used in their home country (if other than Saudi Arabia). The assertion that most nurses are well equipped with knowledge of pain management persisted among all respondents. In addition, the majority of participants expressed interest in learning more about pain management and some of them mentioned that they kept up with learning because they had monthly tests in their country.

5.3.1.1 Perceptions and Satisfaction with Level of Knowledge of Pain Management.

As far as the level of satisfaction with their own and others' knowledge of pain management, most participants felt satisfied. Their responses confirmed that they believed they held basic knowledge concerning pain management. However, they acknowledged that there was a possibility of such basic knowledge being inadequate. The level of satisfaction in their knowledge was identified by other factors, as revealed in their responses. For example, participant F stated that 'Well, regarding my knowledge about pain and pain management, I think it's fairly good'. The statement presents the perception that the participant considered his or her level of knowledge was, to a reasonable degree, good or adequate.

Another nurse (participant E) described the same sentiment, as well as the desire to increase the level of personal knowledge with regard to pain management, stating that 'It's okay, but I want to improve also'.

Unlike participant F, who did not come out strongly on whether 'fairly good' meant he or she needed more training on pain management, participant E was more candid, acknowledging that while he or she had knowledge of pain management, it needed to be improved. In an effort to reinforce that they had basic knowledge on pain management, some participants were more detailed and specific in their responses, as seen by the following response by participant L: 'I can't say that I have full knowledge, but according to experience of mine and also my study in my country also, I had knowledge how to control the pain, how to manage the pain, how to alleviate pain'.

The participant proceeded to detail aspects of the control, management and alleviation of pain to indicate the level of awareness in the field. The statement denotes confidence in the knowledge of pain management. However, it lacks in precision in the level of satisfaction.

While another participant C2 offered that his or her difference in knowledge level was directly tied to the level of education received (i.e. diploma versus Bachelor's degree). The indication is that diploma-level nurses are less informed concerning pain management than their counterparts who have a Bachelor's degree. Participant C2 stated:

Before I had a diploma. So my knowledge for pain was really poor and I couldn't manage the pain, because I didn't have the information on how to manage the pain and how to use the pain scale...But now, when I got my Bachelor's degree in nursing I feel I am more satisfied with myself and the patients themselves also.

The satisfaction level regarding pain management was also inadequate. However, the level of satisfaction increased when the respondent received more training and obtained a Bachelor degree in nursing. Subsequently, the increased level of participant satisfaction was linked to a positive effect in career progression and thus improved the provision of services to patients.

Participant B2 stated that nurses' knowledge may vary depending on their education level. The participant also suggested that different training institutions have different effects on nurses' knowledge levels. Some institutions produce nurses who have minimal skills and knowledge concerning pain management. Participant B2 suggested that:

Some of them [nurses] have a really good knowledge. The rest of them, their knowledge is really low. Because they have graduated from different schools. Some of them have their bachelor's degree or their masters, but the majority only have diploma.

Some nurses have adequate knowledge regarding pain management. This is attributed to the notion that the majority of the nurses have attained a diploma, leaving a smaller population that has degrees and Master's-level training. Essentially, the perception of participants is that those who have Bachelor's or Master's Degrees comprise the segment with good knowledge on pain management.

In addition, many participants reported that the perceived level of knowledge of their nursing peers was also satisfactory, and depending on the unit, sometimes even better than satisfactory in their own specialities. Participant K alluded to diverse strategies of managing pain to show that most of the nurses were informed of the available pain management strategies: 'Almost all of us, as nurses, we know how to manage pain. I think my colleagues also they have idea about how to manage pain with medicines, with diverse therapies, and with psychological support'.

Some respondents were of the opinion that the level of knowledge of pain management among nurses was dependent on the unit where one was stationed. Evidently, some units were accustomed to dealing with cases of pain management more regularly than others; hence, they had more exposure in the field. This is highlighted by respondent H: So with regard to pain management, I've been working here in the burn unit for three years, with my sisters and the group. I think they have good knowledge about pain management because we are dealing with burn patients. We meet [each] case as different cases of burn patients, with different levels of pain. With the degree of their burns, we know how much they are suffering, especially if more extensive burns on the parts of the body. They really experience too much pain. So more of my sisters know how to manage this pain, in accordance with the doctor's orders, and also in accordance with their experience, while they are treating these burn patients

Participant H selectively stated that nurses in units such as the burn unit have to be highly knowledgeable on pain management, as their responsibilities in such a unit demands knowledge on how to aid patients experiencing different levels of pain. Participant H also pointed out that doctors' prescriptions and nurses' experiences are contributing factors to knowledge in pain management.

The same thinking was exhibited by participant J from the surgery unit, who insisted that knowledge concerning pain management was dependent on the unit the nurses were working in: 'It depends on which area, more specifically if you're in surgery they are more knowledgeable when it comes to pain. [Generally satisfied?] Satisfied'.

In contrast, some participants stated that nurses have poor knowledge of pain. This was exemplified by participant A2, who concisely stated that: 'Nurses have a poor knowledge in the area of pain assessment and pain management'.

Others, such as participant N, were of the opinion that their inadequacies in pain management were due to a lack of training. This was evident when the respondent stated that he or she did not have good knowledge of pain management: 'we are not yet well oriented about pain management, because before we don't have any idea about pain management. We are working as simple as basic' (participant N).

Participant N seems to indicate that no training on pain management was given to him or her, and that the little knowledge he or she has of pain management was gained while working in hospitals. Participant N places the blame on a lack of training prior to being attached to hospitals. In addition, some participants were of the opinion that nurses who were already operating in the hospital had adequate knowledge of pain management. However, this was not exhibited by new nurses who arrived at the hospital, implying that hospitals probably conduct in-service training for nurses. Experience was a determinant of knowledge in pain management as mentioned by participant D, who stated that:

Staff nurses' knowledge is good, but that newcomers do not have a full understanding of pain management, so they look to senior staff for support: 'All staff has good idea about how to manage the pain. But the newcomers, they don't have, once their patients complain of pain, they will ask us; sister, what will we do?

Other participants also shared the same point of view about newly appointed nurses. This kind of prejudice was evident when a participant stated: 'When it comes to pain management they [newcomers] are not really knowledgeable or well trained' (participant B). This respondent's perception of newcomers is negative and portrays an element of dissatisfaction in the level of his or her knowledge of pain management as mentioned by participant D.

125

5.3.1.2 Familiarity with, and Knowledge of Pain Medications Used.

The majority of participants cited being familiar with the medications typically used or ordered by the doctors at their facility. In the case of expatriate nurses, they found the medications to be mostly the same as those used in their respective home countries. However, some participants noted them to be different from the medications used in their home countries. For example, participant F stated: 'some are just the same also in our country but there are also some medications that are, yeah, different'. Participant C noted: 'Regarding the medications, the brand differs but the generic is just the same'.

Participants' responses on brand differences and the use of generic medicines in pain management is evidence that the nurses are familiar with some of the medications used in pain management.

Finally, familiarity with medications used to manage pain was discussed in detail by participant E, who noted a lack of use of the pain control and analgesics (PCA) pump. It was noted that there were differences in how different hospitals dealt with cases of pain management. This was exemplified by participant E:

It is same, but in previous hospital, I use PCA pump also, but I didn't use here PCA pump—pain control and analgesics pump. If patient has pain, they will press like that, then automatically the morphine will come. [So you have it in your country?] Yeah, [but] we don't have it here.

5.3.1.3 Interest in Learning about Pain Management.

Despite the perceived adequacy or satisfaction with their level of knowledge, the majority of participants, if asked, expressed an interest to learn about pain, pain management and pain medications. Indeed, four participants reported spending time in self-learning and reading about pain relief medications. The motivation to learn was prompted by the need to face exams and respond to doctors' enquiries on patients' progress, and for the purpose of career advancement. For example, participant K described nurses' desire to learn about pain management as being linked to the need to be self-fulfilled in their careers:

I am also a nurse with six years' experience but I tell you know, I have moderate experience and moderate knowledge about pain and pain management, so of course as a nurse, I have to improve my knowledge. So I would like to learn more.

The participant's motivation to learn is evident, demonstrating a positive attitude towards enhancing pain management skills among nurses through training. The aspect of self-learning was also advanced by other participants. However, their motivation was inclined towards developing the ability to respond to doctors' questions while monitoring and evaluating their patients' progress. This was best expressed by participant C, who noted both the desire to learn and self-learning through reading:

When I was in ICU, because every day, you know, our chief, the chief in ICU, during his rounds...he will ask all about this patient, then you are assigned to this patient—who is the nurse here? So, what are the medications this patient is receiving? So we should learn, what is that medicine for, like that. So I used to read, I had my drug handbook in my room...I am willing to learn. Every day there is more learning also, that's nice.

127

Participant C's motivation is geared towards self-actualisation in his or her career in nursing and maintaining self-esteem in the eyes of the unit's chief. It is evident that the ability to adequately respond to the chief's questions on the medications being administered to patients is essential while working in the ICU. This calls for self-learning and consequently improving the participant's knowledge of the medications used to manage pain.

In contrast, participant D showed no interest in learning more because there were no monthly knowledge tests:

In my country in my previous hospital they were conducting monthly test. They have to assess the improvement of nurses. Upon that they will increase the salary. So that time, we will read everything to pass the test. But here, we don't have any test and they are giving us the salary and the increment. So, after coming here, I did not worry more about it.

The participant's view is that nurses will consider self-learning only if tests are administered and linked to pay increments. Otherwise the participant seems content with his or her knowledge of pain management, at least while practicing at that particular hospital.

5.3.2 Current Pain Management Processes and Practices

The second thematic category revealed from the analysis relates to the pain management process as described by nurses from the time of complaint to resolution. This thematic category was formed from four subthemes: current pain management practices, alternative forms of management used (non-pharmaceutical), level of satisfaction with current pain management practices and processes, and nursing responsibility in pain management. Common responses indicate key themes related to the process of assessing, using alternative methods and checking the chart for PRN or referring to a doctor for pharmaceutical management (14 participants). Concern was expressed by some participants about the tendency of some nurses to neglect providing a quality assessment and/or depending only on pharmaceutical options, thereby neglecting alternative methods of nursing pain management. This was also evident in some responses that follow from the assessment directly to doctor or pharmaceutical options.

5.3.2.1 Alternative Forms of Management Used (Non-pharmaceutical).

Although the importance of pain assessment is recognised, some nurses directly mentioned going to doctor-prescribed medication. Pharmaceutical interventions were deemed more popular among nurses. For example: 'When the nurse assess properly then the nurse can immediately go to the doctor, then the doctor will decide what should be the right medication to give to relieve the pain the same way' (participant A). Participant A noted that nurses never consider the use of non-pharmaceutical interventions during addressing the patient's pain.

When a patient has a PRN medication ordered, the nurse is able to give that medication when needed, according to the orders. However, some participants expressed a belief that the reliance on medication is too quick and that some nurses do not effectively assess the pain or try alternative pain management before using the prescribed medications. It is possible that nurses fail to assess the pain properly because their approach is based on the premise of prescribing a painkiller rather than considering other available options. This was described indepth by one participant:

Personally, I have to tell you that they [some nurses] are not that competent in assessing pain. So if the patient would tell that he is experiencing pain, they will just have to look into the file, not to the patient. If the doctor has ordered pain relief, they

129
will just implement...Otherwise, you will just have to inform the doctor that the patient is experiencing pain. If there is no order, they would not even mind to act independent and do non-pharmacological technique. Even deep breathing, they would not do that, or positioning of the patient, that would, in effect, reduce the level of pain of the patient. Or assessment, also they fail to assess (participant S).

The respondent believed that some nurses are not efficient in assessing pain and, as such, they often rely on patients' files. The noted inexperience with procedures of pain assessment is attributed as the cause of nurses' lack of independence. Therefore, the option of administering pharmaceutical drugs is preferred over other intervention measures, as nurses do not understand the cause of the pain.

The alternative forms of pain management offered by participant H included psychological support and communication, attempting to divert the patient from the pain, deep breathing, positioning, using a heat compress, music therapy, and rest and relaxation.

We can do another method non-pharmacological. We can talk to them. We can divert their attention to something. While conversation, the pain is relieved. Or sometimes, with children, we can provide some music or playing activities to relieve their pain. So they will forget the pain with the use of music, with conversation or with some activities. So this can be used. Non-pharmacological.

Participant H identified communication as an alternative method of relieving pain, where distraction is used to divert the patient's attention. In the case of young children, music and playing activities are identified as alternatives to pharmaceutical drugs. The perception drawn is that participant H is well informed on alternative methods of managing pain and, as such, possesses a positive attitude towards these alternative methods.

Some nurses also drew attention to the possibility of some patients' pain being just a call for attention rather than a need for medication. In such cases, nurses should be able to identify the patients' needs and respond to them appropriately by offering to listen to their fears and thoughts rather than rush for pharmaceutical drugs. This was best explained by participant J: 'Sometimes they just call for pain [management] because they need somebody to be with them...Reassurance, communication, yeah, sometimes really pain is only for crying for help. Sometimes from the start they are afraid because they're in pain'.

Nurses should be able to distinguish when patients are in need of moral support to ease their pain rather than pharmaceutical drugs. However, patients who call for attention through pain are also annoying to some nurses, hence their negative attitude towards alternative methods of managing pain. A good example was provided by participant N: 'There are people who are— who will tell you that they are in pain, but actually they are just malingering. They are not in severe pain, they are just attracting attention'.

Another nurse agreed on the difficulty to assess pain when patients are perceived as 'non cooperative':

When the patient is cooperative and he tell honestly, so we can assess the patients' pain. But the barrier is when the patient is just malingering or making some drama, like that, so how could we implement good medicine if actually he is not in pain but he's just making drama (Participant C).

5.3.2.2 Level of Satisfaction with Current Pain Management Practices and Processes.

Eight participants were generally satisfied with the current pain management and seven participants described the misuse or non-use of assessment tools. The rest of participants did not wish to comment. Other problems that were noted included not properly documenting pain management, dependency on pharmaceutical management while failing to use alternative methods, and needing more communication with patients. For example, participant F linked his or her satisfaction to positive outcomes on the part of patients who have received services to manage pain: 'Yes, [I am satisfied] so far. Yeah, because most of the times, when we give interventions, the patients are relieved from their pain' (participant F).

Some participants were critical of how the process of assessing pain was carried out by nurses and subsequently how effective the procedure was:

Assessment of pain, there is a guideline in that form, what are the things they need to assess for pain, but they are not documenting properly, so that's a problem. So if you will look on the chart there's no documentation how long, where is the pain location. They will write the pain location but not the duration and aggravating factors like that, they will write the intervention but there is no reassessment after that intervention.

The participant was concerned about the proper documentation of pain cases where aspects such as pain location, duration of pain and aggravating factors of the pain were not duly documented, hence insufficient. Similarly, the participant also noted laxity in the reassessment of patients after the implementation of the intervention measures. Dissatisfaction was expressed about the current knowledge of pain management among nurses, especially with the documentation procedures, which are crucial because doctors refer to the patients' files. Concomitantly, participant B2 also noted other deficiencies in the practice, such as the lack of utilising pain management tools:

I will be honest with you, I'm not satisfied. Because they are not using the tools for pain management, like the numerical scale, which is 0–10, zero being nothing and 10 being the highest. I think the staff use their experience with the pain management, from the appearance of the patient, observation of the patient.

Participant B2 believed that dependence on the conventional methods of pain management was inappropriate in the context of current medical practice.

As noted previously, some nurses expressed that other methods of pain management—nonpharmacological methods—are not being used. Participant Q stated: 'Usually here the problem is when the patients complain of pain they don't give first the alternative ways. They would more provide pharmacological methods rather than other method'. This is another example where a participant expressed disappointment with nurses' overreliance on pharmacological interventions.

The health facilities provide a pain management form to aid nurses in the assessment of patients' pain. This was revealed by some participants, who described the use of the pain management form: 'Yeah, [we use the] pain management form. In that form, initially they would assess where the location, intensity, how long, then others also, what patient management they will implement independent and dependent management' (Participant B). Others did not use the pain management form.

Other assessment procedures were also mentioned by participant F, who discussed using facial expressions, pain scales and communication procedures:

Here first if a patient complains of pain we first assess the pain. You can use the facial expression or most of the time we use the pain scale that one to 10 then we ask the patient to rate his pain scale. Then based on that we first try to divert their attention or anything like that. We inform first the doctor or the medical resident on duty.

Participant D provided an insight into the assessment process and the problems with nurses assessing the level of pain rather than patients assessing their own pain:

Pain should be subjective and assessed by the patient...but here, we are the one assessing the pain and we are asking the patient; how is the pain? Yes I have pain here, here, here? So then we will think, maybe four to six is his pain...we will decide, not the patient.

Participant D was of the opinion that nurses should listen to the patient's assessment of the pain he or she is experiencing in order to be in a position to adequately respond to it. This is contrary to the current practice, where nurses usually assess the pain on behalf of the patient.

5.3.2.3 Nursing Responsibility in Pain Management.

Nurses were perceived to play a critical role in the pain management process because they have the greatest extent of contact with the patient. The shortage of doctors aggravates the problem, giving the nurses a more important role in caring for patients over 24 hours.

Nurses have really a big responsibility in pain management because they are the first contact with the patient and for eight hours they are with the patient, unlike the doctors only a few minutes...they are the eyes, the ears of whatever their patients are feeling or telling to them. So they have the big responsibility (participant C).

Nurses also discussed their responsibility to communicate with doctors in discussing a patient's level of pain and pain relief medications, and offering their knowledge of the patient's condition. For example: 'Most of the time, we can discuss with the doctors. For example, if they want to give this medicine, we can tell the doctor—doctor, maybe we should try first to give this medicine' (participant F).

Participant S described the ethical necessity of speaking up if the nurse feels there is an error, but also noted that many nurses do not discuss errors with doctors or pharmacists:

No, they [nurses] didn't [discuss with physicians]—just carried out what the doctor has ordered. We have the right to question the doctor. Then if this does harm to the patient, why don't we ask the doctor that the order is not good, it's not valid? It may do harm to the patient...so anything that will do harm to the patient, we must protect. We must protect our patient (participant S).

The participant believed that nurses should be bold enough to voice any concerns they have about doctors' prescriptions. Ideally, as nurses know their patients better, they are in a better position to caution doctors on any prescriptions that might be harmful. The participant's interest here was purely to ensure professional efficiency rather than fuel professional rivalry. Participant S highlighted the role of the nurse as an advocate for the benefit of the patient.

5.3.3 Nursing Pain Assessment

The use of either or both number and face pain scales were reported by most participants, while only a few reported not using the tools for assessment. Other assessment techniques included communication with patients and vital signs/physical examinations.

5.3.3.1 Nurses' Pain Assessment in Terms of the Tools Used.

The health facility provides guidelines on how nurses should proceed with the assessment of pain for both conscious and unconscious patients:

In the assessment of pain, we are given this guide paper and we are using the Wong– Baker and the numeric scale. So for the conscious patients we're using the numeric scale most of the time, but in between for these unconscious, it depends, sometimes we are using that Wong–Baker because there is a feature, of the faces scale, it's the same scale with a picture.

Participant C identified two tools used in his or her facility and explained the difference between the Wong–Baker face scale, which is used for unconscious patients, and the numeric scale, which is used for conscious patients. However, participant B noted that some nurses are still not using the scale, or they are not familiar with using it or other techniques to assess patients' pain levels. The point of concern here is to identify the reason/s why some nurses do not utilise the scale. It will be valuable to find out whether it is due to ignorance or a lack of knowledge on how to use the scale to deliver services to patients. Participant B noted that the scale is effective, but is not often complied with: Actually for me, it's just the compliance of the staff nurses to use that, because they are not used in telling the patient how would you rate your pain. It's not a practice here in Saudi Arabia...but we are encouraging the staff nurses to use it.

The respondent was supportive of using the scale as a pain assessment tool and was of the opinion that the reason for not using it in Saudi Arabia is that nurses are not inclined to ask patients to rate their pain. However, the respondent strongly agreed with the view that the tool should be encouraged and utilised more.

Participant O stated that there is no form for pain assessment. This is in contrast to the health facility's claims to have custom-based procedures for assessing pain in patients. However, in reality, the pain assessment tools in the hospital guidelines are not made available to the nurses as they should be. Participant Q stated:

We don't have written tools to assess the pain of the patient. I just read in the book, but it's not being implemented because we don't have that assessment form wherein we can initially assess the patient. We don't have that assessment form.

The implication of such an observation is that deficiencies in pain management cannot be wholly directed to nurses; they should also be directed to the administrators of the health facility, who do not meet the required standards of pain assessment. The pain assessment form is one such tool that aids in assessing patients' pain, recording and provides a follow-up procedure to address the patients' severity of pain.

Other participants confirmed that the majority of nurses are not familiar with pain scales. Essentially, although the pain assessment tools are, in some instances, made available to nurses, the challenge is the lack of knowledge and skills among the nurses to utilise the tools. This factor was also highlighted by participants: 'I see most of them don't have that knowledge of how to scale the pain and get it accurately' (participant C2).

The respondent C2 refers to the lack of skills among some nurses in the utilisation of pain assessment tools such as the scale. Such deficiencies could result in inaccurate and insufficient assessments, which are detrimental to diagnosis and treatment. Adequate knowledge and skills in pain assessment is a prerequisite for adequate pain treatment. Hence, there should be greater emphasis on the need for nurses to be accurate and efficient in using pain assessment tools. The deficit in knowledge of pain assessment to the absence of the relevant content in nurse training curricula.

5.3.3.2 Training for Pain Assessment.

Some nurses have received no training in using the tools, so they had to teach themselves or gain on-the-job training. The knowledge and skills deficit is recorded in many studies and was also revealed in this study, as seen by the following response of participant H: 'We don't have actually formal training but with our experience in handling these patients, we are able to do and practice our tools'.

Participant B described doing his or her own research in learning to use the tools: 'I just read and then research, but not that intensive training for those tools'. Participant B exuded motivation and a desire to know and understand the effective use of pain assessment tools.

Another difficulty in using the tool is the language barrier, which limits effective communication with patients. Multicultural settings imply diversity in languages and therefore the necessity of a strategy to adequately communicate with patients from various backgrounds, especially in the context of a health facility. The problem was exemplified by participant D, who had a clear understanding of the use of the instrument, but also had difficulties when trying to explain its use to patients using different languages:

We have Wong–Baker face scale and numerical pain scale...So after an operation, the patient are coming to ICU. So once the patient will awake, they were complaining of pain...So we will ask the patient what is your pain level? So the patient will tell what is the score...But here, the main problem is language problem—Sudanese, Pakistani...different. Because in Saudi Arabia, we cannot explain in Arabic...we don't know how to explain in Arabic.

The participant suggested that the lack of fluency in Arabic is a challenge to effective pain assessment, and this can be a cause of frustration due to nurses' inability to respond to patients in their native language. It is clear that nurses working in diverse settings face considerable challenges in assessing the pain of patients from different cultural backgrounds. Thus, professional nurses should be in a position to utilise credible and reliable cultural assessment procedures to enable the negotiation of a culturally congruent strategy for culturally diverse patients.

5.3.3.3 Perceived Efficacy of Pain Assessment Tools.

Pain assessment tools are undoubtedly essential in pain management, and their efficiency was highlighted by the majority of participants. Participant Z stated: 'I think the tools are very effective and important because it's a standard part to evaluate the pain'.

This participant had a positive attitude towards pain assessment tools, believing that utilising these tools is a key component to the standard evaluation of pain in patients. In contrast, some

respondents were critical of the sole dependence on pain assessment tools, arguing that their use needed to be combined with other data about the patient, such as background information and the social–cultural context of the patient:

Yes, it is effective but it's not only checking the face of the picture and the patient's face, because you need to have that background of the pain and how it started and what's affecting the pain. The location of the pain will affect the facial expression or anything else. Also, we need to know—that some patients cannot express their pain...because of their religious beliefs or cultural beliefs (participant C2).

Participant C2 portrayed a positive attitude towards the utilisation and efficiency of pain assessment tools, but was also quick to note that the pain assessment procedure should not rely only on these tools; other factors, such as patients' background information and religion, are also essential in aiding nurses to actively and efficiently manage patients' pain.

Pain assessment tools are even more essential when assessing children, as children are not in a position to verbally explain the location, duration and level of pain. This was best explained by participant G: 'In paediatrics, I think it's applicable because you cannot ask the baby how much is the pain? How far the pain goes? But you can assess through this Wong–Baker, I think, as far as I can understand'.

This respondent acknowledged that tools such as Wong-Baker are essential when assessing pain in children and therefore showed a positive attitude towards the tool. The assessment criteria of children should also be multidisciplinary in nature to ensure the best diagnosis in children experiencing pain.

5.3.4 Barriers and Limitations to Optimal Pain Management

The key elements within this thematic category include language and communication barriers, workload due to staff shortages, cultural and religious factors, and a lack of education and nurses' updated knowledge regarding pain management. Cultural, religious, gender and age (elderly and paediatric patients) factors were common occurrences among participants' responses, with greater in-depth perceptions of these elements.

5.4.4.1 Language and Communication Barriers.

The language barrier was the most commonly reported barrier to providing effective pain management. This was particularly true for foreign nurses; however, at times, it was even true for Saudi nurses, given the existing variations in accents and dialects of patients. Nurses who cannot directly communicate with patients have difficulty identifying the root cause of a problem and, as such, intervention may not occur in a timely fashion when compared to a situation where both the patient and nurse are able to communicate effectively. In cases of different Arabic dialects, the nurse may only obtain partial information about the patient's problem, hence limiting the effectiveness of pain management:

Honestly if they are trying to express their pain to nurses which don't have that much knowledge regarding Arabic language, that is really a barrier because if a nurse cannot easily understand the patient regarding their pain, cannot even understand what is pain all about (participant A).

Participant A recognised the language barrier as an inhibitory factor in effective pain management. Consequently, patients tend to suffer while trying to communicate their pain.

Participant A believed that nurses are responsible for learning the languages of the patients that they deal with in order to effectively deliver good pain management.

Participant F stated: 'For me, my one big barrier is I don't know much of Arabic, so most of the time I ask my seniors to help me to translate'. The participant recognised his or her lack of ability to speak Arabic as a major challenge in fulfilling his or her duties as a nurse, and the need to constantly seek help from seniors who have a better grasp of the language. The genesis of the problem is the inability of nurses to obtain first-hand information from patients on how they are feeling, as well as the severity of the pain and its duration. This was best expounded by participant G, who noted difficulties with communication: 'The most important in the factors affecting pain really for me is communication, at least more of communication, how to deal with the patient, especially if you're going to ask, if we don't understand each other'.

The participant viewed the communication barrier as a significant factor because it generally affects the process of either being or not being in a position to help the patient. This was also discussed by participant H, who described the problem as not limited to the Arabic language alone, but also to accents within the language. The Arabic language is further compounded by the complexities of having varied accents, sometimes with conflicting meanings, hence leading to misinformation. This is challenging to nurses because misinterpreting a patient who speaks with an unfamiliar accent may lead to the wrong diagnosis, resulting in ineffective pain management:

Sometimes we cannot understand how much pain they are suffering because we cannot actually get what they want. For example, they are complaining of pain and we cannot understand. Sometimes they have a different accent. Those people from the

village they have [different accent]...we cannot understand what they want (participant H).

Participant C2 mentioned the language barrier and the related cultural barriers of working in a different country and the multinational nature of the area:

I think we have true multi-cultural hospitals here. We have so many nurses from different countries and I think that culture is included, because the international nurses cannot understand the local patients here. So maybe culture will be one of the other factors for that—especially for the international nurses. Yeah. Actually, I see most of them don't have that knowledge of how to scale the pain accurately. Because again and again they come from a different culture and also the language is a big barrier for them. Especially when they are newly employed here in Saudi Arabia. They cannot understand the patient in the correct way. So, they cannot manage their pain because they cannot understand what they are asking for (participant C2).

Participant C2 raised the concern of foreign nurses' ability to provide effective pain management to patients in the facility. The concern is based on a lack of coherence between the nurses' and patients' cultures (Leininger, 2002). This participant was also pessimistic regarding foreign nurses' levels of knowledge on pain management, mainly due to the perception that they do not know how to use the pain scale effectively. Further, language barriers are also deemed a challenge to them because they cannot comprehend how patients express pain. Participant A2 demonstrated the connection between language and culture and the effect on communication: 'Communication—especially with international [nurses]—they don't know the culture of Saudi here. The language also. Yeah, the language, the culture, how people express their pain. Yeah, how to ask them, when to ask them'.

The participant expressed dissatisfaction with foreign nurses' capability of assessing Saudi patients, especially due to their lack of comprehension of the language and the culture of patients. It is also important to note that some cultures believe it is wrong to speak to outsiders about pain; rather, they tend to assume that healthcare practitioners will ask the relevant questions, and questioning the judgment of professionals is rude.

5.4.4.2 Workload Due to Nursing Staff Shortages.

A shortage of nursing staff was also perceived as a barrier to providing efficient pain management. Nurses' workloads are overwhelming when the number of patients requiring services on a daily basis exceeds the number of nurses available to provide these services at a given time. This was revealed by Participant B:

Since we have a shortage of staff, if patient will complain of pain every 30 minutes nurses cannot attend those patients. It's not reasonable but if you will see the actual situation it's really difficult to manage, for example, 30 patients with four nurses in the ward.

Participant B considered that nurses are not able to cater for more demanding patients due to the sub-minimal nurse-to-patient ratio. The participant drew attention to the fact that the shortage is alarmingly high when four nurses are in charge of 30 patients. One of the strategies adopted by Saudi Arabia is to attract international nurses by offering better pay and working conditions, and by prohibiting the migration of Saudi nurses to other countries.

Participant D asserted that the situation not only leads to suboptimal pain management, but also patient neglect:

Twelve patient means at the time there is 12 demands and that the nurse is only one. How can she manage the 12? But, we are adjusting our time and we are just managing it, but of course some patient will get neglected.

This respondent expressed frustration about the shortage of nurses, which compromises the quality of care given to patients and affects the efficiency and effectiveness of nursing care.

Shortage of nurses will increase workload of available nurses. This was best explained by participant C:

All I can say is that it's not so satisfactory [pain management], it's not enough, and maybe because of the workload in the setting, because the patient ratio is not ideal, patient–nurse ratio is one is to 10, or to 15, like that, so really it's difficult.

The workload was also tied, by some, to an inability to achieve further education due to a lack of time and overwork, even if opportunities are available:

Also, the shortage and the lack of nurses. That will increase the workload on them. Yeah, that will increase or decrease the time that they spend in education. To summarise, education in general [is a barrier]. Also individually, nurses need to educate themselves individually and unfortunately, they are not. We can't blame them because of the workload (participant A2).

Increased workload means that the available nurses do not have time to advance their knowledge through further education. Essentially, participant A2 argued that the shortage leads to nurses not advancing in their education, as they have to give priority to delivering services to patients. Thus, a shortage of nurses has other ramifications that affect both patients' satisfaction and nurses' career advancement. The argument is logical, as pain

management is affected by both the level of education of nurses and the nurse-to-patient ratio. The dissatisfaction among nurses due to limitations to career advancement and higher nurseto-patient ratios are the key contributors to increased workloads, exhaustion and nurse burnout.

5.4.4.3 Cultural, Religious and Other Factors

Cultural and religious factors were considered important barriers to achieving optimal pain management by many participants. When operating in multicultural settings, the likelihood of having different religious groupings is very high. As such, nurses have to be aware of the beliefs and practices of various religious groupings and their perceptions of pain in order to adequately deliver pain management. 'Beliefs and religion as a Muslim—if someone is really in pain and he tolerates the pain, he will receive a reward from Allah. That's what we believe and that is why some people tolerate their pain' (participant B2).

The participant explained how Muslim religious beliefs affect pain management because of the view that tolerance of pain is rewarded with blessings from the creator Allah.

Religious and spiritual perceptions of the caregiver also play a role in the delivery of pain management services.

But spiritual aspect, I would say that it matters also because once you have faith in God whatever your religion is, so the pain will be easy, you can carry the pain even though it's very severe if your faith in God is strong. So it's just okay, you can deal with it, just like me in my own experience, so at times that I was in pain, this labour when I deliver, so I just pray (participant C). Participant L believed that religion could be an obstacle to effective pain management because some religious beliefs and practices do not allow the use of certain procedures and drugs of modern medicine.

But it [religion] really affects the pain management. So it limits the management you can do for that patient, because there is some culture you also have to follow. You will not ignore this. Because they have their beliefs. You cannot fight with their beliefs (participant L).

Nurses are at a crossroad in such situations, as they are faced with the dilemma of respecting patients' beliefs while also offering the best medical advice despite the patients' religious bias. Participant L also believed that nurses should not ultimately cross boundaries in trying to convince patients to agree to the prescribed pain management strategy if it contrasts with their religious beliefs.

Cultural aspects also play a role, particularly in the expression of pain, and particularly among men. Pain management interventions adopted by nurses should also consider gender perceptions of pain as moulded by the cultural upbringing of the patient. Different cultures have different perspectives on how men and women should respond to painful experiences:

I really agree that culture affects us. In our culture men have to tolerate their pain sometimes and they will describe the man who didn't tolerate the pain negatively, it's like a shame on him to complain of pain, unless that pain is untreatable (participant B2).

This participant was convinced that culture directly affects how patients perceive pain. The example given was related to their culture, where men are brought up to endure pain silently because expressing it publicly is deemed shameful:

Of course, this is very important because we, as a Saudi nurse; we understand the culture, our culture, so we don't expect the patient will complain. We have to ask the patient frequently about their pain. But for the international staff they might be waiting for the patient to establish their pain. That's why I always tell my staff—the new international staff about the Saudi culture. They must ask the patient, it's like shyness to say I am in pain, because you think you are strong, you are a man. It's not the right way to complain...they tolerate, even the severe pain, unless you try to figure out (participant Z).

Participant Z expressed the opinion that Saudi nurses are better in catering for patients in the facility because they understand the Saudi societal construction of gender and how the different genders should respond to pain. The participant believed that international nurses should be taught the Saudi culture in order to adequately deliver good pain management. Among the cultural aspects they should be taught is how to educate patients about their pain rather than expecting an automatic description of the condition by the patients themselves. In the case of Saudi men, nurses must be aware that men are socialised to endure pain in silence. Hence, it is the nurse's duty to probe the cause, severity and duration of pain.

The results also indicated differences in pain tolerance and effective pain management related to demographic characteristics such as gender, age and nationality (Wandner, L, Scipio, C., Hirsh, A., Torres, C., & Robinson, M., 2012). Some gender issues were noted in the cultural differences, and some gender differences are related to cultural effects. Thus, there is a degree of overlap with these two subthemes. These demographic differences are not limited to gender, as previously described. There are also differences between nationalities and ages. In general, children were described as not able to tolerate pain and were therefore medicated

quickly to alleviate their pain. Pain management in children is complex because children are not in a position to aid in effective pain assessment strategies. For example: 'In children we cannot [offer alternative pain management methods]—it's a different factor affecting children, so I think it is more of—for children it is more of giving right away the right medication' (participant G).

This participant believed that when caring for children, nurses cannot be dependent on alternative measures because children cannot vocalise their pain. The participant was of the opinion that children's pain should be dealt with as quickly as possible, as the effect of pain is greater on children than on adults. Nurses should utilise pain assessment tools that have face ratings when dealing with children as young as three years. It is also important to appreciate that children are less likely to react to pain emotively because they cannot verbalise it to gauge the level of the pain.

Gender differences were also noted in pain tolerance. As previously described, males' ability to endure pain is generally tied to cultural reasons, whereas female tolerance was often described in relation to childbirth or other types of general pain tolerance. Saudi females were also described by some nurses as 'over-expressing' pain, with nurses noting that females were generally more expressive than males in terms of pain, which may be related more to cultural differences than gender differences. The gender differentiation of pain is due to the societal construction of how males and females should perceive pain according to the cultural expectations of their ancestors. Nurses' knowledge of such facets is essential in the diagnosis and treatment of pain when caring for Saudi patients, as part of their responsibility is to respect the culture of their patients. For example: 'The Saudi—I believe—the Saudi females are—over-expressive in their pain, and their pain threshold is very low. Just—sore throat or what do you call this—tooth ache—they will ask for more pain reliever' (participant O).

This participant believed that Saudi women are more expressive of their pain than men. In essence, the participant believed that dealing with Saudi women calls for patience and understanding, as their behaviour is directly linked to their cultural upbringing. Some nurses described perceived differences according to nationality: 'In my experience, regarding the pain it's also different with nationality. I know when it comes to people here from Saudi Arabia, they're not much, but with regards to other patient—Egyptian—they cannot tolerate it' (participant Q).

This participant believed that Saudi people are more tolerant of pain compared to other nationalities, such as Egyptians. The respondent said that cultural considerations should be taken into account when dealing with pain expression.

Others asserted that pain tolerance and management depends on the person rather than gender, nationality or age. Participant D expressed this notion, but also noted that the education level of the person with regard to the ailment can make a difference, as someone with no knowledge about the ailment is likely to be more afraid. Individuality is also an essential consideration in pain management. In every society, stereotyping often results in conflicts with individuals who do not conform to society's cultural beliefs and values.

Actually, the thing is that I am doing a male surgical ward so I have only experience with the men. In men also, I don't have any experience with women and children. So in my experience, in between men also, there is pain threshold level is increasing and decreasing. Some male patients, they are tolerating very well...and some patients for a small pinch...Pain is painful for all human beings...If some professional male or female, if they know the situation, they may try to tolerate. It depends upon the patient and his knowledge, his family level, his status, his education (participant D).

This participant disagreed with the notion that all men in Saudi Arabia are capable of tolerating pain equally, arguing that each individual is different. The participant believed that nurses should not rely on stereotyping in pain management; they should instead focus on each patient as an individual, as differences in perceptions of pain are significant in pain management. The respondent thought that an individual's perception of pain is also influenced by their level of knowledge, their status in their family and their professional qualifications.

5.3.5 Factors to Improve Pain Management

The factors that the participants perceived to improve pain management were discussed, along with the need for professional development to provide continuous learning (education) and updating of nurses' knowledge. This education includes coursework, cultural training or orientation, self-learning and professional development opportunities such as seminars and lectures. The intervention measures for the correction of deficiencies in pain management were mostly hinged in education. However, the time for nurses to advance in their education is limited due to an overload of work. The solution put forward was for hospitals to assume responsibility by providing in-service training for nurses at their workplaces.

First thing is education. When the education department in any hospital is active and they will go into the hospital and find what area that nurses need. They know that the nurses have a poor knowledge in the area of pain assessment and pain management. They have the ability to offer those courses to us as nurses, but unfortunately they are not. Yeah, education, that will help us, which is not there.

Education was identified as the key solution. Participant A2 suggested that health facilities should take up the initiative once it is identified that their nurses are not efficient in pain management. The participant portrayed a willingness to learn and be active in such programmes, but was disappointed that health facilities are reluctant to address their staff needs to upgrade their knowledge.

One nurse recalled that during their college years, pain management was sparsely covered, hence the need to advance in the topic despite the experience they have gained:

I remember now my college days; this pain management is not a vast topic. We are getting little knowledge already, but with experience we will get more knowledge. But I think any classes or lectures about pain management and assessment of pain—that will improve all staff knowledge about pain (participant K).

Another nurse felt that inadequacies in the delivery of pain management could be solved through nurses advancing their level of education: 'The university...must upgrade their degree to bachelor. The nursing office must provide courses about pain management, continuing nursing education. All those factors would help' (participant Z).

Participant Z proposed that universities must upgrade the qualification they offer from a diploma in nursing to a Bachelor's degree. The respondent also highlighted the role of pain management courses, which help to improve nurses' knowledge regarding pain and its management. Continuing education among practicing nurses was also identified as a solution to poor pain management in the health facility. Continuing education is essential because it not only keeps the nurses up-to-date with new methodologies of pain assessment, but it also

provides a forum where nurses can share their experiences with the subject and how some situations can be improved.

This is also good thing. This continuing education. I think—I don't know if they give more important about pain management, but in my previous hospital this is one of the continuous education that we're taking—this pain management (participant Q).

This respondent had confidence in continuing education as an effective solution to the associated knowledge deficiencies that may hinder effective pain management. In addition of cultural competence courses, coursework was suggested on pain management. This was particularly true for non-Saudi or foreign nursing staff, who need training during orientation and upon entry to the workforce to enhance their awareness of Saudi culture. The concern posed by foreign nurses regarding cultural education may be attributed to the frustration and challenges they deal with on a daily basis because they cannot effectively communicate with patients:

It really helps if you go deep especially how to deal with these individuals you don't really know, you don't understand what they're—his beliefs, his cultures, his knowledge...Because me in my country, I know my culture so I know how to deal with the patient. I know how to deal with the patients but here, [but] it's a new culture, so very different. You don't know how to give the right words, you don't know how to say this, or he might be offended or something. You might say something to him that is not really applicable with his beliefs so I think they must give that. In my own opinion, I'm not really well-versed about these cultures of the Saudis, so when I come here, I really have these difficulties, really. Because of—how will you say this properly—because they have different perspective, different religion, different from my culture. It's a very big difference. So when I come here I really have those problems with me, how to adjust. So it's very hard.

Participant G was frustrated at not being able to deliver services to patients in Saudi Arabia due to an inability to communicate with the patients and a lack of understanding of their religion, level of education and cultural upbringing. Participant G perceived these challenges as limiting his or her responsibility and roles as a nurse, hence suggesting mandatory cultural induction and education for foreign nurses.

This was noted by participant M, who discussed the multicultural nature of the area and stated that cultural diversity training or orientation would be helpful:

They should have a preview of what they are going to have during their stay in the country. Because we already live in a multi-cultural community. It's not just like, oh we are in our country, so we have only people from my nationality or we are in Saudi Arabia oh we have Saudis, no. We are all, [our] nationalities are all mixed.

5.4 Conclusion

Participants' responses were based on their level of education, individual experience and practice at their current health facility and in other workplaces. Concerning their knowledge and familiarity with pain management and medications, there was a general consensus that they were satisfied in these aspects. However, it was also evident that they appreciated the value of further training to improve their efficiency and effectiveness regarding pain management in their health facility. Furthermore, views were expressed that the level of knowledge of pain management was directly linked to the level of education attained by nurses. The implication is that the higher the level of education of a nurse, the higher the level of his or her efficiency and effectiveness in the field of pain assessment and management. In addition, some of the respondents believed that the level of knowledge of pain management

was also dependent on the institution of learning that one attended, as some nursing schools were deemed to be more prestigious and to produce better skilled nurses in the field. The unit or department assigned to nurses was also considered a contributing factor to the level of knowledge and satisfaction regarding pain management among nurses. Conversely, those who expressed dissatisfaction with their own and other nurses' knowledge concerning pain management, attributed poor knowledge to poor training and deficiencies in their healthcare facilities.

On the issue of identifying familiarity with knowledge of pain management medicines, the respondents recognised differences and similarities in the types of drugs used in different countries. Respondents also noted a shortage of drugs, such as PCA pumps, in the facility, hence limiting effective pain management.

The desire and willingness to learn was noted by most respondents, apart from a few who stated that they would only consider further learning in the field if it would lead to increased wages. The strategies proposed to encourage further training included self-training and continuing education programs for practicing nurses. The desire to learn was linked to motivation factors such as career fulfilment and self-actualisation.

The respondents were concerned about the neglect of non-pharmaceutical measures and the subsequent overdependence on pharmaceutical drugs to deal with pain. Some respondents blamed this overdependence on nurses' inadequate knowledge and skills regarding pain assessment. Non-pharmaceutical interventions that were identified included music therapy, psychological support, deep breathing, rest and relaxation. It was also noted that a challenge faced by nurses during pain assessment was patients lying about their condition as a way of seeking attention and recognition. Moreover, most respondents held the opinion that the

current practices and processes in pain management were satisfactory. However, some problems were also noted, such as insufficient documentation, lack of utilisation of pain assessment tools, over dependency on pharmaceutical drugs and a lack of subjectivity while assessing patients. Respondents agreed on the responsibility of nurses regarding pain management. This was further supported by the notion that nurses act as a link between patients and doctors in relation to diagnosis and treatment procedures.

The identified assessment tools used by nurses include the numeric scale and facial expressions. Wong–Baker was considered essential in the pain management of children and unconscious patients. Emphasis was laid on using pain assessment tools, as they help to increase quality care in pain management. However, some nurses had no knowledge or skills in using such pain assessment tools, thereby compromising on quality care. Nurses believed in integrating pain assessment tools with information such as patient background, religion, culture and their medical file in order to provide effective pain assessments and management.

One barrier that was identified was the language barrier, which frustrates both patients and nurses due to an inability to understand each other. In Addition, another barrier was cultural, where foreign nurses experienced cultural gaps in pain management. The shortage of nurses in Saudi Arabia was classified as a barrier to optimal pain management because it leads to increased workloads for available nurses. This becomes more complex, as the patient–nurse ratio becomes too high, leading to the neglect of some patients. The shortage may also lead to nurse burnout and a lack of time for nurses to advance in their level of education. Other barriers highlighted in regard to effective pain management were cultural and religious barriers, which may limit nurses in implementing pain interventions due to patients' beliefs.

The last theme identified some of the interventions proposed in dealing with poor pain management, including:

- Health facilities to offer in-service pain management training to their nurses
- Collaboration between practicing nurses and nursing institutions to devise an adequate pain management curriculum
- Foreign nurses to be offered an induction course to orient them with the Saudi culture
- Training nurses to deal with diverse cultures in the context of pain management.

Chapter 6: Discussion Chapter

6.1 Introduction

This chapter discusses the results of the data analysis in the context of the existing literature. The chapter discusses the findings in relation to the study's research questions (see Chapter 1), which are answered by both the quantitative and qualitative components of the data collection. Pain management is a complicated topic because it involves, and is affected by, several factors, including patients, healthcare providers and the systems or policies of relevant organisations. However, the principal focus areas of this study are nurses' knowledge and attitudes regarding pain, the role of nurses in pain management and the barriers to providing effective pain relief in the Saudi Arabian context. Attention has been drawn to the effect of patients' cultural backgrounds and that of the healthcare providers in pain assessment and management, including Islamic societies such as Saudi Arabia where Islam is the dominant religion and might not be shared among all health workers. Nonetheless, the main finding of this study concerns the deficiencies of nurses' knowledge on matters relating to efficient pain management.

6.2 Discussion

This discussion includes the results and findings of the survey utilised in the quantitative phase, as well as those from the interviews in the qualitative phase. The findings were analysed and interpreted in relation to the existing literature, and the facts related to their clinical significance were included with some expected outcomes concerning pain management. Each research question is addressed separately in the following sections, and the results are discussed in the context of the relevant literature.

6.2.1 Nurses' Knowledge and Attitudes Regarding Pain Management

The KASRP items were designed to measure participants' knowledge and assess their attitudes regarding pain management. In this study, based on the correct answers provided by each participant for the 40 questions, the overall scores ranged from a minimum of 5.0 % to a maximum of 87.5 %, with a mean score across the 303 participants of 41.76 % (SD = 9.83). The majority of the participants scored 35–47.5 %. There were a few outliers, for example, seven participants scored ≤ 20 % and 11 scored ≥ 60 %. Only two participants obtained a passing score of ≥ 80 %. The average correct response rate in this study (41.76 %) was very low and deviated significantly from the acceptable score of 80 % (McCaffery & Robinson, 2002; Brown, Bowman& Eason, 1999). The low pass rate (two out of 303, or 0.6 %) indicated that nurses in Hail region hospitals have deficits in their knowledge and attitudes regarding pain. Although such an inference is made based on the stipulated standard that a score below 80 % is considered a lack of competence to satisfy patients' needs in pain management, the average percentile result in the present study lies within the wide range observed in other similar studies (Bernardi, Catania & Tridello, 2007;Lewthwaite et al., 2011; Lui, So & Fong, 2008; Matthews & Malcolm, 2007; Plaisance & Logan, 2006; Rahimi-Madiseh, Tavakol& Dennick, 2010; Wang & Tsai, 2010; Yildirim et al., 2008;). For example, Matthews and Malcolm (2007) reported an average correct response rate of 73.8 % for their participants. However, a similar study conducted by Yildirim et al. (2008) to measure the knowledge and attitudes among Turkish nurses using the KASRP tool showed an average correct response rate of 35.4 %, which was lower than that of the current study.

6.2.1.1 Nurses' Pain Assessments

Many studies (Ballantyne, 2006; McCaffery & Ferrell, 1997; Arbour & Gelinas, 2010) provide evidence to suggest that patients' self-reports are the most reliable indicator of pain intensity and are thus considered the reliable standard for pain assessment. While the majority (59.7 %) in the present study agreed with this, a number of nurses (40.3 %) did not always follow this simple rule. Kaki (2009) found that nurses were more likely to accept patients' reports of pain. They take cues from grimacing patients and may disregard smiling patients. In Kaki's (2009) survey, 300 questionnaires were distributed in various healthcare settings at King Khalid National Guard Hospital in Jeddah, Saudi Arabia. Kaki's study found that nearly 23 % of nurses ignored the self-reports of patients for different reasons and failed to see a difference between eight out of 10 and zero out of 10 on the numerical scale of pain assessment. In this case, nurses were ignoring patients' self-reports, which are crucial in assessing patients' pain and consequently could negatively affect its management.

One segment of the KASRP survey is related to two case studies regarding pain assessment. The relevant questions aim to identify the attitudes of nurses regarding pain, and they require participants to determine whether a patient is in pain judging from their smiling or grimacing facial expressions. The percentages of correct answers obtained with regard to smiling or grimacing patients were 9 % and 32 % respectively. The percentage of participants who accurately rated the pain scores of both was lower than that of other research studies conducted internationally. These results show a major deficiency in nurses' knowledge, as well as inappropriateness in their attitudes regarding pain and its management. This is an adverse outcome when compared with the results of Matthews and Malcolm's (2007) study, in which 51.3 % and 77 % of nurse attendants were able to correctly judge the pain of smiling and grimacing patients.

Al-Moriarty (2011), Mathews and Malcolm (2007) and Wilson (2007) advised that pain assessment measures should be used by nurses on a regular basis to select the best intervention measures. Further, other factors such as cultural beliefs and past patient experiences should be taken into account (Finley et al., 2009). Some participants in the interviews showed confidence in the ability to assess pain, and subsequently the procedures that would help him or her to adequately address the problem. However, participants relied on the judgement of doctors in pain relief, showing a lack of confidence in their own ability to help the patient.

6.2.1.2 Nurses' Attitudes

A substantial proportion (32.7 %) of participants in the current study incorrectly believed that changes in vital signs are an accurate indication of the existence of pain. This is a misconception with regard to the pain assessment process, but it is not limited to the present sample of nurses. A study by Coulling (2005) also found that 32 % of participants believed that vital signs represented a primary indicator of the intensity of pain. Other misconceptions that could be linked with nurses' beliefs regarding physiological changes in vital signs have been observed by other researchers (Huth, Gregg, & Lin, 2010; Yildirim et al., 2008; Bernardi, Catania, & Tridello, 2007).

Similar misconceptions were prevalent among nurses in relation to patients' pain-related behaviours, coupled with associated knowledge deficits. The neuromatrix theory of pain suggests that pain is significantly influenced by psychological factors, which are vital components in the perception of pain (Mandeville, 2010). Consequently, the role of nonpharmacological therapies, such as information provision, distraction, relaxation techniques and cognitive behavioural interventions, have gained value in managing pain (Macintyre et al., 2010). Helmrich et al. (2001) conducted an Australian study to ascertain the attitudes of nurses and the application of non-pharmacological methods for pain management. They found that (89.3 %) of the nurses (N = 37) claimed that they used non-pharmacological methods to help in the management of patients' pain. In contrast, several other studies ascertained that the use of such integrated (pharmacological and non-pharmacological) treatments were negligible to non-existent (Eid & Bucknall, 2008; Manias, 2003).

It is well known that many aspects of normal activities are altered in patients suffering from pain. Studies have provided evidence to support the notion that nurses should assess pain depending on non-verbal cues and behavioural manifestations and they may include physiological changes in vital signs. According to some participants, patients in pain are unable to sleep. Nonetheless, participants demonstrated knowledge deficits in this area too, as indicated by a substantial proportion (35.6%) falsely believing that patients who can be distracted easily from pain usually do not have pain of any considerable severity. The belief among nurses that patients whose sleeping habits are unchanged do not have severe pain is also a common observation in other studies (Bernardi, Catania, & Tridello, 2007; Coulling, 2005; Lai et al., 2003; Lui, So, & Fong, 2008; Tsai et al., 2007; Wang & Tsai, 2010; Yildirim et al., 2008). Distraction and relaxation are strategies that help in diverting attention away from pain (Macintyre et al., 2010; Mandeville, 2010; Tse & Chan, 2004). Coulling (2005) explained that these techniques help patients to feel less pain. Further, patients utilise their own coping strategies, such as distraction techniques and sleeping, to divert their attention away from pain. Misconceptions regarding pain assessment should be sorted out to help patients manage pain and to avoid neglecting their pain.

Individuals are influenced by cultural factors, including socioeconomic, geographic, religious and ethnic factors (Curry, 2010). Therefore, patients' individual attitudes and the cultural aspects of their presentation must be considered in pain management. The majority of participants (55.8 %) agreed that patients should be treated individually with due regard to their cultural uniqueness or diverse ways that may influence their perceptions of, and responses to, their pain experience. Many reported studies discuss this matter (Wang & Tsai, 2010; Reiman & Gordon, 2007; Tsai et al., 2007; Lai et al., 2003; Van Niekerk & Martin, 2001). It is also noteworthy that people differ in their beliefs about pain. In this study, 69% of participants thought that a patient's spiritual beliefs could influence his or her perception of pain. Van Niekerk and Martin (2001) stated that 81 % of participants in their study correctly acknowledged that certain religious beliefs may lead patients to consider pain a necessary feature in the fulfilment of life. Curry (2010) believed that nurses should be aware of individuals' distinctive cultural perspectives that may influence the pain management process and its outcomes.

These findings collectively indicate that the majority of participants possessed a fair knowledge base and positive personal beliefs with regard to patient variables of pain perception and the various influences and concepts that may alter pain interpretation and expression. Additionally, the influence of religion as an important consideration in pain management has been discussed, with more than 90 % of participants in survey-based research indicating that religion and associated belief systems influence pain perception (Bernardi, Catania, & Tridello, 2007; Erkes et al., 2001; Plaisance & Logan, 2006; Van Niekerk & Martin, 2001).

6.2.1.3 Nurses' Knowledge of Pharmacology

Pharmacology-based items are vital in pain management and have therefore been given substantial significance in KASRP survey result reporting. It is necessary for health professionals to have knowledge of the pharmacological approaches to managing pain. However, this proved to be another knowledge-deficit area among participants. In the current study, participants showed poor performance on pharmacology-based questions, which is in accordance with the observations made in many other international studies. Knowledge deficits and a lack of ability have been highlighted in many research studies that tested basic pharmacological knowledge, including choice of medication, drug action, routes of administration, untoward effects of opioid analgesics, equi-analgesic dosing (doses of a drug for different routes of administration that would provide equivalent analgesic effects) and selection of drug dosages (Lewthwaite et al., 2011; Wang & Tsai, 2010; Yildirim et al., 2008; Lui, So& Fong, 2008; Tsai et al., 2007; Matthews & Malcolm, 2007). Many studies have indicated that nurses are particularly deficient in knowledge about the drug *Promethazine*, or Phenergan (used to treat allergies and control pain), and its actions. Overall, the percentage of correct scores for *Promethazine*-related questions was 10–30 % (Yildirim et al., 2008; Bernardi, Catania& Tridello, 2007; Reiman & Gordon, 2007; Plaisance & Logan, 2006; Erkes et al., 2001; Brown, Bowman & Eason, 1999). In the current study, the percentage of correct scores for Promethazine-related questions was 37 %, which indicated that the majority of participants incorrectly believed that Promethazine potentiates the analgesic effects of opioids. Indeed, it is known that the sedative effects, respiratory depression and hypotension effects of opioids are increased by Promethazine (McCaffery & Ferrell, 1995). The question on the effectiveness of aspirin and NSAIDs on bone pain was answered incorrectly by the majority (66.3 %) of participants in the current study. This result also reflects the results of similar studies (Lewthwaite et al., 2011; Lui, So& Fong, 2008; Tsai et al., 2007).

Another area where nurses seem to have many misconceptions relates to the basic pharmacology of analgesics—especially opioids. More than two-thirds of participants in the current study (68.3 %) correctly answered that morphine is the best choice in treating cancerrelated pain, but only 14.5 % knew the proper route of analgesic administration for cancer pain. Regarding the question on the peak effect following oral morphine, only 40.9 % of participants gave the correct answer (1–2 hours). Similar findings have been reported in earlier studies (Wang & Tsai, 2010; Yildirim et al., 2008; Reiman & Gordon, 2007; Plaisance & Logan, 2006; Brown, Bowman& Eason, 1999). However, concerning the knowledge of the peak effect following intravenous morphine administration, 76.9 % of participants understood that it takes 15 minutes to reach its peak analgesic effectiveness. Therefore, nurses' knowledge of the pharmacokinetics of oral morphine is a particular concern.

Pain management practice requires a clear knowledge of equi-analgesic dosing. The utilisation of an equi-analgesics chart assists healthcare professionals in considering equivalency between pain-relief medication dosages (Brown, Bowman& Eason, 1999). Healthcare professionals must be conversant about the pharmacology of opioid analgesic preparations and be skilled enough to compute equi-analgesic dosages when managing pain. In view of the vital role of nurses in managing pain, they must be experts in these mathematical calculations to ensure maximum positive outcomes. Nurses must select the analgesic route and dosage of analgesia for as-needed (PRN) medications. The administration of the equi-analgesic dose requires significant care and responsibility to ensure that the transition from one opioid preparation to another does not produce adverse effects, such as an increase in pain (Gordon et al., 1999). Only one-third of participants (30 %) in the current study provided the correct equi-analgesic dosage when changing from intravenous (IV) to oral administration of morphine. A similar knowledge deficit was also reported by Tsai et al. (2007), who established that less than half (45.4 %) of emergency department nurses had the
required knowledge to work out the correct equi-analgesic dose of morphine. This has been a frequent observation across many studies, showing that participating nurses had only a meagre knowledge of equi-analgesic calculations (Lui, So & Fong, 2008; Reiman & Gordon, 2007; Brown, Bowman& Eason, 1999). The knowledge deficits regarding equi-analgesic doses may lead to many problems in the pain management process and could result in major errors in patients' pain management. Different routes are recommended for the administration of analgesics depending on different disease conditions and considering factors such as rapidity of action, maximum effect and patients' comfort. For example, for cancer-related pain, the oral route is the most effective option. It is an internationally accepted fact that the oral route for analgesia administration for cancer patients is the least expensive and most effective medication regime (McCaffery & Ferrell, 1995; Agency for Health Care Policy and Research, 1992). In a study by Matthews and Malcom (2007), only 39.8 % of nurses knew that the oral route is the preferred route of opioid administration to patients with cancer.

The present study also found vast knowledge deficits of, and unfavourable attitudes towards, opioid addiction and opioid-caused respiratory depression. The study highlighted many misconceptions about the effects of opioid analgesics. Interestingly, 82.5 % of participants could correctly identify the definition of addiction, but the majority could not distinguish between terms such as addiction, tolerance and physical dependence. The risk of addiction to opioid analgesics varies between different patient populations and treatment regimens. However, it is least likely to happen when opioids are used for acute pain management. In particular, opioid addiction is a very rare treatment complication in acute surgical pain relief (Ballantyne, 2003).

Deficiencies in the professional knowledge of nurses regarding pain management especially the use of opioids seems to be widespread. Kaki (2009) reported on the results of a survey

aimed at exploring nurses' (N = 325) opinions concerning patient satisfaction regarding pain relief, usage of opioids, follow-up dosing after initial usage, nurses' attitudes regarding addiction to opioids and physical dependence. Only 38.1 % of nurses considered morphine addiction a possibility with PRN (as-needed) treatment. Most nurses had not considered the addiction problem—particularly in situations where patient assessment was based on smiles and grimaces. As tolerance and physical dependence also decide pain relief and management interventions, few nurses (less than 30 %) had considered the issues of tolerance and physical dependence.

It appears that the information nurses receive about acute pain during their educational preparation may be of a superficial nature and without in-depth coverage of all relevant topics. Consequently, nurses are poorly prepared with the required knowledge and skills when confronted with patients in different situations. They are compelled in such instances to depend on doctors for guidance, which may result in patients suffering until the doctor arrives. According to Abdalrahim et al. (2011), another factor in this context is the attitude that comprises unconscious motivations for actions, as well as responses and reactions. The attitudes of nurses regarding their practice—in this instance, pain management—will be formed as per their education, training and on-the-job experience. As their knowledge is not sufficient to develop an appropriate attitude, nurses are being influenced by an incomplete personal belief system. As a result, several misconceptions arise about the understanding of pain, which hinders nurses' effectiveness in pain management practices.

In the current survey, the participant' education levels ranged from diploma to master's qualifications in nursing; however, a large number responded incorrectly to questions regarding pain assessment. For instance, a question was asked about administering sterile water (placebos) to patients and assessing their responses in order to determine whether some

patients were lying about pain. In reality, there is no such tool or practice, but most nurses responded with an incorrect answer, indicating their lack of knowledge on pain management and assessment.

Abdalrahim et al. (2011), whose study had a particular focus on post-surgical pain management, noted another misconception held by nurses, namely that they have the decision-making authority on whether a patient is in pain and therefore may not need to consider the patient's viewpoint. Thus, Abdalrahim et al. (2011) expressed the necessity of educating nurses about pain assessment and management in connection with postoperative pain.

In their survey, Abdalrahim et al. (2011) found that participants' responses indicated the prevalence of many misconceptions about the nature of pain. Abdalrahim et al. (2011) stated that an institution's pain management systems should be constantly monitored to ensure adherence to its pain management guidelines. Purpose-designed forms are useful in overcoming communication issues; however, Saudi hospitals generally do not seem to adopt this practice, as all hospital routines are regulated by the health authorities. Further, continuing training ensures that nurses are aware of current pain management practices (Davidhizar & Giger, 2004). This is important in the context of Saudi Arabia due to the high turnover of staff.

Abdalrahim et al. (2011) also presented the results of several studies on the introduction of nurses' education programs to enhance pain management techniques, practices and routines. In this context, they cited the example of Jordanian establishments of service training programs that focus on educating health team professionals—mainly nurses. The training focuses on the assessment, management and documentation of patients' pain and tries to

develop new practices and routines for nurses to assess and manage pain. The data were collected pre- and post-intervention by nurses, and comprehensive nursing documentation was considered part of the survey. They found that the mean score of 2.16 (on a scale of 0–5) for patient satisfaction regarding pain management increased to 3.26 after the recommended documentation of patients' responses was maintained. Thus, the participants recognised the importance of documenting the responses and reactions of patients. Abdalrahim et al. (2011) found that when implementing a post-surgical pain management program, the nurses' performance improved, with correct answers to the questions increasing to 16/21 (75 %) from a pre-implementation correctness of 9/21 (42.8 %). Hence, it is clear that nurses lack knowledge regarding pain management and that they can benefit from further education and training. Healthcare settings also have to provide protocols in order for nurses to provide quality care in pain management.

In the present study, the KASRP questions that the nurses found most difficult to answer were those that required them to make decisions or personal value judgments. The nurses reported satisfaction in their knowledge of pain management, whilst willing to learn new techniques to enhance their pain management skills.

Some studies have indicated that patients have reservations regarding nurses' knowledge levels and are not entirely satisfied with their pain management (Abdalrahim, 2009; Innis et al. 2004). Some of the reviewed studies have revealed that the introduction of comprehensive pain management practices with due importance assigned to patients' self-reporting is capable of delivering better results and improved patient satisfaction. The nurses in these studies also acknowledged their deficiencies in knowledge and the findings revealed the necessity for nurses to learn more about pain management during their initial training, as well as the need

for hospitals to offer enough further training as per the practical requirements (Al–Khawaldeh et al., 2013).

It was evident from both the reviewed studies and the current research that nurses want to learn and improve, and it is up to the educational organisations such as universities to introduce syllabuses on pain management into their curricula. Another level of responsibility can be assigned to the managements of hospitals to provide training for nurses regarding practical issues and problems they will encounter during their practice. The readiness of nurses to enhance their knowledge is evident through their expressed interest during their interviews to learn about pain management, as well as their description of self-learning through reading and research. In their interviews, they focused on the need for additional coursework and professional development. Their perceptions revealed their interest in focusing on additional coursework and training for professional development to enhance their practices of pain management and to develop communication skills in order to solve any cultural and linguistic problems during their practice.

Results from the interviews showed that the majority of nurses—regardless of whether they have enough knowledge or lack the necessary knowledge—are willing to learn from other nurses with more experience, such as their seniors or doctors. Some participants understood that knowledge is enhanced not only through experience, but also through better education. They stated that their knowledge of pain management increased when they achieved higher academic qualifications in nursing. Nurses expressed the opinion that diploma holders have comparatively poor knowledge of pain management, while those with bachelor and master degrees have better knowledge of pain management and the medicines used. In contrast, others claimed that newcomers do not have knowledge of the practices and medicines frequently used in a health setting. However, when testing participants' knowledge in relation

to various demographics, their academic qualifications had no statistically or clinically significant effect on their levels of knowledge regarding pain management. Some participants despite not having higher educational qualifications, demonstrated good knowledge of pain and were aware of possible alternative methods to medicines, such as diversion therapies. Clearly, the hospital structure and its systems, as well as the nurses' departments, play a crucial role in pain management practices and may enhance nurses' pain management skills. In service courses on the assessment and management of pain, as well as current pharmacological drugs, the pharmacokinetics and pharmacodynamics of these drugs have good prospects for updating nurses' knowledge regarding pain management and may prove to be useful when monitoring practices with the annual use of knowledge assessments via tools such as the KASRP.

The procedures for assessment and management of pain were fairly consistent among interview participants, although management options were employed. Some focused solely on providing pharmacological relief, while others described thoroughly assessing and providing non-pharmaceutical nursing support. Most maintained the focus on assessment, regardless. This dichotomy in responses, with a greater preference by some for alternative methods of pain management, was possibly a reflection of the nurses' education and training. Nurses with more experience, education and training in pain management were able to discern the effectiveness of alternative methods.

The participants in this study generally accepted that they have a high level of responsibility for the pain management of their patients. They claimed that this stems from the fact that nurses maintain frequent contact, and spent significant time, with patients and are therefore suited to more accurately assess their level of pain. To complete this assessment, the participants primarily described the use of numeric and facial pain assessment tools in addition to the patients' visual and physical cues. A lack of training regarding the use of these tools suggests a need for orientation and training in pain assessment.

6.2.2 Barriers to Achieving Optimal Pain Management as Perceived by Nurses

Interview participants cited common barriers to providing optimal pain management as language and communication barriers, cultural and religious barriers, and staff shortages. Language and communication difficulties were common given the multinational nursing staff. Some of these communication difficulties were also related to cultural differences or a lack of cultural knowledge. These cultural differences related to nationality, ethnicity, religion and gender roles. Participants described their own lack of language and cultural knowledge as the cause of the barrier, implying that further education and training would be effective at addressing these barriers. Having knowledge of various cultural and gender tendencies with regard to tolerating pain or expressing pain was felt to be helpful for nurses to more accurately assess the extent of patients' pain.

Pain may be under-reported, under-rated or under-treated if the person responsible for the pain assessment is not culturally receptive. For instance, when a number of Spanish-speaking children with cancer (N = 44) were examined regarding pain, 41 % suffered from pain before their visit to the clinic, and the most frequently noticeable locations on the body outline diagram was the abdomen (53.8%), lower back (46.2%) and upper chest (30.8%). Only 15 % of those who accounted for moderate to relentless pain received medication (Jacob, Sambuco, McCarthy & Hockenberry, 2008). The vocabulary used to describe pain and the body language employed by patients to explain their suffering may not reveal the severity of the pain when translated into English (Narayan, 2010). This may result in under-rated pain, which in turn can remain untreated. This is a clear example of a situation where the language and

cultural barriers hinder patients from correctly relaying the quality and quantity of pain to their caregivers (Jacob et al., 2008). According to the participants, there are instances when even those with adequate knowledge of pain management could not put such knowledge to practice due to a language barrier. Nurses must use their knowledge and expertise to understand the extent and nature of the pain. Their assessment starts with the location of the pain and then assesses its extent.

Of all of the possible limitations, language barriers are the most significant, particularly in the context of countries such as Saudi Arabia, where English-speaking expatriates constitute a major proportion of the nursing staff (Almalki, FitzGerald, & Clark, 2012). In this situation, the language barrier may exist in two contexts: Saudi nurses working with non-Saudi patients and expatriate nurses working with Saudi patients. The problem exists when the nurse does not know Arabic or the patient does not know English. Most expatriate nurses know English better than Arabic, and Saudi patients use Arabic as their primary language and thus know it much better than English. Thus, a language barrier exists when nurses try to communicate with patients to assess their pain. According to Narayan (2010), the language barrier and interpretation problems make it difficult for healthcare providers to adequately assess a patient's pain, leading to suboptimal pain treatment outcomes. Divi et al. (2007) noted that problems may occur if an interpreter is less proficient at translation. Davidhizar and Giger (2004) pointed out that words and meanings may be interpreted differently due to different dialects, and this occurs between nurse and patient or doctor and patient resulting in errors in assessments and incomplete pain management. Ineffective communication between the nurse and the patient results in inadequate pain assessment and eventually the adoption of substandard pain management interventions (Narayan, 2010). Communication issues arise between nurse and patient or doctor and patient that may result in errors in assessments and incomplete pain management.

The language barrier may include lapses in nurse-patient or doctor-patient communication, which may result in errors in assessments and incomplete pain management, despite the presence of qualified and knowledgeable healthcare staff. Hence, it depends on the hospital settings and systems that enable expatriate nurses to face language barriers effectively. Even if expatriate nurses have a degree of fluency in Arabic, the different dialects-where a single word or phrase could be interpreted in a variety of ways—can be puzzling and may result in misconceptions. A barrier also occur when there is a necessity to change interventions or medications as a result of an increase or decrease in the pain being experienced by the patient. It is clear that most patients are not able to convey their pain levels and the nature of their pain and nurses are not able to understand them when the situation is critical. The language barrier occurs when patients or nurses do not speak the same language. If both the patient and nurse can converse in English, there remains an issue with understanding accents and dialects. The language barrier can be mitigated to an extent by using documentation such as purposedesigned forms. However, hospitals generally do not seem to adopt this practice. Matters such as the introduction of specific forms are the responsibility of the policy makers of the hospital system. In the course of academic nursing studies, nurses do not receive any special training regarding such administrative matters and may experiene difficulty in interpreting the results.

A shortage of nursing staff was cited as another factor in pain management. Mitchell (2009) argued that nurse shortages were caused by limited career opportunities, inadequate education, or lack of resources. Nurses may be expected to work long hours, causing stress and reduced performance. Saudi Arabia is expanding services in healthcare infrastructure and human resources to address the shortage of nurses (Mitchell, 2009). With the addition of differences in language and cultural barriers the workload is further increased, as the time taken by nurses to assess each patient increases due to a lack of understanding and problems

with communication between the nurse and patient. Combined, this leads to a heavy workload, which can create further stress for nurses and may translate to a lack of concentration and an increased incidence of drug errors in the ward. According to participant D, the resultant lack of quality pain management may also be due to patient neglect, which arises when patients make more demands on the few available nurses. These observations highlight the necessity for educational institutions to develop effective pain management courses. Hospitals should focus on language and culture training, and increasing nurse numbers. As suggested by many reports, adequate pain assessment and management is obstructed by time constraints. Increased workloads, a shortage of nurses, interruptions and non-nursing responsibilities have negative effects on nurses' time to concentrate on patients' pain requests, as reported by nurses in worldwide studies (Rejeh et al., 2009; Manias et al., 2005; Schafheutle et al., 2001).

The problem of nursing shortages in hospitals and the resulting increased workloads was stressed by most of the interview participants in this study. In some cases, nurses were required to care for 10–15 patients, which limited their ability not only to respond to patients when called, but also to have adequate time to assess patients' pain and provide alternative pain management if needed. This factor, in conjunction with language and cultural barriers, could lengthen the time needed to achieve communication between nurses and patients, and have serious implications for the efficiency of pain management practices.

6.2.3 Demographic and Cultural Determinants Affecting the Delivery of Effective Pain Management

The third research question related to the demographic and cultural determinants that affect the delivery of effective pain management. The results from the quantitative phase of the study showed that demographic factors influenced nurses' knowledge of pain management were (a) work location, (b) hospital type, (c) age, (d) years of experience and (e), nationality. Nurses with the highest levels of knowledge regarding pain management: (a) worked in the ICU (b) were located at Hospital E; (c) were 41–60 years old; (d) had 15 or more years of experience; and (e) were Filipino. The effect of nurses' qualifications, positions, number of pain courses attended, gender and religion were not significant.

The cultural factors that affect the delivery of effective pain management were clarified by many participants in the interviews conducted during the second phase of the study. As revealed during the semi-structured interviews, patients' cultural factors could affect the delivery of effective pain assessment and management. As observed by participants, some patients do not overtly express their pain; rather, they attempt to tolerate it for religious and cultural reasons. Indeed, pain is a subjective experience with attached physical, emotional, religious, spiritual and cultural dimensions. Patients' cultural backgrounds are often a consideration when nurses and doctors examine patients and make judgments.

To effectively relieve pain, nurses should be aware of cultural and religious practices, especially when providing end of life care. These include rituals in the presence of family. Muslims view illness as the atonement of sins; hence, even as they seek medical attention, the treatment adopted also integrates the spiritual aspect by constantly praying and reading the holy Qur'an (Mughees, 2006). Similarly, in the case of Christians, many Hispanics and Latinos, who are basically Catholics, believe that cancer pain is a kind of punishment that must be endured for one to enter heaven (Juarex, 1997). Al-shahri (2002) pointed out that in Muslim societies, spiritual healing is often chosen when faced with terminal illnesses. Further, during the month of Ramadan, some Muslim patients prefer to engage in fasting despite their illness.

Culture has a great influence on beliefs, morals, religion, family roles and descriptions of illnesses, as well as on how individuals feel and express pain (Narayan, 2010; Shepherd et al., 2010). The influence of culture was illustrated in one report, which suggested that in cases of long bone fractures, Caucasian patients were more prone to suffer from pain than patients from other cultural groupings (Todd, Tew & Macdonald, 2000). Such considerations may result in hospital employees not paying much attention to patients' accounts of the severity of their pain (Rupp & Delaney, 2004). Chinese patients may not express their pain because they view pain as a distortion in their body, which must be endured (Chen, Miaskowski, Dodd& Pantilat, 2008). Likewise, they consider that pain should be tolerated impassively. Thus, it may prevent them from sharing their pain like many Hispanic patients do (Anderson et al., 2002; Campbell et al., 2009). Vietnamese and Filipino patients are dependent on the older male member of the family to make medical decisions for them (Hooke, 2007). Nurses should remain professional, suspending their own religious and cultural beliefs and responses, and carrying out their duties and responsibilities towards their patients (Davidhizar & Giger, 2004).

Of significance, participants stated that male patients often attempt to conceal their pain in the belief they should be stoic and endure pain. Alternatively, Almutairi and McCarthy (2012) referred to Saudi women's shyness and inability to express their pain to an 'outsider'. This reticence could mislead nurses when assessing patients' pain, thus interfering with appropriate

nursing care. For example, the use of a face pain scale would not be reliable in these circumstances.

For pain assessment and management, nurses require knowledge and expertise (Leininger & McFarland, 2002). Further, before deciding on the treatment, differences in communication modes and family set-ups should be kept in mind (Thibodeaux & Deatrick, 2007). Nurses must find out how individual patients feel and what they perceive regarding the pain experience in order to tend to them according to their culture (Narayan, 2010). Nurses depend on the responses of patients to conduct their pain assessment. Thus, the language barrier has been clearly identified by the majority of participants as hampering their ability to apply effective pain assessment and management.

Kaki (2009) found that nurses are not aware of cultural perceptions of pain and its management. In this regard, Lovering (2006) conducted a useful study regarding the multicultural healthcare setting and pain management. The researcher assumed that patients and healthcare professionals bring their own cultural attitudes to the communication and interpretation of patients' pain experiences. She particularly mentioned the Saudi Arabian multicultural healthcare settings, which challenge nurses' opinions about cultural beliefs and attitudes to pain. As a result, the study found that it is necessary to understand cultural perceptions and their effect on pain management and assessment. She cited the findings of Ramer et al. (1999) regarding the influence of culture on the perceptions of pain and its expression. Taking a cue from Ramer et al.'s (1999) study, she found that ethnicity influences the expression of pain and patients' responses to it. Hence, pain management practices must consider culture, health behaviour and pain. However, nurses may not have enough knowledge of coordinating culture, health behaviour and pain, as pain management was not a

focus during their studies, nor were they given any training concerning the cultural aspects of assessment.

Similarly, the findings of Lovering (2006) revealed that the experience of pain was different across cultures. Lovering found that Asians, Filipinos, Saudis and Irish like to verbalise pain, while Africans are stoic. However, some Saudi patients are an exception and do not express pain. The findings conclude that pain has personal and cultural meanings.

6.2.4 Pain Management, Culture and Care

Pain management can be described as an act of compassion that relates to the concept of care embodied in the CCDU theory proposed by Leininger. In view of the large volume of literature referring to the importance of the cultural background of both patients and carers in their approaches to pain management, this study assigned a special focus on culture as a factor, along with others such as language and religion. Culture influences how patients respond and express their pain, bearing in mind that pain has social, physical, spiritual and psychological dimensions. In view of this, issues of communication and culture affects non-Saudi nurses' efficiency in delivering appropriate pain management services to patients (Leininger, 2002).

The participants in this study were found to be in a dichotomy regarding the use or non-use of alternative pain management methods. Although a few use them, they generally do not seek to use alternative methods before trying medication. This practice not only makes nursing practice mechanical in nature, but it also neglects patients' pain. For example, nurses do not seem to consider talking to patients in an effort to reduce their pain because they have not been trained that way. Similarly, if the pain persists following the initial interventions, nurses

depend on doctors' instructions, as most of them are not familiar with offering combined pharmacological interventions.

Regarding nurses' responsibility for pain management and use of tools and methods, the results showed that participants used numeric and facial pain assessment tools, skills learned on the job rather than through training. If hospitals offer training in pain management and assessment tools, the survey participants felt capable of showing greater responsibility during pain management and assessment. The nurses are ready to receive the necessary training and would appreciate this to consider care at the bedside because they believe that frequent contact with patients helps them to assess their pain much better.

Hospitals need to establish a system of feedback from patients and nurses in order to understand the practical consequences of present practices. Depending on the feedback, management must continuously develop training programs for newly recruited nurses so they can deal with communication problems such as culture, religion and language.

However, the language barrier cannot be completely overcome by training, as nurses cannot be trained in all languages (and dialects) of the patients who take treatment in a healthcare setting in a country like Saudi Arabia. Hence, in addition to training arrangements, it would be better to arrange translators with nursing backgrounds for nursing teams so they can help nurses overcome language barriers when communicating with patients. The training programs can address cultural differences in a nation and those that occur across nations in patients belonging to different nationalities and religions. For example, some cultural aspects are common to Asians, but patients still differ substantially regarding culture, religion and language. Hence, continuous training or workshops would help nurses regarding ethnicity, religion and gender roles in managing and assessing pain.

This solution is linked to participants' accounts of their lack of knowledge, education and training in considering culture, language, religion and ethnicity during assessments and with subsequent pain management interventions. The participants were aware of their lack of overall knowledge regarding pain management, and they indicated that further training would be beneficial to their practice. Hospitals should provide training regarding culture, as multicultural healthcare settings differ in culture due to their geographical location. For example, the multicultural healthcare setting in the Hail region in Saudi Arabia may differ from a multicultural hospital setting in London in the United Kingdom. The former setting may comprise mostly Asians and few Westerners, while the latter may have mostly Westerners and Latin Americans as well as Asians. Hence, the training regarding culture and religion for nurses depends on the nature of the problems being faced by the healthcare setting so that nurses can overcome communication barriers.

Alongside increased staff numbers, adopting comprehensive pain management would help, as nurses are ready to follow new guidelines that offer better routines for auditing the interventions and self-reporting of patients. This means that the hospital must prompt nurses to use non-pharmacological interventions alongside medications, and it must provide guidelines for the compulsory usage of non-pharmacological interventions. If regular medication is prescribed, alternative interventions should also be considered as an integral part of the planned care. Clear guidelines for medical and alternative therapies should be available. This means that health systems should introduce a comprehensive pain management system in a multicultural healthcare setting.

However, optimal pain management cannot be solely achieved by guidelines, new routines and methods or techniques. Effective role-modelling from senior staff, as well as appropriate pain assessment procedures, would increase nurses' expertise. In addition, healthcare settings must develop culturally appropriate guidelines and should not blindly depend on Western methods, as the cultural orientation of patients in Western healthcare settings is different from that of patients in the Saudi healthcare setting.

6.3 Conclusion

This chapter discussed the knowledge of pain assessment and management among a sample of nursing staff from five hospitals in the Hail region of Saudi Arabia, and explored the barriers they faced in providing effective pain management. The results of this study revealed that the nurse participants, irrespective of being local or expatriate, are not equipped with adequate knowledge regarding pain assessment and management, particularly in pharmacological and non-pharmacological interventions. Further, their professionalism was not evident in assisting patients experiencing pain. The nurses reported many barriers to achieving optimal pain management, such as language and communication barriers, cultural and religious barriers, staff shortages and heavy workloads, and the inadequacy of training received in the area of pain assessment. Regarding the demographic and cultural factors that affect the delivery of effective pain management, the results showed that nurses' nationality, years of experience, age, hospital and department have a substantial effect on their overall ability to assess and manage pain.

Despite nurses spending more time with patients who suffer from pain, patients remain underdiagnosed and their pain is not well manage, due to nurses' ongoing lack of knowledge. Nurses are aware of the effect of opioids in relieving pain, but they do not have enough knowledge about their side effects and are concerned about addiction. They are not in a

position to assess pain in order to change the dosage or medicines through communication or from client reactions after administering an opioid.

Much has been written concerning nurses' insufficient knowledge of the pharmacology of analgesics. Nurses are routinely called upon to administer pain-relieving medications and therefore should know the interpretations of dosages, routes, actions and potential side effects of these medications. One of the major problems indicated in the literature is the lack of understanding in relation to the proper administration of analgesia to patients with cancer pain.

The cause of pain in patients can be varied, such as trauma, burns, postoperative injuries and cancer-related pain. The different origins of patients with pain demand different types of knowledge in order to assess and manage their pain. Equally important is the cultural and linguistic needs of patients in order to enable nurses to assess and manage their pain. It is necessary to understand the different reasons for pain alongside the patients' cultural and linguistic needs.

Nurses must use non-pharmacological interventions alongside medications. With the exception of those who practise some form of intervention, the majority of nurses in this study do not try to use non-pharmacological interventions for pain management. Further, academic studies may enable nurses to use non-pharmacological interventions in practice, as they are trained in knowing causes and effects. Nurses' expertise in non-pharmacological tools encourages them to use the pain assessment tools, as the extent of the pain may decide the usage of non-pharmacological interventions.

Regarding culture, it is necessary for expatriate nurses to understand the cultural perceptions of Saudi patients and how to ask them about pain. Even if they are proficient in the language,

the ways and means may vary for different cultures and patients may be offended or may not want to answer nurses' questions concerning their personal experience of pain. Hence, both culture and language are individual and combined barriers. Similar results were found among both expatriate and local nurses in this Saudi Arabian cohort.

From this study, it is evident that nurses have experience with patients experiencing pain, and that this experience is derived from clinical practice. However, they reported that they did not have enough academic support to formalise their pain management education. The finding that nurses lack knowledge regarding pain, its management and other relevant issues have been discussed in relation to the published results of many similar studies. The review, discussion and findings of the survey emphasised the need to enhance nurses' knowledge regarding pain and its assessment and management. The study also found the necessity to place greater emphasis on instructions regarding pain management during initial nurse training. Any training or formal education needs to be assessed as part of ongoing competencies that need to be achieved in the clinical area.

Chapter 7: Conclusion

7.1 Introduction

This chapter considers the thesis as a whole, as well as final considerations and reflections on the study design and its strengths and limitations. Recommendations are made regarding policy, practice, education and training, and for possible future research directions. This chapter then summarises the thesis with a reflection on the methods is included.

7.2 Reflection on the Study and its Design

The majority of similar studies that have explored nurses' knowledge have focused on quantitative analysis only and not followed up statistical results with a discussion with nurses about clinical practice outcomes. The recommendations based on the present studies highlight the need to explore the barriers that nurses face while managing pain. Thus, this study aimed to explore nurses' knowledge regarding pain and went further to investigate the barriers to providing adequate pain management as perceived by nurses working in Saudi Arabia hospitals. Demographic and cultural factors that affect the delivery of effective pain management have been explored, which was appropriate given the multicultural workforce that was studied. The first phase of this study was based on a survey of knowledge (KASRP) testing views and opinions regarding pain management expressed by a sample of both expatriate and local Saudi nurses who were working in hospitals in the Hail region of Saudi Arabia. The questions in the survey were about the perceived knowledge of nurses regarding pain assessment and management, and the familiarity they have with the medications they administer and the interventions they utilise to manage patients' pain in different

circumstances and in a variety of clinical settings. A total of 303 questionnaires were collected, and the responses were analysed.

The purpose of the second phase of the study (qualitative data) was to obtain an understanding of the clinical practice issues that were considered important by the respondents for the management of pain experienced by Saudi patients. The second phase involved semi-structured interviews with 28 participants to further explore their perceived facilitators and barriers to proper pain management. The purpose of the interviews was to elicit information on how cultural differences among Saudi national and expatriate nurses might affect the assessment and interpretation of patients' pain and what clinical actions are taken following the assessment of pain. The questions posed to the participants were deliberately kept open-ended, in the format of a semi-structured interview. The participants were considered to have an in-depth understanding of the important issues they experienced when assessing and providing pain management as part of their nursing practice. A workforce that has sound practice in pain management is important, as good role-modelling is essential to novice nurses and new recruits to the Saudi Arabian healthcare system.

The sample of participants chosen for the qualitative analysis comprised nurses from various countries other than Saudi Arabia, and with different cultural and religious backgrounds compared to Saudi Arabians. The inclusion of nurses from different nationalities ensured enough diversity so that the differences in opinions based on nationalities might be considered a valid result of the study despite the discrepancy in the respective numbers—the majority of nurses were from the Philippines, a few from India and only four from Saudi Arabia. The participants were of several professional rankings

It was found that although a significant proportion of the participating nurses possessed some knowledge of pharmacological interventions, their knowledge was incomplete. It was also evident that the nurses, by and large, were aware of such knowledge deficiencies and were keen to update their knowledge as part of their commitment to proper patient care. It was identified that health services could make a change to pain management practice by simply promoting interventions and methods that encourage pain assessment and by implementing steps to improve nurse–patient communication. A particular area of neglect is the possible use of non-pharmacological interventions to reduce pain. The main reason for these issues, from the managements' perspective, is a shortage of staff. The nurses agreed that staffing inadequacy was a major reason for insufficient nurse–patient communication.

7.3 Strengths of the Study

The fact that this research study was conducted in a clinical setting and therefore is practicebased adds to the importance of the findings. Thus, the implications to practice are readily applicable to practice and can be implemented. The questions were framed in a manner that enabled the assessment of nurses' knowledge unambiguously. The nurses used the opportunity to assess their knowledge, and the high return rate of the survey may indicate that nurses found the topic of inquiry important for nursing practice. The study also revealed the gaps in their knowledge regarding pain management and the problems they faced as a result. Thus, the study was able to provide real insights into current practice and to determine the strengths and weaknesses of the nurses' pain assessment skills so that appropriate corrective measures could be recommended. The survey was also able to identify positive aspects, such as the attitudes of nurses, in that they expressed great willingness to learn and implement new methods, techniques, interventions and routines to enhance their effectiveness.

The study employed mixed methods of qualitative and quantitative analysis. The quantitative analysis was used to determine the knowledge of participants regarding pain management, and the qualitative analysis was used to explain the findings and barriers as perceived by participants. Due to its multi-dimensional nature, the combination of quantitative and qualitative research allows one to explore range of different aspects within a single study. Moreover, both quantitative and qualitative methods have individual strengths and weaknesses. Thus, using a combination of the two can overcome such limitations and generate a significant amount of information when compared with single-method research (Daymon & Holloway, 2010). The questions in the qualitative phase focused on understanding the problems faced by nurses without exhorting them to mention them directly.

This study has extended the use of the KASRP by considering the problems faced by nurses due to differences in culture, religion, language and perceptions. The strength was to include both local and expatriate nurses working in Saudi Arabia. Indeed, this makes it a truly representative sample, as the nursing workforce in Saudi Arabia largely comprises expatriates, and a focus on nationals alone would not have provided the comparative results between the populations. The reason for including local and expatriate nurses is due to the understanding that both categories of nurses have faced similar problems but in different circumstances. For example, local and expatriate nurses were constrained by issues related to culture, religion and language. Language and cultural barriers were faced by local nurses in Saudi Arabia with expatriate patients, and expatriate nurses faced the same issues with local patients. Whether local or expatriate, nurses faced problems regarding communication with patients.

The sample of respondents chosen for the qualitative analysis comprised nurses whose origins could be traced to several countries. The inclusion of nurses from different nationalities

provided sufficient diversity so that the differences in opinions based on nationalities might be considered interesting despite the discrepancy in the respective numbers.

The nurses may have benefitted from participating in the survey and responding to a questionnaire of this nature. It will help them to assess their own knowledge level and enable them to judge their own attitudes towards pain and implementing best practice care for patients suffering from pain. Most importantly, this study adds to the body of literature in regard to nurses' knowledge and attitudes regarding pain management within the context of an Islamic society. With multifaceted approach to data collection and triangulation methods, this study makes a claim for unique and new knowledge concerning pain management and cultural factors encountered in an Islamic country. This information can be built upon, but it has current applicability and also considers future directions within the recommendations that are made.

7.4 Limitations of the Study

The most important one relates to compromising the generalisability of the research findings. As a result of restricting the current study to the hospitals in a single region of Saudi Arabia, as well as the fact that the survey assessed the situation at a single point of time (time and place restriction), its validity for application to the entire Saudi Arabian healthcare system may be limited. In fact, health services may have addressed these problems and be able to offer best practice solutions. This was not the case in the Hail region.

Despite having a sufficiently large sample size to represent the relevant nursing population, when they were divided on the basis of gender and religion, it was found that the number of male participants and those of Hindu religion—21 and 11 respectively—were comparatively

small, which may result in Type II errors. This survey result should also be considered when deciding on the validity of the statistical analyses of the data provided by the male and Hindu participants. Moreover, data were gathered using convenience sampling, which has the inherent disadvantage of not being representative of the total population with regard to all demographics.

Although it was expected that the questionnaires would be answered individually and privately, and instructions to that effect were given, it cannot be ruled out that at least some participants may have discussed the questions and sought assistance from each other in answering them. However, if this occurred, it did not have a significant effect on the overall scores, with the majority being well below the required 80 %.

Another limitation was the closed questions in the survey, which included true and false matching, and multi-choice questions. Participants may have answered these questions by guesswork rather than using their actual knowledge.

7.5 Recommendations Arising from the Study

7.5.1 Recommendations for practice.

At the practice level, steps should be taken to ensure that nurses assess a patient's pain at regular intervals and record the findings, along with actions taken, on a purpose-designed 'pain assessment form'. Standards for pain assessment by the Joint Commission Nursing staff should utilise valid and reliable tools to assess for pain and use systematic ways of recording results and handing these over to other staff. A standardised recording system may overcome the problems arising from verbal communication lapses.

Another important recommendation in this regard is the need for a greater emphasis on using non-pharmacological interventions such as musical therapy, distraction, relaxation, positioning, exercise, hot–cold packs and other methods. The survey showed that the majority of nurses were unaware of non-pharmacological interventions.

The need for further pain assessment after the administration of medication was not fully understood by the participants. This is of particular concern in multicultural and multilinguistic healthcare settings such as that of Saudi Arabia. Thus, further training in pain control and cultural competency could improve their knowledge regarding these concerns.

Due to communication difficulties, most nurses rely on facial expressions instead of assessing pain using self-reporting by patients. This conflict between what the patient wants to say and what the nurse sees may have a negative implication to practice, contravening the expectations of the theory of pain management, which place a premium on the patients' interests and individual experiences. Thus, it is recommended to have an interpreter during the assessment of patients' pain. Alternatively, nurses should be given language courses (Arabic) in order to enhance their ability to communicate with the predominantly Arabicspeaking patients.

The survey indicated that nurses recognise their lack of knowledge of the language and agree that they are not able to communicate with patients at the desired level. They also agreed that they do not think about non-pharmacological interventions as part of pain management. Nurses must update their knowledge regarding pharmacological and non-pharmacological interventions. When non-pharmacological interventions are included in routine nursing practice, the implications could be a decrease of the required dosage of medications. However, due to a lack of knowledge of alternative methods of pain management that do not involve medications, nurses rely more on medications rather than non-pharmacological and communicative tactics.

Nursing practice could improve if nurses develop knowledge of non-pharmacological interventions alongside advanced communication skills that reduce pain. Policy needs to be developed to orientate new staff and assess their skills on a regular basis. The hospital administration must analyse the reasons underlying shortage of staff and remedy it by a comprehensive recruitment and retention strategy to alleviate nursing shortages and workloads.

7.5.2 Recommendations for training and education.

There is an urgent need for extra education concerning pain assessment and management. A focus on changing the culture of care towards implementing evidence-based nursing pain management practice would result in greater clinical accountability for pain management.

It is recommended that training takes place at various levels of nurse training, with sufficient coverage of the subject in the curriculum of nursing colleges and on-the-job training, supplemented with mandatory attendance of regular training courses and a practical clinical assessment within the healthcare working environment. Training methods need to be introduced by the management of healthcare settings as policy for annual competency achievements. Nurses should enter the workforce with skills in pain management that can be built upon and enhanced, and the educational preparation of nurses must be achieved by educational organisations such as universities.

An additional requirement—especially in multicultural and multi-linguistic hospital settings—is the development of the cultural competence of nurses. Moreover, cultural orientation for nurses should be implemented, and the program should include aspects of local population healthcare beliefs and practices, as well as the principles of transcultural nursing, in order to provide culturally congruent care. The language course should also be provided to nurses, as people use language to express their pain, which is embedded in culture. These courses could help to develop nurses' communication skills to communicate with patients of an Arabic-speaking background.

7.5.3 Recommendations for policy.

Hospital policies, which include the systems installed in healthcare settings and the stipulated procedures to be followed by nurses, require greater attention to implement better and more effective pain management practice. Changes to clinical practice require leadership and clear policy to support best practice. The hospital policy must aim to adopt best practice and also regularly update policies and guidelines relating to pain management. Policy makers must establish and monitor the standards for pain assessment and management as recommended by JCAHO. Moreover, the health service administrators and the policy makers of the Saudi health services must revise the nurses' recruitment and retention strategy. This will result in a more comprehensive recruitment and retention strategy

7.5.4 Recommendations for future research.

The present findings, while contributing to the knowledge base on the subject in general, have particular significance for hospital systems in Saudi Arabia. However, as outlined above,

there are certain limitations to the study, and future research on the subject should take these limitations into account. A study along the same lines as the present study, but with the inclusion of patients' self-reporting of their experiences of pain management, will give added value to the findings. This methodology of gathering data from both sources provides an opportunity to cross-check the responses of patients and nurses so that any anomalies can be corrected.

Similarly, supplementary information obtained from other relevant personnel, such as the management of healthcare settings and the teaching staff of nurse education facilities, will be equally useful.

The replication of this study involving hospitals in other regions of Saudi Arabia is strongly recommended in order to explore other healthcare settings and determine the validity or otherwise of the applicability of the present findings to the entire country. Similar studies should also be conducted to evaluate nursing students' knowledge and attitudes regarding pain management.

Another useful opportunity for research would be to carry out similar studies in different locations and in varying healthcare settings involving nurses who have successfully completed an intensive pain education program prior to the study. Such a study has the potential to definitively delineate the effect of knowledge deficiencies from any other limiting factors in pain assessment and management.

Further, research that includes the involvement of the management of healthcare settings or higher officials and faculty of nurse education institutions such as universities would provide a great opportunity to consider best practice guidelines. In view of the roles played by both of these groups in the education and management of nurses, it would be prudent to seek their viewpoints as part of future research into pain management policy implementation and practice reviews.

7.6 Study Conclusion

The goal of pain management is to address all dimensions of pain in order to provide maximum relief with minimal side effects. Regarding the management of pain in patients, a review of the research literature, anecdotal reports and the general opinions of both healthcare personnel and patients suggest that the majority of hospital patients do not receive adequate pain care. This has been attributed to many factors, including nurses' lack of required knowledge and skills, attitude issues, myths, fears and cultural issues.

The insights gained from this study into the knowledge and attitudes of nurses involved in pain care at five major hospitals in Saudi Arabia will be very valuable to healthcare providers, the public and other authorities concerned with the healthcare system in Hail, Saudi Arabia. The findings conclude that nurses at these hospitals display an extensive knowledge deficit, resulting in attitudes that are not conducive to good pain management practice. The knowledge deficiency is particularly a problem in certain aspects, such as basic pharmacology in general and the use of opioid analgesics in particular. In addition, the apparent total lack of knowledge of the potential use of non-pharmacological approaches in pain management is concerning. There is ample literature on the effectiveness of alternate methods such as music therapy, distraction, relaxation, meditation, exercise and hot–cold packs in managing pain. The knowledge deficiency is compounded by issues arising from nurses' attitudes and idiosyncratic beliefs, such as unwillingness to consider patients' self-reporting of pain and

opting to place greater trust in their behavioural manifestations, erroneous perceptions regarding opioid addiction, and exaggerated fears about serious side effects of opioids.

The nurses who participated in this study were different from those in several other reported studies in that they accurately perceived that they were not endowed with the knowledge required for handling the important clinical role of pain management. They attributed problems with overall pain management knowledge to deficiencies in their primary nurse training curriculum, as well as a lack of opportunities for subsequent training. On a positive note, these nurses were eager to update their knowledge, as was evident by their attempts to enhance their knowledge through self-study.

Another major problem area—perhaps a rare situation applicable only to countries such as Saudi Arabia—relates to the nurse–patient communication barrier. This is due to the fact that a very high proportion of foreign nationals constitute the nursing workforce. These nurses, who are not proficient in speaking Arabic, are required to care for Saudi nationals who, by and large, speak only Arabic. The result is a major breakdown in communication, and effective communication is a fundamental requirement in the provision of proper healthcare. Indeed, some of the points referred to earlier, such as nurses not willing to consider patients' self-reports, are a manifestation of the language barrier. It also diminishes opportunities for the health education of patients by nurses. Coupled with the language barrier is the cultural barrier.

Although the nurses in this study showed an interest in learning about pain management and medications, the findings indicated that nurses in Hail region hospitals—irrespective of being local or expatriate—have a major deficiency in their knowledge regarding pain assessment and management. They also have problematic attitudes towards pain assessment and face

problems in communicating with patients while assessing and managing their pain. They offer interventions with minimum communication and with few non-pharmacological interventions. The main barriers identified by these nurses were language and communication, increased workloads due to staff shortages, cultural and religious factors, a lack of education, and a lack of knowledge regarding pain management.

Organisational changes to healthcare settings are capable of promoting the understanding of the cultural orientation of pain management by nurses. The study recommends changes to the educational system so nurses leave educational institutions with more knowledge about pain management, which is crucial for the quality care of patients. It has been found that problems regarding communication, which arise from differences in culture, religion and language, are equal to the lack of knowledge of pharmacological and non-pharmacological interventions. This lack of knowledge is further complicated by a lack of cultural competency regarding culture, language and religion, as well as perceptions of patients not being able to self-report their pain experiences and will be remedied by cultural orientation competency courses.

References

- Abdalrahim, M. (2009). Postoperative pain assessment and management: The effects of an educational program on Jordanian nurses' practice, knowledge, and attitudes. Retrieved
 1 February 2012 fromhttps://gupea.ub.gu.se/bitstream/2077/20316/
 4/gupea 2077 20316 4.pdf
- Abdalrahim, M. S., Majali, S. A., Stomberg, M. W. & Bergbom, I. (2011). The effect of postoperative pain management program on improving nurses' knowledge and attitudes toward pain. *Nurse Education in Practice*, 11(4), 250–255.
- Abdul Rahman, M., Abu-Saad Huijer, H. & Noureddine, S. (2013). Lebanese nurses' knowledge regarding pain management. *Lebanese Journal of Nursing*, *3*, 10–12.
- Aboul-Enein, F. H. (2002). Personal contemporary observations of nursing care in Saudi Arabia. *International Journal of Nursing Practice*, 8(4), 228–230.
- Abu-Zinadah, S. (2006). Nursing situation in Saudi Arabia. Retrieved 12 May 2013 from http://scfhs.info/NursingSymposium/index.php?option=com_content&task=v iew&id=16&Itemid=32
- Agency for Health Care Policy and Research. (1992). *Clinical practice guideline. Acute pain management: Operative or medical procedures and trauma*. (AHCPR Pub No. 92-0032). Rockville, MD: Agency for Health Care Policy and Research.
- Al Mazrou, Y., & Salem, A.M. (2004). [Primary health care guide]. Riyadh, Saudi Arabia, Ministry of Health, [in Arabic].
- Al-Aameri, A. S. (2000). Job satisfaction and organizational commitment for nurses. *Saudi Medical Journal*, *21*(6), 531–535.
- Al-Ahmadi, H. (2002). Job satisfaction of nurses in Ministry of Health hospitals in Riyadh, Saudi Arabia. *Saudi Medical Journal*, *23*(6), 645–650.

- Al-Ahmadi, H. (2009). Factors affecting performance of hospital nurses in Riyadh Region, Saudi Arabia. *International Healthcare*, 22(1), 40–54.
- Alamri, A. S., Rasheed, M. F. & Alfawzan, N. M. (2006). Reluctance of Saudi youth towards the nursing profession and the high rate of unemployment in Saudi Arabia: Causes and effects. Riyadh, Saudi Arabia: King Saud University.
- Aldossary, A., While, A. & Barriball, L. (2008). Health care and nursing care in Saudi Arabia. *International Nursing Review*, 55(1), 125–128.
- Alharthi, F. et al., (1999). *Health over a century*. Riyadh, Saudi Arabia: Ministry of Health and ASBAR Centre for Studies Research and Communication.
- Al-Homayan, A., Shamsudin, F., Subramaniam, C. & Islam, R. (2013). Analysis of Health Care System - Resources and Nursing Sector in Saudi Arabia, *Advances in Environmental Biology*, 7(9): 2584-2592.
- Alhusaini, H. A. (2006). Obstacles to the efficiency and performance of Saudi nurses at the Ministry of Health, Riyadh Region: Analytical field study. Riyadh, Saudi Arabia: Ministry of Health [in Arabic].
- Al-khawaldeh, O. A., Al-hussami, M. & Darawad, M. (2013). Knowledge and attitudes regarding pain management among Jordanian nursing students. *Nurse Education Today*, 33(4), 339–345.
- Al-Krenawi, A. & Graham, J. R. (2000). Culturally sensitive social work practice with Arab clients in mental health settings. Retrieved 15 February 2013 http://www.socialworkers.org/pressroom/events/911/alkrenawi.asp?print=1
- Almalki, M., FitzGerald, G. & Clark, M. (2012) The relationship between quality of work life and turnover intention of primary health care nurses in Saudi Arabia, *BMC Health Services Research*, 12 (1), 314.
- Almeida, T. F., Roizenblatt, S. & Tufik, S. (2004). Afferent pain pathways: A neuroanatomical review. *Brain Research*, *1000*(1–2), 40–56.

- Al-Moitary, M. O. (2011). Nursing student's knowledge and application of pain assessment at King Saud University. Saudi Arabia: King Saud University Nursing College.
- Almutairi, A. F. & McCarthy, A. (2012). A multicultural nursing workforce and cultural perspectives in Saudi Arabia: An overview. *The Health*, *3*(3), 71–74.

Al-Osimy, M. (1994). Nursing in Saudi Arabia. Saudi Arabia: King Fahd National Library.

- Al-Shaer, D., Hill, P. & Anderson, M. (2011). Nurses' knowledge and attitudes regarding pain assessment and intervention. *MEDSURG Nursing*, 20(1), 7–11.
- Al-Shahri, M. Z. (2002). Culturally sensitive caring for Saudi patients. *Journal of Transcultural Nursing*, *13*(2), 133–138.
- Ambuel, B., Hamlett, K. W., Marx, C. M. & Blumer, J. L. (1992). Assessing distress in pediatric intensive care environments: The COMFORT scale. *Journal of Pediatric Psychology*, 17(1), 95–109.
- American Geriatric Society (AGS). (2002). Panel on persistent pain in older persons: The management of persistent pain in older persons. *Journal of American Geriatric Society*, 50(6), 205–224.
- American Pain Foundation. (2009). *Pain resource guide: Getting the help you need.* Retrieved 18 July 2013 from http://www.painfoundation.org/learn/ publications/files/PainResourceGuide2009.pdf
- Anderson, K. O., Richman, S. P., Hurley, J., Palos, G., Valero, V., Mendoza, T., . . . Cleeland,
 C. S. (2002). Cancer pain management among minority outpatients: Perceived needs and barriers to optimal control. *Cancer*, 94(8), 2295–2304.
- Apfelbaum, J. L., Chen, C., Mehta, S. & Gan, T. J. (2003). Postoperative pain experience: Results from a national survey suggesting postoperative pain continues to be undermanaged. *Anesthesia and Analgesia*, 97, 534–540.

- Arbour, C. & Gelinas, C. (2010). Are vital signs valid indicators for the assessment of pain in postoperative cardiac surgery ICU adults? *Intensive & Critical Care Nursing*, 26(2), 86–90.
- Argoff, C. E. & Fine, P. G. (2010). Application of multimodal therapy for acute pain: An interactive clinical case challenge. Retrieved 5 September 2011 from www.medscapeCME.com/nurses
- Arnstein, P. & Marie, B. (2010). Managing chronic pain with opioids: A call for change. *Nurse Practitioner Healthcare Foundation*, 1–14. Retrieved 22 August 2012 from http://www.nphealthcarefoundation.org/programs/downloads/white_ paper_opioids.pdf
- Attir, M. O. (1985). Ideology, value changes, and women's social position in Libyan society.
 In E. W. Fernea (Ed.), *Women and the family in the Middle East; New voices of change* (pp. 121-133). Austin: University of Texas Press.
- Baguley, T. (2004). Understanding statistical power in the context of applied research. *Applied Ergonomics*, *35*(2), 73–80.
- Baidawi, S., Turner, S., Trotter, C., Browning, C., Collier, P., O'Connor, D. & Sheehan, R.
 (2011). Older prisoners a challenge for Australian corrections. *Trends & Issues in Crime and Criminal Justice, 426*, 421–440.
- Bair, M. J., Kroenke, K., Sutherland, J. M., McCoy, K. D., Harris, H. & McHorney, C. A.
 (2007). Effects of depression and pain severity on satisfaction in medical outpatients: Analysis of the Medical Outcomes Study. *Journal of Rehabilitation Research & Development*, 44(2), 143–152.
- Baldridge, K. H. & Andrasik, F. (2010). Pain assessment in people with intellectual or developmental disabilities. *American Journal of Nursing*, 110(12), 28–35.
- Ballantyne, J. C. (2003). Chronic pain following treatment for cancer: The role of opioids. *The Oncologist, 8*(6), 567–575.
- Ballantyne, J. C. (2006). The Massachusetts General Hospital handbook of pain management (3rd ed.). Philadelphia, PA: Lippincott Williams & Wilkins.
- Benhamou, D., Berti, M., Brodner, G., De Andres, J., Draisci, G., Moreno-Azcoita, M., . . .
 Viel, V. (2008). Postoperative analgesic therapy observational survey (PATHOS): A practice pattern study in 7 central/southern European countries. *Pain, 136*(1–2), 134–141.
- Bernardi, M., Catania, G. & Tridello, G. (2007). Knowledge and attitudes about cancer pain management. A national survey of Italian hospice nurses. *Cancer Nursing*, 30(2), 20– 26.
- Berry, P. H. & Dahl, J. L. (2000). The new JCAHO pain standards: Implications for pain management nurses. *Pain Management Nursing*, *1*(1), 3–12.
- Berry, P. H., Covington, E., Dahl, J., Katz, J. & Miaskowski, C. (2006). Pain: Current understanding of assessment, management, and treatments. Reston, VA: National Pharmaceutical Council, Inc.

Blaxter, M. (2010). Health (2nd ed.). Cambridge, UK: Polity Press.

- Bouvette, M., Fothergill-Bourbonnais, F. & Perreault, A. (2002). Implementation of the pain and symptom assessment record (PSAR). *Journal of Advanced Nursing*, *40*(6), 685– 700.
- Bowsher, D. (1999). The lifetime occurrence of Herpes zoster and prevalence of post-herpetic neuralgia: A retrospective survey in an elderly population. *European Journal of Pain*, 3(4), 335–342.
- Brawley, O. W., Smith, D. E. & Kirch, R. A. (2009). Taking action to ease suffering: Advancing cancer pain control as a health care priority. *CA Cancer Journal for Clinicians*, 59(5), 285–289.

- Breivik, H., Borchgrevink, P. C., Allen, S. M., Rosseland, L. A., Romundstad, L., Hals, E. K.
 B., . . . Stubhaug, A. (2008). Assessment of pain. *British Journal of Anaesthesia*, *101*(1), 17–24.
- Breivik, H., Cherny, N., Collett, B., de Conno, F., Filbet, M., Foubert, A. J., ... Dow, L. (2009). Cancer-related pain: A pan-European survey of prevalence, treatment, and patient attitudes. *Annals of Oncology*, 20(8), 1420–1433.
- Breivik, H., Collett, B., Ventafridda, V., Cohen, R. & Gallacher, D. (2006). Survey of chronic pain in Europe: Prevalence, impact on daily life, and treatment. *European Journal of Pain*, 10(4), 287–333.
- Brennan, F., Carr, D. B. & Cousins, M. (2007). Pain management: A fundamental human right. *Anesthesia and Analgesia*, *105*(1), 205–221.
- Brinkmann, S. & Kvale, S. (2005). Confronting the ethics of qualitative research. *Journal of Constructivist Psychology*, 18(2), 157–181.
- Brinkus, R. & Narayan, M. C. (2002). Communication as the fabric of community-based nursing practice. In J. M. Sorrell & G. M. Redmond (Eds.), *Community-based nursing practice: Learning through students' stories*. Philadelphia, PA: F. A. Davis.
- Brown, A., Bowman, J. & Eason, R. (1999). Assessment of nurses' attitudes and knowledge regarding pain management. *Journal of Continuing Education in Nursing*, *30*(3), 132–139.
- Brown, L. D. (2008). In-season prediction of batting averages: A field test of empirical Bayes and Bayes methodologies. *Annals of Applied Statistics*, *2*(1), 113–152.
- Bruera, E., Willey, J. S., Ewert-Flannagan, P. A., Cline, M. K., Kaur, G., Shen, L., . . .
 Palmer, J. L. (2005). Pain intensity assessment by bedside nurses and palliative care consultants: A retrospective study. *Support Care Cancer*, *13*(4), 228–231.

Brunier, G., Carson, M.G., & Harrison, D.E. (1995). What do nurses know and believe about patients with pain? Results of a hospital survey. *Journal of Pain and Symptom Management*, 10(6), 436-445.

Bryman, A (2001) Social Research Methods, Oxford: Oxford University Press.

- Buxton, B. P. (1999). The physiology and psychology of pain. In C. Starkey (Ed.), *Therapeutic modalities for athletic therapists* (2nd ed., pp. 36–67). Philadelphia, PA:
 F. A. Davis.
- Cairns, M., Thompson, M. & Wainwright, W. (2003). *Transitions in dying and bereavement:* A psychosocial guide for hospice and palliative care. Winnipeg, Canada: Health Professions Press.
- Callister, L. C. (2003). Cultural influences on pain perceptions and behaviors. *Home Health Care Management & Practice*, *15*(3), 207–211.
- Campbell, L. C., Andrews, N., Scipio, C., Flores, B., Feliu, M. H. & Keefe, F. J. (2009). Pain coping in Latino populations. *Journal of Pain*, 10(10), 1012–1019.
- Canadian Nurses Association. (2011). Standards and best practices. Ottawa, Canada.
- Castillo, J.J. (2009) Population Sampling Techniques. Retrieved: May 18, 2012 from http://www.experiment-resources.com/population-sampling.html
- Catalano, E. M. & Hardin, K. N. (2004). *Dores crônicas*. São Paulo, Brazil: Summus Editorial.
- Central Intelligence Agency. (2013). *The world factbook*. Retrieved 28 November 2013 fromhttps://www.cia.gov/library/publications/the-world-factbook/geos/sa.html
- Chang, S. O., Kim, H. S., Sjöström, B. & Schwartz-Barcott, D. (2001). Ways of assessing post-operative pain. *Journal of Korean Academy of Nursing*, *31*(7), 1189–1196.
- Chaudhari, S. & Feaver, S. (2011). Organization of pain management services. *Anaesthesia & Intensive Care Medicine*, *12*(2), 37–38.

- Chen, L. M., Miaskowski, C., Dodd, M. & Pantilat, S. (2008). Concepts within the Chinese culture that influence the cancer pain experience. *Cancer Nursing*, *31*(2), 103–108.
- Chen, T., Gelgor, L. & Bajorek, B. (2004). Implementation and evaluation of a multidisciplinary chronic pain management service in rural NSW. Sydney, NSW: Faculty of Pharmacy, University of Sydney.
- Cherniack, E. P., Ceron-Fuentes, J., Florez, H., Sandals, L., Rodriguez, O. & Palacios, J. C. (2008). Influence of race and ethnicity on alternative medicine as a self-treatment preference for common medical conditions in a population of multiethnic urban elderly. *Complementary Therapies in Clinical Practice, 14*(2), 116–123.
- Christo, P. J. & Mazloomdoost, D. (2008). Cancer pain and analgesia. *Annals of the New York Academy of Sciences, 1138*, 278–298.
- Cicero, T. J., Wong, G., Tian, Y., Lynskey, M., Todorov, A. & Isenberg, K. (2009). A response to McQuay and Moore editorial 'Opioids in non-cancer pain—A critical read of Cicero et al.'. *Pain, 146*(3), 325–326.
- Clarke, K. A. & Iphofen, R. (2008). The effects of failing to believe patients' experience of chronic pain. *Nursing Times*, *104*(8), 30–31.
- Cohen, L., Manion, L. & Morrison, K. (2013). Research methods in education. Taylor & Francis. Retrieved 18 October 2013 fromhttp://books.google.com.au/books?id=mLh0Oza3V1IC
- Cohen, R. A. & Van Nostrand, J. F. (1995). Trends in the health of older Americans: United States. *Vital Health Stat*, *3*(30), 1–328.
- Coker, E., Papaioannou, A., Kaasalainen, S., Dolovich, L., Turpie, I. & Taniguchi, A. (2010). Nurses' perceived barriers to optimal pain management in older adults on acute medical units. *Applied Nursing Research*, 23(3), 139–146.

Coon, D. & Mitterer, J. O. (2008). Psychology: A journey (3rd ed.). US: Cengage Learning.

- Cordell, W., Keene, K., Giles, B., Jones, J., Jones, J. & Brizendine, E. J. (2002). The high prevalence of pain in emergency medical care. *American Journal of Emergency Medicine*, 20(3), 165–169.
- Coulling, S. (2005) Nurses' and doctors' knowledge of pain after surgery. *Nursing Standard, 19*(34), 41–49.
- Cousins, M. J., Brennan, F. & Carr, D. B. (2004). Pain relief: A universal human right. *Pain*,112(1–2), 1–4.
- Cousins, M.J., Bridenbaugh, P.O., Carr, D.B., & Horlocker, T.T. (2009). *Cousins & Bridenbaugh's Neural Blockade In Clinical Anesthesia and Management of Pain*, 4th ed. Philadelphia: Lippincott Williams & Wilkins.
- Craig, K. D., Lilley, C. M. & Gilbert, C. A. (1996). Social barriers to optimal pain management in infants and children. *Clinical Journal of Pain*, 12(3), 232–242.
- Creswell, J. W. (2003). *Qualitative, quantitative, and mixed methods approaches* (2nd ed.). Thousand Oaks, CA: Sage.
- Creswell, J. (2008). *Educational research: Planning, conducting, and evaluating quantitative and qualitative research*. New Jersey: Pearson: Merrill Prentice Hall.
- Creswell, J. W. (2009). Mapping the field of mixed methods research. *Journal of Mixed Methods Research, 3,* 95–108.
- Creswell, J. W., & Plano Clark, V. L. (2007). *Designing and conducting mixed methods research*. Thousand Oaks, CA: Sage.
- Creswell, J. W., & Plano Clark, V. L. (2011). *Designing and conducting mixed methods research*. (2nd ed.). Thousand Oaks, CA: Sage.
- Curry, M. N. (2010). Culture's effects on pain assessment and management. *American Journal of Nursing*, *110*(4), 38–47.
- D'Arcy, Y. M. (2007). Pain management: Evidence-based tools and techniques for nursing professionals. Tucson, AZ: Lawyers and Judges Publishing Company.

Daniels, R. & Nicoll, L. H. (2011). Contemporary medical-surgical nursing (Vol. 1, 2nd ed.). Delmar, Cengage Learning.

David, M. & Sutton C. D. (2004). Social research: The basics. London: Sage Publications.

- Davidhizar, R. & Bartlett, D. (2000) Helping your patients control their pain. *The ABNF Journal*, 11(1), 13–17.
- Davidhizar, R. & Giger, J. (2004). A review of the literature on care of clients in pain who are culturally diverse. *International Nursing Review*, *51*(1), 47–55.
- Davitz, L. L. & Davitz, J. R. (1985). Culture and nurses' inference of suffering. In LA Copp (Ed.). *Perspectives on pain*. New York, NY: Churchill Livingstone.
- Daymon, C. & Holloway, I. (2010). *Qualitative research methods in public relations and marketing communications*. UK: Taylor & Francis.
- Defrin, R., Eli, I., & Dorit, P. (2011). Interactions among sex, ethnicity, religion and gender role expectations of pain. *Gender Medicine*, 8(3), 172 183.
- Denzin, N. K., & Lincoln, Y. S. (2011). *The SAGE Handbook of Qualitative Research*. SAGE Publications.
- Disorbio, M., Bruns, D. & Barolat, G. (2006). Assessment and treatment of chronic pain: A physician's guide to a biopsychosocial approach. *Practical Pain Management, 6*, 1–10.
- Divi, C., Koss, R. G., Schmaltz, S. P. & Loeb, J. M. (2007). Language proficiency and adverse events in US hospitals: A pilot study. *International Journal of Health Care Quality*, 19(2), 60–67.
- Dix, P., Sandhar, B., Murdoch, J. & MacIntrye, P. (2004). Pain on medical wards in a district general hospital. *British Journal of Anesthesia*, *92*(2), 235–237.
- Dolin, S. J., Cashman, J. N. & Bland, J. M. (2002). Effectiveness of acute postoperative pain management: I. Evidence from published data. *British Journal of Anaesthesia*, 89(3), 409–423.

- Drayer, R. A., Henderson, J. & Reidenberg, M. (1999). Barriers to better pain control in hospitalized patients. *Journal of Pain and Symptom Management*, 17(6), 434–440.
- Duignan, M. & Dunn, V. (2009) Perceived Barriers to Pain Management. *Emergency Nurse*. *16*(9): 31-35.
- Duke, G., Haas, B. K., Yarbrough, S. & Northam, S. (2013). Pain management knowledge and attitudes of baccalaureate nursing students and faculty. *Pain Management Nursing*, 14(1), 11–19.
- Dunajcik, L. (1999). Chronic nonmalignant pain. In M. McCaffery & C. Pasero, Pain clinical manual (2nd ed., pp. 467–521). St Louis, MO: Mosby.
- Edwards, H. E., Nash, R. E., Najman, J. M., Yates, P. M., Walsh, A. M., McDowell, J. K., . . . Dewar, A. (2001). Determinants of nurses' intention to administer opioids for pain relief. *Nursing and Health Sciences*, *3*(3), 149–159.
- Edwards, R. R., Wasan, A. D., Bingham, C. O., Bathon, J., Haythornthwaite, J. A., Smith, M. T. & Page, G. G. (2009). Enhanced reactivity to pain in patients with rheumatoid arthritis. *Arthritis Research & Therapy*, *11*, 1–9.
- Eid, T. & Bucknall, T. (2008). Documenting and implementing evidence-based post-operative pain management in older patients with hip fractures. *Journal of Orthopaedic Nursing*, *12*(2), 90–98.
- El Sissi, W., Arnaout, A., Chaarani, M. W., Fouad, M., El Assuity, W., Zalzala, M., . .
 .Youseif, E. (2010). Prevalence of neuropathic pain among patients with chronic lowback pain in the Arabian Gulf Region assessed using the leeds assessment of neuropathic symptoms and signs pain scale. *Journal of International Medical Research*, 38(6), 2135–2145.
- El-Gilany, A. & Al-Wehady, A. (2001). Job satisfaction of female Saudi nurses. *Eastern Mediterranean Health Journal*, 7(1–2), 31–37.

Elazia, P. M. (2012). Culture and society (Term Paper). Kilifi, Kenya: Pwani University.

- Elcigil, A., Maltepe, H., Esrefgil, G. & Mutafoglu, K. (2011). Nurses' perceived barriers to assessment and management of pain in a university hospital. *Journal of Pediatric Hematology/Oncology*, 33(1), 33–38.
- Elliott, A.M., Smith, B., H., Penny, K.I., Smith, W., C., & Chambers, W.A. (1999). The epidemiology of chronic pain in the community. *Lancet*, *354*, 1248–52.
- Elwell, F. W. (2000). Verstehen: Max Weber's HomePage. Retrieved 11 December 2012 from www.faculty.rsu.edu/users/f/felwell/www/Theorists/Whome.htm
- Erdek, M. A. & Pronovost, P. J. (2004). Improving assessment and treatment of pain in the critically ill. *International Journal for Quality in Health Care, 16*(1), 59–64.
- Erkes, E. B., Parker, V. G., Carr, R. L. & Mayo, R. M. (2001). An examination of critical care nurses' knowledge and attitudes regarding pain management in hospitalized patients. *Pain Management Nursing*, 2(2), 47–53.
- Farrell, C. S. (2010). Chronic pain management in a reservation border town. Tucson, AZ: University of Arizona. Retrieved 13 August 2011 from

http://www.nursing.arizona.edu/Library/Farrell_Cynthia_Practice_Inquiry.pdf

Fein, A. (2012). Nociceptors and The Perception Of Pain. Cell.uchc.edu, (February).

Retrieved from http://cell.uchc.edu/pdf/fein/nociceptors_fein_2012.pdf

- Fenwick, C. (1998). Post operative pain experiences in Central Australian Aboriginal women:What do we understand? *B.Nursing (hons) thesis*. Lismore, NSW: Southern CrossUniversity.
- Fenwick, C. (2006). Assessing pain across the cultural gap: Central Australian Indigenous peoples' pain assessment. *Contemporary Nurse*, 22(2), 218–227.
- Ferguson, C. J. (2009). An effect size primer: A guide for clinicians and researchers. *Professional Psychology: Research and Practice*, 40(5), 532–538.
- Ferrell, B. & McCaffery, M. (2008). Knowledge and attitudes survey regarding pain. Retrieved 29 April 2011 from http://prc.coh.org

Field, A. (2009). Discovering statistics using SPSS. London: Sage Publications.

- Finley, G. A., Kristjansdottir, O. & Forgeron, P. A. (2009). Cultural influences on the assessment of children's pain. *Pain Research & Management*, 14(1), 33–37. Retrieved 28 January 2013 from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2706562/
- Flowers, D. L. (2004). Culturally competent nursing care: A challenge for the 21st century. *Critical Care Nurse*, *24*(4), 48–52.

Foley, K. (2011). How well is cancer pain treated? Palliative Medicine, 25(6), 398-401.

- Fosnocht, D. E., Swanson, E. R. & Barton, E. D. (2005). Changing attitudes about pain and pain control in emergency medicine. *Emergency Medicine Clinics of North America*, 23(2), 297–306.
- Foster, N. E., Bishop, A., Thomas, E., Main, C., Horne, R., Weinman, J. & Hay, E. (2008).
 Illness perceptions of low back pain patients in primary care: What are they, do they change and are they associated with outcome? *Pain*, *136*(1–2), 177–187.
- Furrow, B.R. (2001). Pain management and provider liability: no more excuses. Journal of Law, Medicine & Ethics, 29, 28-51.
- Gallagher, E. B. (2002). Modernization and health reform in Saudi Arabia. In A. C. Twaddle (Ed.). *Health care reform around the world* (pp. 181–197). London, England: Auburn House.
- Galluzzi, K. E. (2007). Managing neuropathic pain. *Journal of the American Osteopathic Association, 107*(10), 39–48.
- Ganzberg, S. (2010). Pain management part II: Pharmacologic management of chronic orofacial pain. *Anesthesia Progress*, *57*(3), 114–118.
- Gatchel, R. J. & Okifuji, A. (2006). Evidence-based scientific data documenting the treatment and cost-effectiveness of comprehensive pain programs for chronic nonmalignant pain. *Journal of Pain 7*(11), 779–793.

- Gerwin, R. (2002). Myofascial and Visceral Pain Syndromes: Visceral-Somatic Pain Representations. *Journal Of Musculoskeletal Pain*, *10*(1), 165–175.
- Gilman, S. & Newman, S. (1992). Manter and Gatz's essentials of clinical neuroanatomy and neurophysiology (8th ed.). Philadelphia. PA: F. A. Davis.
- Goldstein, D. J., Lu, Y., Detke, M. J., Hudson, J., Iyengar, S., & Demitrack, M. A. (2004).
 Effects of duloxetine on painful physical symptoms associated with depression.
 Psychosomatics, 45, 17–28.
- Gordon, D. (1999). Pain management in the elderly. *Journal of Perianaesthesia Nursing* 14(6), 367–372.
- Gordon, D. B., Dahl, J., Miaskowski, C., McCarberg, B., Todd, K. H., Paice, J. A., ... Carr, D. B. (2005). American pain society recommendations for improving the quality of acute and cancer pain management. *Archives of Internal Medicine*, *165*(14), 1574–1580.
- Gordon, D. B., Pellino, T., Miaskowski, C., McNeill, J. A., Paice, J. A., Laferriere, D. &
 Bookbinder, M. (2002). A 10-year review of quality improvement monitoring in pain management: Recommendations for standardized outcomes measures. *Pain Management Nursing*, 3(4), 116–130.
- Gordon, D. B., Rees, S. M., McCausland, M. R., Pellino, T. A., Sanford-Ring, S., Smith-Helmenstine, J. & Danis, D. M. (2008). Improving reassessment and documentation of pain management. *Joint Commission Journal on Quality and Patient Safety/Joint Commission Resources*, 34(9), 509–517.

Gordon, D. B., Stevenson, K. K., Griffie, J., Muchka, S., Rapp, C. & Ford-Roberts, K. (1999). Opioid equianalgesic calculations. *Journal of Palliative Medicine*, *2*(2), 209–218.

Gray, D. E. (2004). Doing research in the real world. London: Sage Publications.

- Greene, J. C., & Caracelli. (2003). Making paradigmatic sense of mixed methods practice. In Tashakkori, A., & C. Teddlie (Eds.), *Handbook of Mixed Methods in Social and Beha vioral Research*. Thousand Oaks, CA: Sage Publications.
- Gregory, J. & Haigh, C. (2008). Multidisciplinary interpretations of pain in older patients on medical units. *Nurse Education in Practice*, *8*(4), 249–257.
- Gunningberg, L. & Idvall, E. (2007). The quality of postoperative pain management from the perspectives of patients, nurses and patient records. *Journal of Nursing Management*, 15(7), 756–766.
- Habiba, M. A. (2000). Examining consent within the patient–doctor relationship. *Journal of Medical Ethics*, 26(3), 183–187.
- Hader, C. & Guy, J. (2004). Your hand in pain management. *Nursing Management, 35*(11), 21–27.
- Hampton T. (2005). When shingles wanes but pain does not. *Journal of American Medical* Association, 293, 2459–2460.
- Hanna MN, González-Fernández, M., Barrett, A.D., Williams, K.A. & Pronovost, P. (2012).
- Does patient perception of pain control affect patient satisfaction across surgical units in a tertiary teaching hospital? *American Journal of Medical Quality*, 27(5), 411–416.
- Hasselstrom, J., Liu-Palmgren, J. & Rasjo-Wraak, G. (2002). Prevalence of pain in general practice. *European Journal of Pain, 6*(5), 375–385.
- Haverkamp, B. E. (2005). Ethical perspectives on qualitative research in applied psychology. *Journal of Counseling Psychology*, *52*(2), 146–155.
- Helmrich, S., Yates, P., Nash, R., Hobman, A., Poulton, V. & Berggren, L. (2001). Factors influencing nurses' decisions to use non-pharmacological therapies to manage patients' pain. *Australian Journal of Advanced Nursing*, 19(1), 27–35.

- Henson, R.K. (2001). Understanding internal consistency reliability estimates: a conceptual primers on coefficient alpha. *Measurement and Evaluation in Counseling and Development, 34*, 177-188.
- Heron, J. (1996). *Co-operative inquiry: Research into the human condition*. Thousand Oaks, CA: Sage Publications.
- Hill, C. R. & Thompson, B. (2004). Computing and interpreting effect sizes. In J. C. Smart (Ed.), *Higher education: Handbook of theory and research* (pp. 175–196). New York, NY: Kluwer.
- Hooke, M. C. (2007). Social, cultural, and religious influences on child health promotion. In
 M. J. Hockenberry & D. Wilson (Eds.), *Wong's nursing care of infants and children* (8th ed., pp. 28–54). St Louis, MO: Mosby.
- Horgas, A. & Yoon, S. (2008). Pain management. In E. Capezuti, D. Zwicker, M. Mezey & T. Fulmer (Eds.), Evidence-based geriatric nursing protocols for best practice. New York, NY: Springer.
- Howell, D., Butler, L., Vincent, L., Watt-Watson, J. & Stearns, N. (2000). Influencing nurses' knowledge, attitudes and practice in cancer pain management. *Cancer Nursing*, 23(1), 55–63.
- Hoyle, R. H., Harris, M. J. & Judd, C. M. (2002). *Research methods in social relations*.London, England: Thomson Learning, Inc.
- Huang, X. L. Ma, S. L. Zhang, J. F. Zhang, J. P. & Lu, Y. H. (2001). An investigation and analysis on how much knowledge of pain treatment nurses master from 27 provinces and cities. *Chinese Journal of Pain Medicine*, 7(1), 22–25.
- Huffman, J. C. & Kunik, M. E. (2000). Assessment and understanding of pain in patients with dementia. *The Gerontologist*, *40*(5), 574–581.
- Huth, M. M, Gregg, T. L. & Lin, L. (2010). Education changes Mexican nurses' knowledge and attitudes regarding pediatric pain. *Pain Management Nursing*, *11*(4), 201–208.

- Igumbor, E. U., Puoane, T., Gansky, S. & Plesh, O. (2011). Prevalence and risk indicators of chronic pain in a rural community in South Africa. *Journal for Rural and Tropical Public Health*, 10, 61–69.
- Innis, J., Bikaunieks, N., Petryshen, P., Zellermeyer, V. & Ciccarelli, L. (2004). Patient satisfaction and pain management: An educational approach. *Journal of Nursing Care Quality*, 19(4), 322–327.
- International Association for the Study of Pain & European Federation of IASP Chapters. (2005). Factsheet 4A: Unrelieved pain is a major global healthcare problem. Retrieved 9 November 2013 from http://www.iasppain.org/AM/Template.cfm?Section=Press_ Release&Template=/CM/ContentDis- play.cfm&ContentID=2908
- International Association for the Study of Pain. (2013). Culture of pain. *Pain Clinical Updates*. Retrieved 23 August 2013 from http://www.iasp-pain.org/AM/Template.cfm?Section=Clinical Updates
- Jackson, C. L. & Gary, R. (1991). Nursing: Attitudes, perceptions and strategies for progress in Saudi Arabia. *Annals of Saudi Medicine*, *11*(4), 452–458.
- Jacob, E., Sambuco, G., McCarthy, K. & Hockenberry, M. (2008). Intensity, location, and quality of pain in Spanish-speaking children with cancer. *Pediatric Nursing*, 34(1), 45–52.
- Jamison, R. N., Serraillier, J. & Michna, E. (2011). Assessment and treatment of abuse risk in opioid prescribing for chronic pain. *Pain Research and Treatment, 2011*, 1–12.
- Jänig, W. (2012). Autonomic reactions in pain. *Pain, 153*(4), 733–735. Retrieved 1 September 2013 from http://linkinghub.elsevier.com/retrieve/pii/ S0304395912000577?showall=true
- Janmohamed, S., & College, D. (2009). *Pain Management in Long-term Care Patients*. D'Youville College.

- Janssen, S. A. (2002). Negative affect and sensitization to pain. *Scandinavian Journal of Psychology*, 43(2), 131–137.
- Johnson, R. B., & Onwuegbuzie, A. J. (2004). Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, *33*(7), 14-26.
- Joint Commission on Accreditation of Healthcare Organizations (JCAHO). (2001). JCAHO pain standards for 2001. Retrieved 1 June 2011 from http://www.jcaho.org
- Joint Commission Resources. (2010). Approaches to pain management: An essential guide for clinical leaders. Joint Commission Resources. Retrieved from http://store.jcrinc.com/assets/1/14/APM10 Sample Pages2.pdf
- Jones, K. R., Fink, R., Pepper, G., Hutt, E., Vojir, C. P., Scott, J., Clark, L., & Mellis, K. (2004). Improving nursing home staff knowledge and attitudes about pain. *The Gerontologist*, *44*(4), 469-478.
- Jones, T., Brownlee, S. & Cantor, J. (2002). Impact of health attitudes on health seeking behavior. Rutgers Center for State Health Policy. Retrieved 6 February 2011 from http://www.cshp.rutgers.edu/Downloads/6080.pdf
- Joshi, G. P. & Ogunnaike, B. O. (2005). Consequences of inadequate postoperative pain relief and chronic persistent postoperative pain. *Anesthesiology Clinics of North America*, 23(1), 21–36.
- Jovey, R. (2008). Pain pathways and pathophysiology. In R. Jovey (Ed.), *Managing pain: The Canadian healthcare professional's reference* (pp. 9–18). Ontario, Canada: Baker Edwards Consulting Inc.
- Juarex, G. (1997). Culture and pain. Quality of Life, 4(4), 86-69.
- Kaki, A. M. (2011). Medical students' knowledge and attitude toward cancer pain management in Saudi Arabia. *Saudi Medical Journal*, 32(6), 628–632.

- Kaki, A. M., Daghistani, M. F. & Msabeh, A. A. (2009). Nurses' knowledge of pharmacological measures on acute pain management in Western Saudi Arabia. *Saudi Medical Journal*, 30(2), 279–283.
- Katsma, D.L., Souza, C.H., 2000. Elderly pain assessment and pain management knowledge of long-term care nurses. *Pain Management Nursing 1* (3), 88–95.
- Klauenberg, S., Maier, C., Assion, H., Hoffmann, A., Krumova, E. K., . . . Juckel, G. (2008).
 Depression and changed pain perception: Hints for a central disinhibition mechanism.
 Pain, 140(2), 332–343.
- Kotrlik, J. W. & Williams, H. A. (2003). The incorporation of effect size in information technology, learning, and performance research. *Information Technology, Learning* and Performance Journal, 21(1), 1–7.
- Kvale, D. (1996). Interviews. London: Sage Publications.
- Lai, Y. H., Chen, M. L., Tsai, L. Y., Lo, L. H., Wei, L. L., Hong, M. Y., . . . Guo, S. L. (2003). Are nurses prepared to manage cancer pain? A national survey of nurses' knowledge about pain control in Taiwan. *Journal of Pain and Symptom Management*, 26(5), 1016–1025.
- Lasch, K. E. (2000). Culture, pain, and culturally sensitive pain care. *Pain Management Nursing, 1*(Suppl. 1), 16–22.
- Lasch, K. E. (2002). Culture and pain. Pain Clinical Updates, 10(5), 1-4.
- Layzell, M. (2008). Current interventions and approaches to postoperative pain management. *British Journal of Nursing*, 17(7), 414–419.
- Leininger, M. (1995). *Transcultural nursing: Concepts, theories, and practice* (2nd ed.). Columbus, OH: McGraw-Hill College Custom Series.
- Leininger, M. (1996). Culture care theory, research and practice. *Nursing Science Quarterly*, 9(2), 71–78.

- Leininger, M. (2001). *Cultural care and universality: A theory of Nursing*. Sudbury, MA: Jones and Bartlett Publishing.
- Leininger, M. M. (1991). *Culture care diversity and universality: A theory of nursing*. New York, NY: National League for Nursing.
- Leininger, M. M. (2002). Essential transcultural nursing care concepts, principles, examples, and policy statements. In M. M. Leininger & M. R. McFarland (Eds.), *Transcultural nursing: Concepts, theories, research and practice* (3rd ed., pp. 45–70). New York, NY: McGraw Hill.
- Leininger, M. M. & McFarland, M. R. (2002). *Transcultural nursing: Concepts, theories, research, and practice* (3rd ed.). New York, NY: McGraw-Hill.
- Lewis, S., Heitkemper, M. & Dirksen, S. (2004). *Medical surgical nursing; Assessment and management of clinical problems* (6th ed.). St Louis, MO: Mosby.
- Lewthwaite, B. J., Jabusch, K. M., Wheeler, B. J., Schnell-Hoehn, K. N., Mills, J., Estrella-Holder, E. & Fedorowicz, A. (2011). Nurses' knowledge and attitudes regarding pain management in hospitalized adults. *Journal of Continuing Education in Nursing*, 42(6), 251–258.
- Liu, L., So, W. & Fong, D. (2008). Knowledge and attitudes regarding pain management among nurses in Hong Kong medical units. *Journal of Clinical Nursing*, 17, 2014– 2021.
- Long, T. & Johnson, M. (2000). Rigour, reliability and validity in qualitative research. *Clinical Effectiveness in Nursing 4*, 30–37.
- Lorenz, K. A., Sherbourne, C. D., Shugarman, L. R., Rubenstein, L. V., Wen, L., Cohen, A., .
 . . Asch, S. M. (2009). How reliable is pain as the fifth vital sign? *Journal of the American Board of Family Medicine*, 22(3), 291–298.
- Lovering, S. (2006). Cultural attitudes and beliefs about pain. *Journal of transcultural nursing*, *17*(4), 389–395.

Low, M. C. (1985). The experience of symptoms. London, England: Lexington Books.

- Ludwig-Beymer, P. (2008). Transcultural aspects of pain. In M. M. Andrews, J. S. Boyle (Eds.), *Transcultural concepts in nursing care* (5th ed.). Philadelphia, PA: Wolters Kluwer Health/Lippincott, Williams and Wilkins.
- Lui, L. Y., So, W. K. & Fong, D. Y. (2008). Knowledge and attitudes regarding pain management among nurses in Hong Kong medical units. *Journal of Clinical Nursing*, 17(15), 2014–2021.
- Macintyre, P. E. (2001). *Acute pain management: A practical guide*. London, England: Saunders.
- Macintyre, P. E. & Schug, S. A. (2007). *Acute pain management: A practical guide* (3rd ed.). Edinburgh, Scotland: Saunders Elsevier.
- Macintyre, P. E., Schug, S. A., Scott, D. A., Visser, E. J., Walker, S. M. (2010). Acute pain management: Scientific evidence (3rd ed.). Melbourne, Vic: Australian and New Zealand College of Anaesthetists and Faculty of Pain Medicine.
- Macintyre, P., Schug, S., Scott, S., Visser, E. & Walker, S. (2010). *Acute pain management* (3rd ed.). Melbourne, Vic: Australian and New Zealand College of Anaesthetists and Faculty of Pain Medicine (ANZCA & FPM).
- MacLaren, J. E., & Cohen, L. L. (2005). A comparison of distraction strategies for venipuncture distress in children. *Journal of Pediatric Psychology*, 30, 387–396.
- Macrae, W. A. (2008). Chronic post-surgical pain: 10 years on. *British Journal of Anaesthesia, 101*(1), 77–86.
- Magnusson, J. E. & Fennell, J. A. (2011). Understanding the role of culture in pain: Māori practitioner perspectives relating to the experience of pain. *New Zealand Medical Journal*, *124*(1328), 41–51.
- Maio, R., Garrison, H., Spaite, D., Desmond, J., Gregor, M., Stiell, I., . . . O'Malley, P.(2002). Emergency medical services outcomes project IV (EMSOP IV): Pain

measurement in out-of-hospital outcomes research. *Annals of Emergency Medicine*, 40(2), 172–179.

- Majidyar, A. (2013). Saudi Arabia's forgotten Shi'ite spring. American Enterprise Institute (AEI) Middle Eastern Outlook, (5). Retrieved 19 September 2013 from http://www.aei.org/files/2013/08/16/-saudi-arabias-forgotten-shiitespring_15503177253.pdf
- Malterud, K. (2001). Qualitative research: Standards, challenges and guidelines. *Lancet, 358*(9280), 483–488.
- Mandeville, A. L. (2010). Non-pharmacological methods of acute pain management. In L. Bromley & B. Brandner (Eds.), *Acute pain* (p. 19). United Kingdom: Oxford University Press.
- Manias, E. (2003). Medication trends and documentation of pain management following surgery. *Nursing and Health Sciences*, *5*(1), 85–94.
- Manias, E., Bucknall, T. & Botti, M. (2005). Nurses' strategies for managing pain in the postoperative setting. *Pain Management Nursing*, 6(1), 18–29.
- Mäntyselkä, P., Kumpusalo, E., Ahonen, R., Kumpusalo, A., Kauhanen, J., Viinamäki, H., . . . Takala, J. (2001). Pain as a reason to visit the doctor: A study in Finnish primary health care. *Pain*, *89*(2–3), 175–180.
- Marks, R. M. & Sachar, E. J. (1973). Undertreatment of medical inpatients with narcotic analgesics. *Annals of Internal Medicine*, 78(2), 173–181.
- Marrone, S. R. (1999). Designing a competency-based nursing practice model in a multicultural setting. *Journal for Nurses in Staff Development*, 15(2), 56–62.
- Matthews, E. & Malcolm, C. (2007). Nurses' knowledge and attitudes in pain management practice. *British Journal of Nursing*, *16*(3), 174–179.

- Mazzilli, C. & Davis, A. (2007). *Health care seeking behaviour in Somalia* (No. 10) (p. 52). Retrieved 28 January 2012 from http://www.unicef.org/somalia/SOM_ HealthcareseekingbehaviourReport 10-WEB.pdf
- McCaffery, M. (2002). What is the role of nondrug methods in the nursing care of patients with acute pain? *Pain Management Nursing*, *3*(3), 77–80.
- McCaffery, M. & Ferrell, B. R. (1995). Nurses' knowledge about cancer pain: A survey of five countries. *Journal of Pain and Symptom Management, 10*(5), 356–369.
- McCaffery, M. & Ferrell, B. R. (1996). Correcting misconceptions about pain assessment and use of opiod analgesics: Education strategies aimed at public concerns. *Nursing Outlook, 44*(4), 184–190.
- McCaffery, M. & Ferrell, B. R. (1997). Nurses' knowledge of pain assessment and management: How much progress have we made? *Journal of Pain and Symptom Management*, 14(3), 175–186.
- McCaffery, M. & Pasero, C. (1999). Pain: Clinical manual (2nd ed.). St. Louis, MO: Mosby.
- McCaffery, M. & Robinson, E. S. (2002). Your patient is in pain: Here's how you respond. *Nursing2002, 32*(10), 36–45.
- McCaffery, M., Ferrell, B. R. & Pasero, C. (2000). Nurses' personal opinions about patients' pain and their effect on recorded assessments and titration of opioid doses. *Pain Management Nursing*, 1(3), 79–87.
- McCaffery, R., Frock, T. L. & Garguilo, H. (2003). Understanding chronic pain and the mindbody connection. *Holistic Nursing Practice*, 17(6), 281–287.
- McLean, S. A., Clauw, D. J., Abelson, J. L. & Liberzon, I. (2005). The development of persistent pain and psychological morbidity after motor vehicle collision: Integrating the potential role of stress response systems into a biopsychosocial model.
 Psychosomatic Medicine, 67, 783–790.

- McLean, S., Domeier, R., DeVore, H., Hill, E., Maio, R. & Frederiksen, S. (2004). The epidemiology of pain in the pre-hospital setting. *Prehospital Emergency Care*, 6(4), 402–405.
- McWilliams, L. A., Cox, B. J. & Enns, M. W. (2003). Mood and anxiety disorders associated with chronic pain: An examination in a nationally representative sample. *Pain, 106*, 127–133.
- Mebrouk, J. (2008). Perception of nursing care: Views of Saudi Arabian female nurses. *Contemporary Nurse, 28*,149-161.
- Melzack, R. (1975). The McGill Pain Questionnaire: Major properties and scoring methods. *Pain, 1*(3), 277–299.
- Melzack, R. & Wall, P. D. (1965). Pain mechanisms: A new theory. *Science*, *150*(3699), 971–979.
- Melzack, R. & Wall, P. D. (2008). *The challenge of pain. The medical classic, now with a new introduction* (2nd ed.). United Kingdom: Penguin.
- Mendell, J. R. & Sahenk, Z. (2003). Painful sensory neuropathy. New England Journal of Medicine, 348(13), 1243–1255.
- Merriam, S. B. (2009). *Qualitative research: A guide to design and implementation*. San Francisco, CA: John Wiley and Sons.
- Molony, S. L., Kobayashi, M., Holleran, E. A. & Mezey, M.(2005). Assessing pain as the fifth vital sign in long-term care facilities: Recommendations from the field. *Journal of GerontologicalNursing*, *31*(3), 16–24.
- Migiro, S. O., & Magangi, B. A. (2011). Mixed methods: A review of literature and the future of the new research paradigm. *African Journal of Business Management*, 5(10), 3757-3764.
- Millan, M. J. (1999). The induction of pain: An integrative review. *Progress in Neurobiology*, *57*, 1–164.

- Miller-Rosser, K. Chapman, Y. Francis, K. (2006) Historical, Cultural, and Contemporary Influences on the Status of Women in Nursing in Saudi Arabia. *Online Journal of Issues Nursings*, 11(3), 8.
- Mills, R. (2004). The pleasures of health, the joys of illness, and the delights of longing to meet my maker: The Islamic perspective of life and death. Retrieved 24 October 2013 from http://www.muslim- canada.org/health_sickness
- Ministry of Health. (2008). *Health statistical year book 2008*. Retrieved 12 May 2011 from http://www.moh.gov.sa/Ministry/Statistics/book/flash/1429/MOH_Report_1429.html
- Ministry of Health. (2011). *Health statistical year book 2011*. Retrieved 6 March 2013 from http://www.moh.gov.sa/Ministry/Statistics/Book/Pages/default.aspx
- Mitchell, J. E. (2009). Job satisfaction and burnout among foreign-trained nurses in Saudi Arabia: A mixed-methods study. (Thesis). University of Phoenix, Arizona.
- Moceri, J. T. & Drevdahl, D. J. (2012). Nurses' knowledge and attitudes toward pain in the emergency department. *Journal of Emergency Nursing*, 1–7.
- Monsivais, D. (2011). Promoting culturally competent chronic pain management using the clinically relevant continuum model. *Nursing Clinics of North America*, *46*(2), 163–169.
- Moustakas, C. (1994). *Phenomenological research methods*. Thousand Oaks, CA: Sage Publications.
- Mufti, M. H. (2000). *Healthcare development strategies in the Kingdom of Saudi Arabia*. New York: Kluwer Academic/Plenum.
- Mughees, A. (2006). Better caring for Muslim patients. *World of Irish Nursing & Midwifery,* 14(7), 24–25.
- Musclow, S. L., Sawhney, M. & WattdWatson, J. (2002). The emerging role of advanced nursing practice in acute pain management throughout Canada. *Clinical Nurse Specialist*, 16(2), 63–67.

- Narayan, M. C. (2010). Culture's effects on pain assessment and management. *American Journal of Nursing*, *110*(4), 38–47.
- National Center for Health Statistics. (2006). *Health, United States, 2006*. Hyattsville, MD. Retrieved 1 February 2013 from http://www.cdc.gov/nchs/data/hus/hus06.pdf
- Newton-John, T. (2005). Psychological management of chronic pain. In A. Holdcroft & S. Jaggar (Eds.), *Core topics in pain* (pp. 293–298). Cambridge, England: Cambridge University Press.
- National Institutes of Health [Internet]. NIH guide: new directions in pain research: I. Bethesda, MD: National Institutes of Health. 1998 Sept 4. Retrieved January 11, 2013 fromhttp://grants.nih.gov/grants/guide/pa-files/PA-98-102.html
- Ott, B. B., Al-Khadhuri, J. & Al-Junaibi, S. (2003). Preventing ethical dilemmas: Understanding Islamic health care practices. *Pediatric Nursing*, *29*(30), 227–253.
- Pasero, C. & McCaffery, M. (2004). Comfort-function goals: A way to establish accountability for pain relief. *American Journal of Nursing*, *104*(9), 77–78, 81.
- Pasero, C. & McCaffery, M. (2007). Orthopaedic postoperative pain management. *Journal of PeriAnesthesia Nursing*, 22(3), 160–174.
- Perry, K., Nicholas, M. & Middleton, J. (2010). Comparison of a pain management program with usual care in a pain management center for people with spinal cord injury-related chronic pain. *Clinical Journal of Pain, 26*(3), 206–216.
- Peter, E. & Watt-Watson, J. (2002). Unrelieved pain: An ethical and epistemological analysis of distrust in patients. *Canadian Journal of Nursing Research*, *34*(2), 65–80.
- Phillips, C. J. (2006). Economic burden of chronic pain. *Expert Review of Pharmacoeconomics & Outcomes Research, 6*(5), 591–601.
- Phillips, C. J., & Schopflocher, D. (2008). The economics of chronic pain. In S. Rashiq, D. Schopflocher, P. Taenzer, & E. Jonsson (Eds.), *Health policy perspectives on chronic pain*. Weinheim, Germany: Wiley-Blackwell.

- Plaisance, L. & Logan, C. (2006). Nursing students' knowledge and attitudes regarding pain. *Pain Management Nursing*, 7(4), 167–175.
- Polomano, R. C., Dunwoody, C. J., Krenzischek, D. A. & Rathmell, J. P. (2008). Perspective on pain management in the 21st century. *Journal of PeriAnesthesia Nursing*, 23(1), 4– 14.
- Poole, H., White, S., Blake, C., Murphy, P. & Bramwell, R. (2009). Depression in chronic pain patients: Prevalence and measurement. *Pain Practice*, 9(3), 173–80.
- Porth, C. (2004). *Essentials of pathophysiology*. Baltimore, MD: Lippincott Williams & Wilkins.
- Powell, R., Downing, J., Ddungu, H. & Mwangi-Powell, F. (2009). Guide to pain management in low-resource settings. Retrieved 25 March 2013 from http://www.iasp pain.org/AM/Template.cfm?Section=Guide_to_Pain_Management_in_Low_Resource _Settings&Template=/CM/ContentDisplay.cfm&ContentID=12173
- Purnell, L. (2005). The Purnell model for cultural competence. *Journal of Multicultural Nursing & Health*, *11*(2), 7–15.
- Rahimi-Madiseh, M., Tavakol, M. & Dennick, R. (2010). A quantitative study of Iranian nursing students' knowledge and attitudes towards pain: Implication for education. *International Journal of Nursing Practice*, 16(5), 478–483.
- Ramer, L., Richardson, J., Cohen, M., Bedney, C., Danley, K. & Judge, E. (1999).
 Multimeasure pain assessment in an ethnically diverse group of patients with cancer. *Journal of Transcultural Nursing*, 10(2), 94–101.
- Registered Nurses' Association of Ontario. (2002). *Toolkit: Implementation of clinical practice guidelines*. Toronto, Canada: Registered Nurses' Association of Ontario.
- Reiman, M. T. & Gordon, M. (2007). Pain management competency evidenced by a survey of pediatric nurses' knowledge and attitudes. *Pediatric Nursing*, 33(4), 307–312.

- Rejeh, N., Ahmadi, F., Mohammadi, E., Kazemnejad, A. & Anoosheh, M. (2008). Nurses' experiences and perceptions of influencing barriers to post-operative pain management. *Scandinavian Journal of Caring Sciences*, 23(2), 274–281.
- Richardson, G. (2012). Pain expression in different cultures: A qualitative study of the analysis for the cues of pain in different cultures (Bachelor Thesis). Vaasa, Finland: Novia University of Applied Sciences.
- Rieman, M. T. & Gordon, M. (2007). Pain management competency evidenced by a survey of pediatric nurses' knowledge and attitudes. *Pediatric Nursing*, 33(4), 307–312.
- Rosdahl, C. B. & Kowalski, M. T. (2008). *Textbook of basic nursing*. Philadelphia, PA: Lippincott Williams & Wilkins. Retrieved 9 April 2013 from http://books.google.com.au/books?id=odY9mXicPlYC
- Rupp, T. & Delaney, K. A. (2004). Inadequate analgesia in emergency medicine. Annals of Emergency Medicine, 43(4), 494–503.
- Rutherford, A. (2001). *Introducing ANOVA and ANCOVA: A GLM approach*. Thousand Oaks, CA: Sage Publications.
- Sawyer, J., Haslam, L., Daines, P. & Stilos, K. (2010). Pain prevalence study in a large Canadian teaching hospital. Round 2: Lessons learned? *Pain Management Nursing*, 11(1), 45–55.
- Schafheutle, E. I., Cantrill, J. A. &Noyce, P. R. (2001). Why is pain management suboptimal on surgical wards? *Journal of Advanced Nursing*, *33*(6), 728–737.
- Schechter, N. L., Berde, C. B. & Yaster, M. (Eds.). (2003). Pain in infants, children and adolescents (2nd ed.). Philadelphia, PA: Lippincott Williams & Wilkins.
- Schopflocher, D., Jovey, R. et al. (2010). The burden of pain in Canada: Results of a Nanos survey. Pain Research & Management: In Press.

Shannon, V. H. (2011). Embrace the lessons. Bloomington, IN: Balboa Press.

- Shepherd, E., Woodgate, R. S. & Sawatzky, J. (2010). Pain in children with central nervous system cancer: A review of the literature. *Oncology Nursing Forum*, 37(4), 318–330. Retrieved 12 July 2011 from http://www.ons.org/Publications/ONF/
- Sinclair, M. (2007). Editorial: A guide to understanding theoretical and conceptual frameworks. *Evidence Based Midwifery*, *5*(2), 39.
- Sloman, R., Rosen, G., Rom, M. & Shir, Y. (2005). Nurses' assessment of pain in surgical patients. *Journal of Advanced Nursing*, 52(2), 125–132.
- Sloman, R., Wruble, A., Rosen, G. & Rom, M. (2006). Determination of clinically meaningful level of pain reduction in patients experiencing acute postoperativepain. *Pain Management Nursing*, 7(4), 153–158.
- Smith, J. E. (2002). Analysis of differences in entry-level RN practice by educational preparation. *Journal of Nursing Education*, *41*(11), 491–494.
- Solowiej, K., Mason, V. & Upton, D. (2009). Review of the relationship between stress and wound healing: Part 1. *Journal of Wound Care, 18*(9), 357–366.
- Sommer, M., de Rijke, J. M., van Kleef, M., Kessels, A. G., Peters, M. L., Geurts, J. W., ... Marcus, M. A. (2008). The prevalence of postoperative pain in a sample of 1490 surgical inpatients. *European Journal of Anaesthesiology*, 25(4), 267–274.
- Standage, T. (2005). *A history of the world in six classes*. New York, NY: Walker and Company.
- Strassels, S. A. (2008). Concern about opioid addiction as a barrier to acute pain management, University of Tennessee Advanced Studies in Pharmacy, 5(2), 48–51.
- Symes, D. (2011). *The heart attack recovery plan: The positive approach to managing your lifestyle*. London, England: Ebury Publishing. Retrieved 12 January 2013 from http://books.google.com.au/books? id=8tXu_jOCUEAC
- Tapp, J. & Kropp, D. (2005). Evaluating pain management delivered by direct care nurses. Journal of Nursing Care Quality, 20(2), 167–173.

- Taylor, A. L. (2007). Addressing the global tragedy of needless pain: Rethinking the United Nations' single convention on narcotic drugs. *Journal of Law Medicine and Ethics*, 35(4), 556–570.
- Teddlie, C. & Tashakkori, A. (2009). *Foundations of Mixed Methods Research*. ThousandOaks, CA: Sage Publications.
- Textor, L. H. & Porock, D. (2006). The pain management knowledge of nurses practicing in a rural Midwest retirement community. *Journal for Nurses in Staff Development*, 22(6), 307–312.
- Thibodeaux, A. G. & Deatrick, J. A. (2007). Cultural influence on family management of children with cancer. *Journal of Pediatric Oncology Nursing*, *24*(4), 227–233.
- Tobin, G. A. & Begley, C. M. (2004). Methodological rigour within a qualitative framework. *Journal of Advanced Nursing, 48*(4), 388–396.
- Todd, I. A., Tew, T. E. & Macdonald, D. W. (2000). Arable habitat use by wood mice (Apodemus sylvaticus). 1. Macrohabitat. *Journal of Zoology*, *250*(3), 299–303.
- Toothaker, L. E. (1991). *Multiple comparisons for researchers*. Newbury Park, CA: Sage Publications.
- Tracey, I. & Mantyh, P. H. (2007). The cerebral signature for pain processing and its modulation. *Neuron*, 55(3), 377–391.
- Tsai, F. C., Tsai, Y. F., Chien, C. & Lin, C. C. (2007). Emergency nurses' knowledge of perceived barriers in pain management in Taiwan. *Journal of Clinical Nursing*, 16(11), 2088–2095.
- Tsang, A., Von Korff, M., Lee, S, Alonso, J., Karam, E., Angermeyer, M. C., . . . Watanabe, M. (2008). Common chronic pain conditions in developed and developing countries:
 Gender and age differences and comorbidity with depression-anxiety disorders. *Journal of Pain*, 9(10), 883–891.

- Tse, M. Y. & Chan, S. H. (2004). Knowledge and attitudes in pain management: Hong Kong nurses' perspective. *Journal of Pain & Palliative Care Pharmacotherapy*, 18(1), 47– 58.
- Tumulty, G. (2001). Professional development of nursing in Saudi Arabia. *Journal of Nursing Scholarship*, 33(3), 285–290.
- Twycross, A. (2002). Educating nurses about pain management: The way forward. *Journal of Clinical Nursing*, *11*(6), 705–714.
- Vadivelu, N., Mitra, S. & Narayan, D. (2010). Recent advances in postoperative pain management. *Yale Journal of Biology and Medicine*. Retrieved 11 February 2011 from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC2844689/
- Vallerand, A. H., Collins-Bohler, D., Templin, T. & Hasenau, S. M. (2007). Knowledge of and barriers to pain management in caregivers of cancer patients receiving homecare. *Cancer Nursing*, 30(1), 31–37.
- Vallerand, A. H., Riley-Doucet, C., Hasenau, S. M. & Templin, T. (2004). Improving cancer pain management by homecare nurses. *Oncology Nursing Forum*, 31(4), 809–816.
- van Dijk, M., Peters, W. B., van Deventer, P. & Tibboel, D. (2005). The COMFORT
 Behavior Scale: A tool for assessing pain and sedation in infants. *American Journal of Nursing*, 105(1), 33–36.
- Van Niekerk, L. M. & Martin, F. (2001). Tasmanian nurses' knowledge of pain management. International Journal of Nursing Studies, 38(2), 141–152.
- Van Niekerk, L. M. & Martin, F. (2003). The impact of the nurse–physician relationship on barriers encountered by nurses during pain management. *Pain Management Nursing*, 4(1), 3–10.
- Vidyasagar, G. & Rea, D. M. (2004). Saudi women doctors: Gender and careers within Wahhabic Islam and a 'westernised' work culture. *Women's Studies International Forum, 27*, 261-280.

- Vincent, C. & Denyes, M. (2004). Relieving children's pain: Nurses' abilities and analgesic administration practices. *Journal of Pediatric Nursing*, 19(1), 40–50.
- Vitor, A. O., Da Ponte, E. L., Soares, P. M., Rodrigues, M. E. D. S., De Sousa Lima, R. C., Carvalho, K. D. M., ... Vasconcelos, S. M. M. (2008). The psychophysiology of pain: a literature review. *Reciis*, 2(1), 85–94.
- Walsh, N. E., Brooks, P., Meike Hazes, J. M., Walsh, R. M., Dreinhöfer, K., Woolf, A. D., . .
 Lidgren, L. (2008). Standards of care for acute and chronic musculoskeletal pain:
 The bone and joint decade. *Archives of Physical Medicine Rehabilitation*, 89, 1830–1845.
- Walston, S., Al-Harbi, Y. & Al-Omar, B. (2008). The changing face of health care in Saudi Arabia. *Annals of Saudi Medicine*, 28(4), 243–250.
- Wand, B. M., Bird, C., McAuley, J. H., Dore, C. J., MacDowell, M. & De Souza, L. H.
 (2004). Early intervention for the management of acute low back pain. *Spine*, 29(21), 2350–2356.
- Wandner, L, Scipio, C., Hirsh, A., Torres, C., & Robinson, M., (2012). The perception of pain in others: How gender, race, and age influence pain expectations. *The Journal of Pain*, *13*(3), 220-227.
- Wang, H. L. & Tsai, Y. F. (2010). Nurses' knowledge and barriers regarding pain management in intensive care units. *Journal of Clinical Nursing*, 19(21–22), 3188– 3196.
- Watt-Watson, J. (1999). Pain and pain control. In W. Phipps, J. Sands & J. Marek (Eds.),
 Medical-surgical nursing concepts and clinical practice (6th ed., pp. 321–346). St
 Louis, MO: Mosby.
- Watt-Watson, J., McGillion, M., Hunter, J., Choiniere, M., Clark, A. J., von Baeyer, C. L. &
 Webber, K. (2009). A survey of pain curricula in prelicensure health science faculties in Canadian universities. *Pain Research and Management*, 14(6), 439–444.

- Watt-Watson, J., McGillion, M., Hunter, J., Choiniere, M., Clark, A. J., Dewar, A., . . .
 Webber, K. (2009). A survey of prelicensure pain curricula in health science faculties in Canadian universities. *Pain Research Management*, *14*(6), 439–444.
- Watt-Watson, J., Stevens, K., Garfinkle, M., Steiner, L. & Gallop, P. (2001). Relationship between nurses' knowledge and pain management outcomes for their postoperative cardiac patients. *Journal of Advanced Nursing*, 365(4), 535–545.
- Weech-Maldonado, R., Elliot, M., Schiller, C., Allyson, H. & Hays, R. D. (2012). Can hospital cultural competency reduce disparities in patient experiences with care? *Medical Care*, 50(11: Supplement 3), 48–55.
- Wells, N., Pasero, C. & McCaffery, M. (2008). Improving the quality of care through pain assessment and management. In R. G. Hughes (Ed.), *Patient safety and quality: An evidence-based handbook for nurses*. Rockville, MD: Agency for Healthcare Research and Quality. Retrieved 1 November 2012 from http://www.ncbi.nlm.nih.gov/pubmed/21328752
- Welman, J. C. & Kruger, S. J. (2005). *Research methodology*. 3rd ed. Johannesburg: Oxford University Press.
- Whelan, C., Jin, L. & Meltzer, D. (2004). Pain and satisfaction with pain control in hospitalized medical patients: No such thing as low risk. *American Medical Association*, 164(2), 175–180.
- Wilkinson, L. (1999). Statistical methods in psychology journals: Guidelines and explanations. *American Psychologist*, 54(8), 594–604.
- Williams, S. L., Haskard, K. B. & DiMatteo, M. R. (2007). The therapeutic effects of the physician–older patient relationship: Effective communication with vulnerable older patients. *Clinical Interventions in Aging*, 2(3), 453–467.

- Wilson-Stronks, A., Lee, K. K., Cordero, C. L., Kopp, M. L. & Galvez, E. (2008). One size does not fit all: Meeting the health care needs of diverse populations [electronic resource]. Oakbrook Terrace, IL: The Joint Commission.
- Wilson, B. (2007). Nurses' knowledge of pain. *Journal of Clinical Nursing*, 16(6), 1012–1020.
- Winterowd, C., Beck, A. T., & Gruener, D. (2003). Cognitive Therapy with Chronic Pain Patients. Springer Publishing Company.
- Woolf, C. J. (2004). Pain: Moving from symptom control toward mechanism-specific pharmacologic management. *Annals of Internal Medicine*, 140(6), 441–451.
- Wozniak, R. H. (1992). *Mind and body: Rene Descartes to William James*. Bethesda, MD and Washington, DC: National Library of Medicine and American Psychological Association.
- Wu, C. L., Naqibuddin, M. & Rowlingson, A. J. (2003). The effect of pain on health-related quality of life in the immediate postoperative period. *Anesthesia and Analgesia Journal*, 97(4), 1078–1085.
- Yates, P., Dewar, A., & Edwards, H. (1998). The prevalence and perception of pain amongst hospital in-patients. *Journal of Clinical Nursing*, 7, 521-530.
- Yava, A., Çicek, H., Tosun, N., Özcan, C., Yildiz, D. & Dizer, B. (2013). Knowledge and attitudes of nurses about pain management in Turkey. *International Journal of Caring Sciences*, 6(3), 494–505.
- Yildirim, Y. K., Cicek, F. & Uyar, M. (2008). Knowledge and attitudes of Turkish oncology nurses about cancer pain management. *Pain Management Nursing*, 9(1), 17–25.
- Young, C. et al. (2008). Transition from acute to chronic pain and disability: A model including cognitive, affective, and trauma factors. Pain, 134(1–2):69–79.

- Young, J., Siffleet, J., Nikoletti, S. & Shaw, T. (2006). Use of a behavioural pain scale to assess pain in ventilated, unconscious and/or sedated patients. *Intensive & Critical Care Nursing*, 22(1), 32–39.
- Zikmund, W. (2003). *Business research methods* (7th ed.). Mason, OH: Thomson Learning South Western.
- Zodpey, S. P. (2004). Sample size and power analysis in medical research. *Indian Journal of Dermatology, Venereology and Leprology, 70*(2), 123–128.
- Zuccaro, S. M., Vellucci, R., Sarzi-Puttini, P., Cherubino, P., Labianca, R. & Fornasari, D.
 (2012). Barriers to pain management: Focus on opioid therapy. *Clinical Drug Investigation*, 32(1), 11–19.

Appendices

Appendix A: Studies that Utilised the KASRP Tool

Researchers	Year	Location	Methodology	Sample	Major Findings
Yava, Çicek, Tosun, Özcan, Yildiz & Dizer	2013	Turkey	KASRP tool	246 nurses from different departments	 Mean score: 39.65% Items relating to medications and dosages received the lowest scores
Lewthwaite	2011	Canada	KASRP tool	324 nurses from different departments	 Mean score: 79% Almost half of the sample scored 80% or greater Items relating to medications and dosages received the lowest scores
Wang & Tsai	2010	Taiwan	KASRP tool	370 nurses from intensive care units	 Mean score: 53.4% Deficits in nurses' knowledge regarding pharmacology
Rahimi- Madiseh, Travakol & Dennick	2010	Iran	KASRP tool	146 student nurses	 Mean score: 37% Case studies regarding analgesia dosage received the lowest scores
Huth, Gregg & Lin	2010	Mexico	Pediatric KASRP tool	106 pediatric nurses	 Mean score pre-educational initiative: 46.6% Mean score post-educational initiative: 55.6%
Yildirim,Cice k & Uyar	2008	Turkey	KASRP tool	68 oncology nurses	 Mean score: 35.4% Case studies regarding analgesia dosage received the lowest scores Deficits in nurses' knowledge regarding pharmacology
Lui, So & Fong	2008	China	KASRP tool	143 nurses	 Mean score: 47.7% Deficits in nurses' knowledge regarding pharmacology and non- pharmacology interventions
Tsai, Tsai, Chien & Lin	2007	Taiwan	KASRP tool	249 emergency department nurses	 Mean score: 49.2% Case studies regarding analgesia dosage received the lowest scores Overconcern about possibilities of developing respiratory depression and addiction Deficits in nurses' knowledge regarding pharmacology
Rieman & Gordon	2007	USA	Pediatric KASRP tool	295 pediatric nurses	 Mean score: 74% Deficits in nurses' knowledge regarding pharmacology and side effects of narcotics
Bernardi, Catania & Tridello	2007	Italy	KASRP tool	66 hospice nurses	 Mean score: 62.7% Case studies regarding analgesia dosage received the lowest scores Deficits in nurses' knowledge regarding self-evaluation and opioids
Matthews & Malcolm	2007	Northern Ireland	KASRP tool	113 nurses	 Mean score: 73.8% Overconcern about possibilities of developing respiratory depression Underestimation of pain
Plaisance &	2006	USA	KASRP tool	313 student	• Mean score: 64.0%

Logan				nurses	Overconcern about possibilities of addiction.Case studies regarding analgesia
					dosage received the lowest scores
Coulling	2005	UK	KASRP tool	49 nurses 33 doctors (one hospital)	 Mean score: 71% Deficits in nurses' knowledge regarding opioid administration 47% believed nurses' lack of knowledge was a barrier to pain management Overconcern about possibilities of developing respiratory depression and addiction
Tapp &	2005	USA	KASRP tool	23 surgical	• Mean score: 69.4%
Kropp	2000			nurses	 Overconcern about possibilities of developing respiratory depression and addiction Deficits in nurses' knowledge regarding pharmacology
Tse & Chan	2004	China	KASRP tool	601 nurses	• Mean score: 44.0%
	2001			(three hospitals)	 Deficits in nurses' knowledge regarding pharmacology and non- pharmacology Exaggerated fears of respiratory depression
Vincent &	2004	LISA	KASRP tool	67 nediatric	Maan soore: 77%
Denyes	2004	0.5A		nurses	 Unbelieving and underestimating patients' self-reports of pain Overconcern about possibilities of developing respiratory depression for patients
Lai, Chen,	2003	Taiwan	KASRP tool	1,797 nurses	• Mean score: 50.5%
Tsai, Lo, Wei, Hong, Hsiu,				from different departments	• Case studies regarding analgesia dosage received the lowest scores
Hsiao-Sheen, Chen, Kao, Huang, Chang, Chen & Guo					• Deficits in nurses' knowledge regarding analgesics and patients' self-reports of pain
Van-Niekerk & Martin	2001	Australia	KASRP tool	1,015 nurses from different departments	 Mean score: 71% Deficits in nurses' knowledge regarding pharmacology
Erkes, Carr & Mayo	2001	USA	KASRP tool	30 intensive care nurses	 Pre-education mean score: 72.9% Post-education mean score: 86.2%
Howell, Butler, Vincent, Watt-Watson & Stearns	2000	Canada	KASRP tool	53 oncology nurses	 Deficits in nurses' knowledge regarding assessment and management of pain Underestimatingpatients' self- reports of pain Deficits in nurses' knowledge regarding analgesic side effects

Appendix B: Demographic Data

Please respond to the questions below by ticking ($\sqrt{}$) one appropriate answer:

1. What is your gender?

⊖ Male

◯ Female

2. How old are you?

..... Years

3. What is your nationality?

🔘 Saudi

- O Non-Saudi Specify:-
- 4. What is your highest nursing education?

O Diploma

◯ Bachelors

O Master

- ODoctorate
- 5. What is your religious belief?
- \bigcirc Muslim
- O Christian—Please indicate what faith
- \bigcirc Buddhist
- ⊖ Hindu

○ None (Atheist)

- Other specify.....
- 7. What is your current position?
- ◯ Staff nurse

 \bigcirc Head nurse

- O Nursing manager (Supervisor, Director)
- Other, please indicate.....
- 8. In which department are you working now?

.....

9. Did you ever attend or take any course about pain assessment or management?

○ Yes—Please indicate how many.....

🔿 No

Appendix C: Consent Form for Participants



School of Health Sciences, Nursing and Midwifery

> GPO Box 71 Bundoora VIC 3083 Australia Ph: +61 3 9925 7447 Fax: +61 3 9467 5286

Prescribed Consent Form for Persons Participating in Research Projects Involving Interviews, Questionnaires or Disclosure of Personal Information

Portfolio School of		Science, Engineering and Health Sciences						
		Health Sciences						
Name o	of participant:							
Project Title:			Pain Management in Hail Region Hospitals					
Name(s) of investigators: (1)		Mr Hamdan Albaqawi (PhD Candidate)						
		(2)	Prof Phil Maude	Phone:	+61 3 99257447			
		(3)	Prof Lina Shahwan-Akl	Phone:	+61 3 99257443			
1.	I have received a statement explaining the interview/questionnaire involved in this project.							
2.	I consent to participate in the above project, the particulars of which - including details of the interviews or questionnaires - have been explained to me.							
3.	I authorise the investigator or his or her assistant to interview me or administer a questionnaire.							
4	I acknowledge th	at:						

- (a) Having read the Plain Language Statement, I agree to the general purpose, methods and demands of the study.
- (b) I have been informed that I am free to withdraw from the project at any time and to withdraw any unprocessed data previously supplied.
- (c) The project is for the purpose of research and/or teaching. It may not be of direct benefit to me.
- (d) The privacy of the personal information I provide will be safeguarded and only disclosed where I have consented to the disclosure or as required by law.
- (e) The security of the research data is assured during and after completion of the study. The data collected during the study may be published, and a report of the project outcomes will be provided to the Saudi Ministry of Health. Any information that will identify me will not be used.

Participant's Consent

Participant:

(Signature)

Date:

Date:

Witness:

(Signature)

Participants should be given a photocopy of this consent form after it has been signed.

Any complaints about your participation in this project may be directed to the Executive Officer, RMIT Human Research Ethics Committee, Research & Innovation, RMIT, GPO Box 2476V, Melbourne, 3001. The telephone number is (03) 9925 2251. Details of the complaints procedure are available from the above address.



School of Health Sciences, Nursing and Midwifery

GPO Box 71 Bundoora VIC 3083 Australia Ph: +61 3 9925 7447 Fax: +61 3 9467 5286

Project Title:

Pain Management in Hail Region Hospitals.

Investigator:

Hamdan Albaqawi (PhD Candidate- Nursing and Midwifery, School of Health Sciences, RMIT University, <u>S3259986@student.rmit.edu.au</u>)

Supervisors:

- Associate Professor Phillip Maude, PhD, School of Health Sciences, RMIT University, <u>Phillip.Maude@rmit.edu.au</u> Phone: 99257447
- Associate Professor Lina Shahwan-Akl PhD, School of Health Sciences, RMIT University, <u>lina.shahwan-akl@rmit.edu.au</u> Phone: 99257443

DearProspective participant,

You are invited to participate in a research project being conducted at RMIT University. This information sheet describes the project in straightforward language, or 'plain English'. Please read it carefully and be confident that you understand its contents before deciding whether to participate. If you have any questions about the project, please ask the investigator.

Who is involved in this research project? Why is it being conducted?

This research is being conducted by Hamdan Albaqawi as a requirement for the award of a PhD in the Discipline of Nursing and Midwifery, RMIT University, Australia, under the supervision of Assoc. Prof. Phillip Maude and Assoc. Prof. Lina Shahwan-Akl. The research has the approval of the Saudi Arabia Ministry of Health and the RMIT University Ethics committee.

Why have you been approached?

As nurses working in the Hail Region Hospital, you have been approached to participate in this study.

What is the project about? What are the questions being addressed?

You have been invited to respond to a survey in order to determine nurses' knowledge and attitudes regarding pain management and to identify possible barriers to achieving optimal pain management in Hail Region Hospitals. You may also respond to a request to participate in a face-to-face interview by personally contacting the researcher through email. Your participation in both the survey and the face-to-face interview is completely voluntary and would be greatly appreciated.

If I agree to participate, what will I be required to do?

You will find a questionnaire attached to this Plain Language Statement, which has been distributed by the Nursing Education Department. After reviewing and understanding this Plain Language Statement, you will be required to complete demographic questions and the survey, which will take 15–20 minutes. The survey will include true/false statements and multiple choice questions. This survey will be anonymous, and participants have the right to withdraw from the study at any time. Questionnaires must be completed in your own time, and you will be required to submit the completed questionnaire in a sealed box located on your ward.
If you wish to participate in the face-to-face interview, which will be voice recorded, you will be required to sign a consent form. The face-to-face interview will take around 30–45 minutes. Participants in the face-to-face interview will be asked about the result from the survey and what they think about pain management in Hail Region Hospitals and whether there are any barriers or enhancing factors to providing effective pain management. Interview data will be confidential, and audiorecordings will be transcribed by the researcher using a code and not your name or any identifiable data.

What are the risks or disadvantages associated with participation?

There are no risks associated with your participation in this research project. All responses will remain confidential, be reported as group data and will have no influence on your employment.

What are the benefits associated with participation? It is hoped that this project will help to identify barriers to achieving optimal pain management in Hail Region Hospitals and how nurses from different cultures could affect the delivery of effective pain management. The project will explore and describe pain management practices and has the potential to improve the quality of nursing care of patients in these hospitals. This study will add to the existing body of knowledge regarding pain management.

What will happen to the information I provide?

All information gathered as part of this research will be securely stored for a period of five years in the School of Health Sciences, RMIT University. The data can only be accessed by the researcher and supervisors. After five years, the data will be destroyed. The data collected will be analysed, and the results may be published in academic journals or conferences without including any personal information that has the potential to identify either you or your health agency.

What are my rights as a participant?

Your participation in this research is voluntary. As a participant, you have the right to: withdraw your participation at any time; have any unprocessed data withdrawn and destroyed, provided it can be reliably identified, and provided that so doing does not increase your risk; and have any questions answered at any time.

Due to the nature of this data collection process, I am not obtaining written informed consent unless you elect to be interviewed.

Whom should I contact if I have any questions?

If you have any questions regarding this research, please contact the researcher <u>S3259986@student.rmit.edu.au</u> or his supervisors at the following addresses <u>Phillip.Maude@rmit,edu.au</u> and <u>Lina.Shahwan-Akl@rmit,edu.au</u>

You may also contact the following person in Saudi Arabia should you have any concerns about this research:

General Directorate of Nursing, Ministry of Health Saudi Arabia Muneera Bint Hamdan Al-Osaimy (General Director)

Yours Sincerely

Hamdan Albaqawi Associate Professor Phillip Maude Associate Professor Lina Shahwan-Akl

Any complaints about your participation in this project may be directed to the Executive Officer, RMIT Human Research Ethics Committee, Research & Innovation, RMIT, GPO Box 2476V, Melbourne, 3001. The telephone number is (03) 9925 2251. Details of the complaints procedure will also be available from this address.

Appendix E: Letter to Participate in Interview

Dear Participant,

If you wish to participate in the face-to-face interview, please register your name and contact details at the Education Department in your hospital. The researcher will arrange a suitable time to conduct the face-to-face interview with you. This interview will be voice recorded, and you will be required to sign a consent form prior to the commencement of the interview.

The face-to-face interview will take around 30–45 minutes to obtain an in-depth knowledge of theenhancing factors and barriers to appropriate pain management by nurses in Hail hospitals. Interview data will be confidential, and audiorecordings will be transcribed by the researcher using a code and not your name or any identifiable data.

Sincerely yours, Hamdan Albaqawi (PhD Candidate) RMIT University School of Health Sciences

Appendix F: Knowledge and Attitudes Survey Regarding Pain



April 2008

Dear Colleague:

The "Knowledge and Attitudes Survey Regarding Pain" tool can be used to assess nurses and other professionals in your setting and as a pre and post test evaluation measure for educational programs. The tool was developed in 1987 and has been used extensively from 1987 - present. The tool was revised and is now being tested in pain education courses to conduct psychometric analysis on this updated version. There have been minor edits in April 2008.

Regarding issues of reliability and validity: This tool has been developed over several years. Content validity has been established by review of pain experts. The content of the tool is derived from current standards of pain management such as the American Pain Society, the World Health Organization, and the Agency for Health Care Policy and Research. Construct validity has been established by comparing scores of nurses at various levels of expertise such as students, new graduates, oncology nurses, graduate students, and senior pain experts. The tool was identified as discriminating between levels of expertise. Test-retest reliability was established (r>.80) by repeat testing in a continuing education class of staff nurses (N=60). Internal consistency reliability was established (alpha r>.70) with items reflecting both knowledge and attitude domains.

Regarding analysis of data: We have found that it is most helpful to avoid distinguishing items as measuring either knowledge or attitudes. Many items such as one measuring the incidence of addiction really measures both knowledge and attitude about addiction. Therefore, we have found the most benefit to be gained from analyzing the data in terms of the percentage of complete scores as well as in analyzing individual items. For example, we have found it very helpful to isolate those items with the least number of correct responses and those items with the best scores.

Enclosed for your use is a copy of our instrument and an answer key. You may use and duplicate the tool for any purpose you desire in whole or in part. References to some of our studies which have included this tool or similar versions are included below.

We also acknowledge the assistance of several of our pain colleagues including Pam Kedziera, Judy Paice, Deb Gordon, June Dahl, Hob Osterlund, Chris Pasero, Pat Coyne and Nessa Coyle in the current revisions. If using or publishing the tool results please cite the reference as **"Knowledge and Attitudes Survey Regarding Pain" developed by Betty Ferrell, RN, PhD, FAAN and Margo McCaffery, RN, MS, FAAN**, (http://prc.coh.org), revised 2008.

We hope that our tool will be a useful aid in your efforts to improve pain management in your setting.

Sincerely,

Betty R. Ferrell, RN, PhD, FAAN Research Scientist Margo McCaffery, RN, MS, FAAN Lecturer and Consultant

References:

1. McCaffery M, Ferrell BR (1994). "Understanding opioids & addiction." Nursing 94, 24(8): 56-59.

2. Ferrell BR, McCaffery M, Rhiner M. (1992). "Pain and addiction: An urgent need for changing nursing education." Journal of Pain and Symptom Management, 7(2): 117-124.

3. Ferrell BR, Grant M, Ritchey KJ, Ropchan R, Rivera LM (1993). "The Pain Resource Nurse Training program: A unique approach to pain management." Journal of Pain and Symptom Management, 8(8): 549-556

4. Ferrell BR, McCaffery M (1997). "Nurses' knowledge about Equianalgesic and opioid dosing." <u>Cancer Nursing</u>, 20(3): 201-212

5. McCaffery M, Ferrell BR (1997). "Nurses' knowledge of pain assessment and management: How much progress have we made?" Journal of Pain and Symptom Management, 14(3): 175-188

6. Ferrell BR, Virani R (1998). "Institutional commitment to improved pain management: Sustaining the effort." Journal of Pharmaceutical Care in Pain and Symptom Control, 6(2): 43-55

7. McCaffery M, Ferrell BR (1997). "Influence of professional vs. personal role on pain assessment and use of opioids." <u>The Journal of Continuing Education in Nursing</u>, 28(2): 69-77

8. McCaffery M, Ferrell BR (1999). "Opioids and pain management - What do nurses know?" <u>Nursing</u> 99, 29(3): 48-52

9. McCaffery M, Ferrell BR, Pasero C (2000). "Nurses' personal opinions about patients' pain and their effect on recorded assessments and titration of opioid doses." <u>Pain Management Nursing</u>, 1(3): 79-87

Knowledge and Attitudes Survey Regarding Pain

True/False - Circle the correct answer.

T 1. Vital signs are always reliable indicators of the intensity of a patient's pain. F Т 2. Because their nervous system is underdeveloped, children under two years of age have F decreased pain sensitivity and limited memory of painful experiences. F 3. Patients who can be distracted from pain usually do not have severe pain. T F 4. Patients may sleep in spite of severe pain. T T F 5. Aspirin and other nonsteroidal anti-inflammatory agents are NOT effective analgesics for painful bone metastases. Т F 6. Respiratory depression rarely occurs in patients who have been receiving stable doses of opioids over a period of months. Т F 7. Combining analgesics that work by different mechanisms (e.g., combining an opioid with an NSAID) may result in better pain control with fewer side effects than using a single analgesic agent. Т F 8. The usual duration of analgesia of 1-2 mg morphine IV is 4-5 hours. 9. Research shows that promethazine (Phenergan) and hydroxyzine (Vistaril) are reliable T F potentiators of opioid analgesics. Т F 10. Opioids should not be used in patients with a history of substance abuse. 11. Morphine has a dose ceiling (i.e., a dose above which no greater pain relief can be Т F obtained). Т F 12. Elderly patients cannot tolerate opioids for pain relief. F 13. Patients should be encouraged to endure as much pain as possible before using an opioid. Т T F 14. Children less than 11 years old cannot reliably report pain so nurses should rely solely on the parent's assessment of the child's pain intensity. T F 15. Patients' spiritual beliefs may lead them to think pain and suffering are necessary. F T 16. After an initial dose of opioid analgesic is given, subsequent doses should be adjusted in accordance with the individual patient's response. T F 17. Giving patients sterile water by injection (placebo) is a useful test to determine if the pain is real. 18. Vicodin (hydrocodone 5 mg + acetaminophen 500 mg) PO is approximately equal to 5-10 T F mg of morphine PO. 19. If the source of the patient's pain is unknown, opioids should not be used during the pain T F evaluation period, as this could mask the ability to correctly diagnose the cause of pain. T F 20. Anticonvulsant drugs such as gabapentin (Neurontin) produce optimal pain relief after a single dose. F 21. Benzodiazepines are not effective pain relievers unless the pain is due to muscle spasm. T Т F 22. Narcotic/opioid addiction is defined as a chronic neurobiologic disease, characterized by behaviors that include one or more of the following: impaired control over drug use, compulsive use, continued use despite harm, and craving.

Multiple Choice - Place a check by the correct answer.

- 23. The recommended route of administration of opioid analgesics for patients with persistent cancer-related pain is
 - a. intravenous b. intramuscular
 - c. subcutaneous
 - d. oral
 - e. rectal
- 24. The recommended route administration of opioid analgesics for patients with brief, severe pain of sudden onset such as trauma or postoperative pain is
 - a. intravenous
 - _____b. intramuscular
 - _____ c. subcutaneous
 - _____ d. oral
 - _____e. rectal
- 25. Which of the following analgesic medications is considered the drug of choice for the treatment of <u>prolonged</u> moderate to severe pain for cancer patients?
 - a. codeine
 - _____ b. morphine
 - _____ c. meperidine
 - _____ d. tramadol
- 26. Which of the following IV doses of morphine administered over a 4 hour period would be equivalent to 30 mg of oral morphine given q 4 hours?
 - a. Morphine 5 mg IV
 - b. Morphine 10 mg IV
 - _____ c. Morphine 30 mg IV
 - _____ d. Morphine 60 mg IV
- 27. Analgesics for post-operative pain should initially be given
 - a. around the clock on a fixed schedule
 - _____ b. only when the patient asks for the medication
 - c. only when the nurse determines that the patient has moderate or greater discomfort
- 28. A patient with persistent cancer pain has been receiving daily opioid analgesics for 2 months. Yesterday the patient was receiving morphine 200 mg/hour intravenously. Today he has been receiving 250 mg/hour intravenously. The likelihood of the patient developing clinically significant respiratory depression in the absence of new comorbidity is
 - _____a. less than 1%
 - _____b. 1-10%
 - _____ c. 11-20%
 - _____ d. 21-40%
 - _____e. > 41%
- 29. The most likely reason a patient with pain would request increased doses of pain medication is
 - a. The patient is experiencing increased pain.
 - b. The patient is experiencing increased anxiety or depression.
 - c. The patient is requesting more staff attention.
 - d. The patient's requests are related to addiction.
- 30. Which of the following is useful for treatment of cancer pain?
 - _____a. Ibuprofen (Motrin)
 - b. Hydromorphone (Dilaudid)
 - c. Gabapentin (Neurontin)
 - _____ d. All of the above

- 31. The most accurate judge of the intensity of the patient's pain is
 - a. the treating physician
 - _ b. the patient's primary nurse
 - c. the patient
 - d. the pharmacist
 - e. the patient's spouse or family
- 32. Which of the following describes the best approach for cultural considerations in caring for patients in pain: a. There are no longer cultural influences in the U.S. due to the diversity of the population.
 - b. Cultural influences can be determined by an individual's ethnicity (e.g., Asians are stoic, Italians are expressive, etc).
 - c. Patients should be individually assessed to determine cultural influences.
 - d. Cultural influences can be determined by an individual's socioeconomic status (e.g., blue collar workers report more pain than white collar workers).
- 33. How likely is it that patients who develop pain already have an alcohol and/or drug abuse problem?

<1% 5 - 15%25 - 50% 75 - 100%

- 34. The time to peak effect for morphine given IV is
 - a. 15 min.
 - b. 45 min.
 - c. 1 hour d. 2 hours
- 35. The time to peak effect for morphine given orally is
 - a. 5 min.
 - b. 30 min.
 - c. 1-2 hours
 - d. 3 hours
- 36. Following abrupt discontinuation of an opioid, physical dependence is manifested by the following:
 - a. sweating, yawning, diarrhea and agitation with patients when the opioid is abruptly discontinued
 - b. Impaired control over drug use, compulsive use, and craving
 - c. The need for higher doses to achieve the same effect.
 - d. a and b

Case Studies

Two patient case studies are presented. For each patient you are asked to make decisions about pain and medication.

Directions: Please select one answer for each question.

- Patient A: Andrew is 25 years old and this is his first day following abdominal surgery. As you enter his room, he smiles at 37 you and continues talking and joking with his visitor. Your assessment reveals the following information: BP = 120/80; HR = 80; R = 18; on a scale of 0 to 10 (0 = no pain/discomfort, 10 = worst pain/discomfort) he rates his pain as 8.
 - On the patient's record you must mark his pain on the scale below. Circle the number that represents your assessment A. of Andrew's pain.

0	1	2	3	4	5	6	7	8	9	10
Nop	ain/dis	comfort	:						Wo	rst
]	Pain/dis	comfort

B. Your assessment, above, is made two hours after he received morphine 2 mg IV. Half hourly pain ratings following the injection ranged from 6 to 8 and he had no clinically significant respiratory depression, sedation, or other untoward side effects. He has identified 2/10 as an acceptable level of pain relief. His physician's order for analgesia is "morphine IV 1-3 mg q1h PRN pain relief." Check the action you will take at this time.

- 1. Administer no morphine at this time. 2. Administer morphine 1 mg IV now.
- 3. Administer morphine 2 mg IV now.
- 4. Administer morphine 3 mg IV now.

- 38. Patient B: Robert is 25 years old and this is his first day following abdominal surgery. As you enter his room, he is lying quietly in bed and grimaces as he turns in bed. Your assessment reveals the following information: BP = 120/80; HR = 80; R = 18; on a scale of 0 to 10 (0 = no pain/discomfort, 10 = worst pain/discomfort) he rates his pain as 8.
 - A. On the patient's record you must mark his pain on the scale below. Circle the number that represents your assessment of Robert's pain:

0	1	2	3	4	5	6	7	8	9	10
Nop	oain/dis	comfort	:						W Pain/	orst discomfort

- B. Your assessment, above, is made two hours after he received morphine 2 mg IV. Half hourly pain ratings following the injection ranged from 6 to 8 and he had no clinically significant respiratory depression, sedation, or other untoward side effects. He has identified 2/10 as an acceptable level of pain relief. His physician's order for analgesia is "morphine IV 1-3 mg q1h PRN pain relief." Check the action you will take at this time:
- Administer no morphine at this time.
 Administer morphine 1 mg IV now.
- _3. Administer morphine 2 mg IV now.
- 4. Administer morphine 3 mg IV now.

Answer Key

Knowledge and Attitudes Survey Regarding Pain

True/False - Circle the correct answer.

- **F** 1. Vital signs are always reliable indicators of the intensity of a patient's pain.
- **F** 2. Because their nervous system is underdeveloped, children under two years of age have decreased pain sensitivity and limited memory of painful experiences.
- **F** 3. Patients who can be distracted from pain usually do not have severe pain.
- T 4. Patients may sleep in spite of severe pain.
- **F** 5. Aspirin and other nonsteroidal anti-inflammatory agents are NOT effective analgesics for painful bone metastases.
- T 6. Respiratory depression rarely occurs in patients who have been receiving stable doses of opioids over a period of months.
- T 7. Combining analgesics that work by different mechanisms (e.g., combining an opioid with an NSAID) may result in better pain control with fewer side effects than using a single analgesic agent.
- **F** 8. The usual duration of analgesia of 1-2 mg morphine IV is 4-5 hours.
- **F** 9. Research shows that promethazine (Phenergan) and hydroxyzine (Vistaril) are reliable potentiators of opioid analgesics.
- F 10. Opioids should not be used in patients with a history of substance abuse.
- **F** 11. Morphine has a dose ceiling (i.e., a dose above which no greater pain relief can be obtained).
- **F** 12. Elderly patients cannot tolerate opioids for pain relief.
- **F** 13. Patients should be encouraged to endure as much pain as possible before using an opioid.
- **F** 14. Children less than 11 years old cannot reliably report pain so nurses should rely solely on the parent's assessment of the child's pain intensity.
- T 15. Patient's spiritual beliefs may lead them to think pain and suffering are necessary.
- T 16. After an initial dose of opioid analgesic is given, subsequent doses should be adjusted in accordance with the individual patient's response.
- **F** 17. Giving patients sterile water by injection (placebo) is a useful test to determine if the pain is real.
- T 18. Vicodin (hydrocodone 5 mg + acetaminophen 500 mg) PO is approximately equal to 5 10 mg of morphine PO.
- **F** 19. If the source of the patient's pain is unknown, opioids should not be used during the pain evaluation period, as this could mask the ability to correctly diagnose the cause of pain.
- **F** 20. Anticonvulsant drugs such as gabapentin (Neurontin) produce optimal pain relief after a single dose.
- T 21. Benzodiazepines are not effective pain relievers unless the pain is due to muscle spasm.
- T 22. <u>Narcotic/opioid addiction</u> is defined as a chronic neurobiologic disease, characterized by behaviors that include one or more of the following: impaired control over drug use, compulsive use, continued use despite harm, and craving.

Answer Key

Knowledge and Attitudes Survey Regarding Pain

Multiple Choice - Place a check by the correct answer.

- 23. The recommended route of administration of opioid analgesics for patients with persistent cancer-related pain is
 - a. intravenous
 - b. intramuscular
 - _____ c. subcutaneous
 - X_d. oral
 - e. rectal
- 24. The recommended route of administration of opioid analgesics for patients with brief, severe pain of sudden onset, such as trauma or postoperative pain is
 - X a. intravenous
 - b. intramuscular
 - _____ c. subcutaneous
 - _____ d. oral
 - e. rectal
- 25. Which of the following analgesic medications is considered the drug of choice for the treatment of prolonged moderate to severe pain for cancer patients?
 - a. codeine
 - X b. morphine
 - _____ c. meperidine
 - _____ d. tramadol
- 26. Which of the following IV doses of morphine administered over a 4 hour period would be equivalent to 30 mg of oral morphine given q 4 hours
 - a. Morphine 5 mg IV
 - X_b. Morphine 10 mg IV
 - c. Morphine 30 mg IV
 - d. Morphine 60 mg IV
- 27. Analgesics for post-operative pain should initially be given
 - X a. around the clock on a fixed schedule
 - b. only when the patient asks for the medication
 - c. only when the nurse determines that the patient has moderate or greater discomfort
- 28. A patient with persistent cancer pain has been receiving daily opioid analgesics for 2 months. Yesterday the patient was receiving morphine 200 mg/hour intravenously. Today he has been receiving 250 mg/hour intravenously. The likelihood of the patient developing clinically significant respiratory depression in the absence of new comorbidity is
 - X_a. less than 1%
 - _____b. 1-10%
 - _____ c. 11-20%
 - d. 21-40%
 - _____e. > 41%
- 29. The most likely reason a patient with pain would request increased doses of pain medication is
 - X_a . The patient is experiencing increased pain.
 - b. The patient is experiencing increased anxiety or depression.
 - c. The patient is requesting more staff attention.
 - d. The patient's requests are related to addiction.
- 30. Which of the following is useful for treatment of cancer pain?
 - _____a. Ibuprofen (Motrin)
 - b. Hydromorphone (Dilaudid)
 - _____ c. Gabapentin (Neurontin)
 - X_d. All of the above

- 31. The most accurate judge of the intensity of the patient's pain is
 - a. the treating physician
 - b. the patient's primary nurse
 - X c. the patient
 - _____ d. the pharmacist
 - e. the patient's spouse or family
- 32. Which of the following describes the best approach for cultural considerations in caring for patients in pain:
 - a. There are no longer cultural influences in the U.S. due to the diversity of population.
 b. Cultural influences can be determined by an individual's ethnicity (e.g., Asians are stoic, Italians are

expressive, etc).

X c. Patients should be individually assessed to determine cultural influences.

_____d. Cultural influences can be determined by an individual's socioeconomic status (e.g., blue collar workers report more pain than white collar workers).

33. How likely is it that patients who develop pain already have an alcohol and/or drug abuse problem?

<1% **5 - 15%** 25 - 50% 75 - 100%

- 34. The time to peak effect for morphine given IV is
 - <u>X</u> a. 15 min.
 - b. 45 min.
 - c. 1 hour
 - d. 2 hours
- 35. The time to peak effect for morphine given orally is
 - a. 5 min.
 - b. 30 min
 - <u>X</u> c. 1-2 hours
 - d. 3 hours
- 36. Following abrupt discontinuation of an opioid, physical dependence is manifested by the following:
 - \underline{X} a. sweating, yawning, diarrhea and agitation with patients when the opioid is abruptly discontinued
 - b. Impaired control over drug use, compulsive use, and craving
 - c. The need for higher doses to achieve the same effect.

d. a and b

Case Studies

Two patient case studies are presented. For each patient you are asked to make decisions about pain and medication.

Directions: Please select one answer for each question.

37. Patient A: Andrew is 25 years old and this is his first day following abdominal surgery. As you enter his room, he smiles at you and continues talking and joking with his visitor. Your assessment reveals the following information: BP = 120/80; HR = 80; R = 18; on a scale of 0 to 10 (0 = no pain/discomfort, 10 = worst pain/discomfort) he rates his pain as 8.

A. On the patient's record you must mark his pain on the scale below. Circle the number that represents your assessment of Andrew's pain.

0 1 2 3 4 5 6 7 **8** 9 10 No pain/discomfort Worst Pain/discomfort

B. Your assessment, above, is made two hours after he received morphine 2 mg IV. Half hourly pain ratings following the injection ranged from 6 to 8 and he had no clinically significant respiratory depression, sedation, or other untoward side effects. He has identified 2/10 as an acceptable level of pain relief. His physician's order for analgesia is "morphine IV 1-3 mg q1h PRN pain relief." Check the action you will take at this time.

- _____1. Administer no morphine at this time.
- 2. Administer morphine 1 mg IV now.
- 3. Administer morphine 2 mg IV now.

X_4. Administer morphine 3 mg IV now.

38. Patient B: Robert is 25 years old and this is his first day following abdominal surgery. As you enter his room, he is lying quietly in bed and grimaces as he turns in bed. Your assessment reveals the following information: BP = 120/80; HR = 80; R = 18; on a scale of 0 to 10 (0 = no pain/discomfort, 10 = worst pain/discomfort) he rates his pain as 8. A. On the patient's record you must mark his pain on the scale below. Circle the number that represents your assessment of Robert's pain:

0	1	2	3	4	5	6	7	8	9	10	
Nop	pain/dis	comfor	t						Pai	Worst n/discomfo	rt

B. Your assessment, above, is made two hours after he received morphine 2 mg IV. Half hourly pain ratings following the injection ranged from 6 to 8 and he had no clinically significant respiratory depression, sedation, or other untoward side effects. He has identified 2/10 as an acceptable level of pain relief. His physician's order for analgesia is "morphine IV 1-3 mg qlh PRN pain relief." Check the action you will take at this time:

- 1. Administer no morphine at this time.
- 2. Administer morphine 1 mg IV now.
- 3. Administer morphine 2 mg IV now.
- X_4. Administer morphine 3 mg IV now.

Appendix G: Ethics Approval from RMIT University



30th March 2011

Hamdan Albaqawi

Dear Hamdan

ASEHAPP 04 – 11 ALBAQAWI Pain Management in Hail Region Hospitals

Thank you for submitting your amended application for review.

I am pleased to inform you that the CHEAN has approved your application for a period of <u>33 Months</u> to <u>December 2013</u> and your research may now proceed.

The CHEAN would like to remind you that:

All data should be stored on University Network systems. These systems provide high levels of manageable security and data integrity, can provide secure remote access, are backed up on a regular basis and can provide Disaster Recover processes should a large scale incident occur. The use of portable devices such as CDs and memory sticks is valid for archiving; data transport where necessary and for some works in progress. The authoritative copy of all current data should reside on appropriate network systems; and the Principal Investigator is responsible for the retention and storage of the original data pertaining to the project for a minimum period of five years.

Annual reports are due during December for all research projects that have been approved by the College Human Ethics Advisory Network (CHEAN).

The necessary form can be found at: http://www.rmit.edu.au/governance/committees/hrec

Yours faithfully,

Linda Jones Acting Chair, Science Engineering & Health College Human Ethics Advisory Network 'A'

Cc Supervisor/s: Phillip Maude School of Health Sciences Lina Shawan-Akl School of Health Sciences **RMIT University**

Science Engineering and Health

College Human Ethics Advisory Network (CHEAN)

Plenty Road Bundoora VIC 3083

PO Box 71 Bundoora VIC 3083 Australia

Tel. +61 3 9925 7096 Fax +61 3 9925 6506 • www.rmit.edu.au

Appendix H: Permission from Hospitals

 Dear Mr. Hamdan Mohammad Albaqawi 15 December 2010 It is my pleasure to inform you that we approved your study titled: Pain Management in Hail Region Hospitals of Saudi Arabia. Please be informed that in conducting this study, you as Principle Investigator is required to abide by the rules and regulations of the Government of Saudi Arabia and Ministry of Health. The approval of this proposal will automatically be suspended 2 March 2012 pending the replication to renew the approval. Please observe the following: Please observe the following: Personal indentifying data should only be collected when necessary for research; The data collected should only be used for this proposal; Data should be stored securely so that only a few authorised users are permitted access to the database; Secondary discloser of personal indentified data is not allowed An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA 	م م الا	الرقــــم : التاريـــخ :/	وزارة المحة Ministry of Health	الملكال حية السبحول تي وزارة الصحـة يرية العامة للشنون الصحية بمنطقة حالل
It is my pleasure to inform you that we approved your study titled: Pain Management in Hail Region Hospitals of Saudi Arabia. Please be informed that in conducting this study, you as Principle Investigator is required to abide by the rules and regulations of the Government of Saudi Arabia and Ministry of Health. The approval of this proposal will automatically be suspended 2 March 2012 pending the replication to renew the approval. Please observe the following: 1. Personal indentifying data should only be collected when necessary for research; 2. The data collected should only be used for this proposal; 3. Data should be stored securely so that only a few authorised users are permitted access to the database; 4. Secondary discloser of personal indentified data is not allowed An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	Dear M	lr. Hamdan Mohammad Alba	aqawi	15 December 2010
 Please be informed that in conducting this study, you as Principle Investigator is required to abide by the rules and regulations of the Government of Saudi Arabia and Ministry of Health. The approval of this proposal will automatically be suspended 2 March 2012 pending the replication to renew the approval. Please observe the following: Personal indentifying data should only be collected when necessary for research; The data collected should only be used for this proposal; Data should be stored securely so that only a few authorised users are permitted access to the database; Secondary discloser of personal indentified data is not allowed An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs: Hail, KSA 	lt is my Region	/ pleasure to inform you that Hospitals of Saudi Arabia.	we approved your stud	y titled: Pain Management in Hail
 Please observe the following: Personal indentifying data should only be collected when necessary for research; The data collected should only be used for this proposal; Data should be stored securely so that only a few authorised users are permitted access to the database; Secondary discloser of personal indentified data is not allowed An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA 	Please abide b The ap replica	be informed that in conduction by the rules and regulations on proval of this proposal will a tion to renew the approval.	ng this study, you as Pr f the Government of Sa automatically be suspen	inciple Investigator is required to uudi Arabia and Ministry of Health. ded 2 March 2012 pending the
 Personal indentifying data should only be collected when necessary for research; The data collected should only be used for this proposal; Data should be stored securely so that only a few authorised users are permitted access to the database; Secondary discloser of personal indentified data is not allowed An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	Please	observe the following:		
An ethical approval from human ethics committee in RMIT Australia is required to begin collecting your data at Hail region hospitals. General Director of Health Affairs in Hail Region support this research project and look forward to receiving a copy of the final result and dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	1. 2. 3. 4.	Personal indentifying data s The data collected should of Data should be stored secur the database; Secondary discloser of pers	should only be collected nly be used for this pro rely so that only a few a sonal indentified data is	l when necessary for research; posal; uthorised users are permitted access to not allowed
dissertation. Should you have any inquires or concerns in relation to this approval do not hesitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	An eth collect suppo	nical approval from human et ting your data at Hail region rt this research project and lo	thics committee in RMI hospitals. General Dire bok forward to receiving	T Australia is required to begin etor of Health Affairs in Hail Region g a copy of the final result and
Should you have any inquires or concerns in relation to this approval do not itestitate to contact Mr. Talal Majed on We wish you every success in your research endeavor. Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	disser	tation.		to contact
Sincerely, Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	Shoul Mr. T We w	d you have any inquires or co alal Majed on vish you every success in you	r research endeavor.	
Dr. Nawaf Al Harthy General Director of Health Affairs Hail, KSA	Since	rely,		
	Dr. N Gener Hail,	awaf Al Harthy ral Director of Health Affairs KSA	5	
		1		

Knowledge about Pain and Pain Management

• What is your experience of working with other nurses and their overall level of knowledge about pain and pain management? What do you think about your knowledge? Are you familiar with the medications given here? Do you spend time reading about medications? Are the medications used in Saudi Arabia are similar to the medications in your country? Would you like to learn more about pain and medications?

Assessment of pain

• Explain how pain is managed in your hospital. Are you satisfied? What about the assessment of pain? Which tools are you using to assess pain, and are these tools effective or not? Do you have training in how to use these tools?

Service and environmental barriers to pain

• What factors assist your hospital to improve pain management, and what are the limitations? Do you need intensive courses about pain? What about these factors (too busy, shortage, large number of patients, lack of educational preparation for nurses, no courses given about pain, limited communication or language barriers, patients opioid addiction), and do you think these are reasonable excuses? Tell me about your department and how many staff you have in the shift. What you will do if a patient is in pain? What about your decision-making regarding this patient? What you will do if you do not have narcotics in your department? What about the other methods for relieving pain?

Patient factors of pain

• Do you think that patients should tolerate pain? If yes, why do you think so? Do you tolerate pain? Do you think that some patients are scared to ask for pain relief? Which of these do you think can tolerate much more pain: men, women or children?

Culture and pain

• What about cultural factors? Do you think cultural factors could influence your attitude towards pain management? Do you think it could influence patients' attitudes towards expressing their pain? What about religious beliefs? What is your understanding about Muslim beliefs about pain (i.e. tolerating pain)?

Pain management

 Do nurses discuss medications with doctors and pharmacists? How much responsibility do you think a nurse has for pain management? Would a nurse initiate pain management by seeking a prescription? How do you think nurses manage PRN medication? Can you give me examples of patients who have had effective PRN pain management and ones that have not?

Validation of data

• The survey I conducted recently revealed that nurses have a low level of overall knowledge regarding pain. Do you think this finding is correct for most of the nurses you work with? Can you think of any reasons why nurses would have problems understanding pain and pain management?

Appendix J: Conference Abstract

WEI International European	October 14-17, 2012
Academic Conference Proceedings	Zagreb, Croatia

NURSES' KNOWLEDGE REGARDING PAIN MANAGEMENT IN HAIL REGION HOSPITALS, SAUDI ARABIA.

Hamdan Albaqawi

Bcs, RN, MSN, PhD candidate (School of Health Sciences), RMIT University, Australia

Abstract: Pain is a human experience that affects overall quality of life and one of the most common reasons for people seeking health care. Effective pain management requires precise knowledge, attitude and competent assessment skills. The aims of this study are to determine nurses' knowledge and attitude regarding pain management and seek to identify possible barriers to achieve optimal pain management in Hail Region Hospitals in Saudi Arabia (SA). This explorative descriptive mixed methods study sampled local and expatriate nurses who are working in Hail Region Hospitals. This study include two phases, the first phase involved administration of a questionnaire to nurses (N = 303) which sought to identify the nurses' knowledge and attitudes regarding pain management using the "Knowledge and Attitudes Survey Regarding Pain" (NKASRP). The second phase involved semi-structured interviews (N=28) to further explore their perceived facilitators and barriers to proper pain management. The interviews illicit information on how cultural differences of Saudi national and expatriate nurses might have an impact on the assessment and interpretation of patients' pain and how it will affect the delivery of effective pain management, as well as identifying the barriers to achieve optimal pain management in Hail Region Hospitals. Data were analyzed by using descriptive statistics, Measures of Variability and Inferential Statistics. The average of correct response rate was only 41.75%, with rates ranging from 5% to 87%. The qualitative data analysed using thematic analysis. The finding indicates inadequate knowledge regarding pain, pain assessment, pain management and pain medications. It's also highlighted some barratries that affecting nurses to provide an effective pain management and determined the demographic and cultural factors that impact on the delivery of effective pain management.

Keywords: Pain assessment, Pain management, Nurses' Knowledge, Nurses' attitudes, Saudi Arabia.